Applications of hyperbolic geometry to continued fractions and Diophantine approximation

by

Robert Hines

B.S., Mathematics, University of Michigan, 2005M.S., Mathematics, The Ohio State University, 2009

A thesis submitted to the Faculty of the Graduate School of the University of Colorado in partial fulfillment of the requirements for the degree of Doctor of Philosophy Department of Mathematics

2019

This thesis entitled: Applications of hyperbolic geometry to continued fractions and Diophantine approximation written by Robert Hines has been approved for the Department of Mathematics

Prof. Katherine E. Stange

Prof. Elena Fuchs

Date _____

The final copy of this thesis has been examined by the signatories, and we find that both the content and the form meet acceptable presentation standards of scholarly work in the above mentioned discipline.

Hines, Robert (Ph.D., Mathematics)

Applications of hyperbolic geometry to continued fractions and Diophantine approximation Thesis directed by Prof. Katherine E. Stange

This dissertation explores relations between hyperbolic geometry and Diophantine approximation, with an emphasis on continued fractions over the Euclidean imaginary quadratic fields, $\mathbb{Q}(\sqrt{-d}), d = 1, 2, 3, 7, 11$, and explicit examples of badly approximable numbers/vectors with an obvious geometric interpretation.

The first three chapters are mostly expository. Chapter 1 briefly recalls the necessary hyperbolic geometry and a geometric discussion of binary quadratic and Hermitian forms. Chapter 2 briefly recalls the relation between badly approximable systems of linear forms and bounded trajectories in the space of unimodular lattices (the *Dani correspondence*). Chapter 3 is a survey of continued fractions from the point of view of hyperbolic geometry and homogeneous dynamics. The chapter discusses simple continued fractions, nearest integer continued fractions over the Euclidean imaginary quadratic fields, and includes a summary of A. L. Schmidt's continued fractions over $\mathbb{Q}(\sqrt{-1})$.

Chapters 4 and 5 contain the bulk of the original research. Chapter 4 discusses a class of dynamical systems on the complex plane associated to polyhedra whose faces are two-colorable (i.e. edge-adjacent faces do not share a color). To any such polyhedron, one can associated a right-angled hyperbolic Coxeter group generated by reflections in the faces of a (combinatorially equivalent) right-angled ideal polyhedron in hyperbolic 3-space. After some generalities, we discuss a simpler system, billiards in the ideal hyperbolic triangle. We then discuss continued fractions over $\mathbb{Q}(\sqrt{-1})$ and $\mathbb{Q}(\sqrt{-2})$ coming from the regular ideal right-angled octahedron and cubeoctahedron.

Chapter 5 gives explicit examples of numbers/vectors in $\mathbb{R}^r \times \mathbb{C}^s$ that are badly approximable over number fields F of signature (r, s) with respect to the diagonal embedding. One should think of these examples as generalizations of real quadratic irrationalities, which we discuss first as our prototype. The examples are the zeros of (totally indefinite anisotropic *F*-rational) binary quadratic and Hermitian forms (the Hermitian case arises when *F* is CM). Such forms can be interpreted as compact totally geodesic subspaces in the relevant locally symmetric spaces $SL_2(\mathcal{O}_F) \setminus SL_2(F \otimes \mathbb{R})/SO_2(\mathbb{R})^r \times SU_2(\mathbb{C})^s$. We discuss these examples from a few different angles: simple arguments stemming from Liouville's theorem on rational approximation to algebraic numbers, arguments using continued fractions (of the sorts considered in chapters 3 and 4) when they are available, and appealing to the Dani correspondence in the general case. Perhaps of special note are examples of badly approximable *algebraic* numbers and vectors, as noted in 5.10.

Chapter 6 considers approximation in \mathbb{R}^n (in the boundary $\partial \mathcal{H}^n$ of hyperbolic *n*-space) over "weakly Euclidean" orders in definite Clifford algebras. This includes a discussion of the relevant background on the " SL_2 " model of hyperbolic isometries (with coefficients in a Clifford algbra) and a discription of the continued fraction algorithm. Some exploration in the case $\mathbb{Z}^3 \subseteq \mathbb{R}^3$ is included, along with proofs that zeros of anisotropic rational Hermitian forms are "badly approximable," and that the partial quotients of such zeros are bounded (conditional on increasing convergent denominators).

Chapter 7 considers simultaneous approximation in $\mathbb{R}^r \times \mathbb{C}^s$ as a subset of the boundary of $(\mathcal{H}^2)^r \times (\mathcal{H}^3)^s$ over a diagonally embedding number field of signature (r, s). A continued fraction algorithm is proposed for norm-Euclidean number fields, but not even convergence is established. Some exploration and experimentation over the norm-Euclidean field $\mathbb{Q}(\sqrt{2})$ is included.

Finally, chapter 8 includes some miscellaneous results related to the discrete Markoff spectrum. First, some identities for sums over Markoff numbers are proven (although they are closely related to Mcshane's identity). Secondly, transcendence of certain limits of roots of Markoff forms is established (a simple corollary to [AB05]). These transcendental numbers are badly approximable with only ones and twos in their continued fraction expansion, and can be written as infinite sums of ratios of Markoff numbers indexed by a path in the tree associated to solutions of the Markoff equation $x^2 + y^2 + z^2 = 3xyz$. The geometry in this chapter can all be associated to a once-punctured torus (with complete hyperbolic metric), going back to the observation of H. Cohn [Coh55] that the Markoff equation is a special case of Fricke's trace identity

$$tr(A)^{2} + tr(B)^{2} + tr(AB)^{2} = tr(A)tr(B)tr(AB) + tr(ABA^{-1}B^{-1}) + 2, \ A, B \in SL_{2}$$

in the case that A and B are hyperbolic with parabolic commutator of trace -2 (in particular for the torus associated to the commutator subgroup of $SL_2(\mathbb{Z})$).

Contents

Chapter

1	Som	ie hyper	bolic geometry	1
	1.1	Introd	uction	1
	1.2	Upper	half-space model in dimensions two and three	3
	1.3	Binary	y quadratic and Hermitian forms	4
	1.4	The g	eodesic flow in \mathcal{H}^2 and \mathcal{H}^3	6
2	Bad	ly appr	oximable systems of linear forms and the Dani correspondence	7
	2.1	Introd	$uction \ldots \ldots$	7
	2.2	The s	pace of unimodular lattices and Mahler's compactness criterion	7
	2.3	Comp	act quotients from anisotropic forms	8
	2.4	Dani o	correspondence	9
3	Con	tinued	fractions in \mathbb{R} and \mathbb{C}	13
	3.1	Introd	uction	13
	3.2	Simple	e continued fractions	13
		3.2.1	Invertible extension and invariant measure	15
		3.2.2	Direct proof of ergodicity	17
		3.2.3	Consequences of ergodicity	18
		3.2.4	Geodesic flow on the modular surface and continued fractions	20
		3.2.5	Mixing of the geodesic flow	25

	3.3	Neares	st integer continued fractions over the Euclidean imaginary quadratic fields $\ . \ .$	26
		3.3.1	Appendix: Monotonicity of denominators for $d = 7, 11$	33
	3.4	An ov	erview of A. L. Schmidt's continued fractions	35
4	Con	tinued f	fractions for some right-angled hyperbolic Coxeter groups	41
	4.1	Introd	uction	41
	4.2	A gen	eral construction	41
		4.2.1	Right-angled hyperbolic Coxeter groups from two-colorable polyhedra $\ . \ . \ .$	41
		4.2.2	A pair of dynamical systems	42
		4.2.3	Symbolic dynamics	44
		4.2.4	Invariant measures and ergodicity	44
	4.3	Billiar	ds in the ideal triangle	45
	4.4	Octah	edral ("Super-Apollonian") continued fractions over $\mathbb{Q}(\sqrt{-1})$	49
		4.4.1	Digression on the Apollonian super-packing and the Super-Apollonian group	49
		4.4.2	A pair of dynamical systems	54
		4.4.3	A Euclidean algorithm	57
		4.4.4	Geometry of the first approximation constant for $\mathbb{Q}(i)$	59
		4.4.5	Quality of rational approximation	60
		4.4.6	Invertible extension and invariant measures	64
	4.5	Cubec	octahedral continued fractions over $\mathbb{Q}(\sqrt{-2})$	70
		4.5.1	Dual pair of dynamical systems, invertible extension, and invariant measures	70
		4.5.2	Geometry of the first approximation constant for $\mathbb{Q}(\sqrt{-2})$	77
		4.5.3	Quality of approximation	78
	4.6	Antip	risms and questions	81
5	Cha	racteriz	ations and examples of badly approximable numbers	85
	5.1	Introd	uction	85
	5.2	Simple	${\rm e}$ continued fractions, the modular surface, and quadratic irrationalities $\ . \ . \ .$	88

vii

	5.3	Bounded partial quotients (nearest integer)	92
	5.4	Orbits bounded away from fixed points (right-angled)	93
	5.5	Bounded geodesic trajectories (Dani correspondence)	95
	5.6	Examples (zeros of totally indefinite anisotropic rational binary quadratic and Her-	
		mitian forms)	96
		5.6.1 Totally indefinite binary quadratic forms	96
		5.6.2 Totally indefinite binary Hermitian forms over CM fields	97
		5.6.3 Examples from quadatic forms	98
		5.6.4 Examples from Hermitian forms	99
	5.7	Examples via nearest integer continued fractions	00
	5.8	Examples via right-angled continued fractions)6
	5.9	Examples à la Liouville)9
	5.10	Non-quadratic algebraic examples over CM fields	10
6	Cont	inued fractions for weakly Euclidean orders in definite Clifford algebras 11	4
6	Cont 6.1	tinued fractions for weakly Euclidean orders in definite Clifford algebras 11 Introduction 11	14
6	Cont 6.1 6.2	inued fractions for weakly Euclidean orders in definite Clifford algebras 11 Introduction 11 Clifford algebras 11	14 14 15
6	Cont 6.1 6.2 6.3	Linued fractions for weakly Euclidean orders in definite Clifford algebras 11 Introduction 11 Clifford algebras 11 A special case 11	14 14 15
6	Cont 6.1 6.2 6.3 6.4	tinued fractions for weakly Euclidean orders in definite Clifford algebras 11 Introduction 12 Clifford algebras 12 A special case 12 S^n as a projective line and the Vahlen group 12	14 14 15 16
6	Cont 6.1 6.2 6.3 6.4 6.5	inued fractions for weakly Euclidean orders in definite Clifford algebras 11 Introduction 12 Clifford algebras 12 A special case 12 S^n as a projective line and the Vahlen group 12 Continued fractions for weakly Euclidean orders 12	14 14 15 16 17
6	Cont 6.1 6.2 6.3 6.4 6.5 6.6	Linued fractions for weakly Euclidean orders in definite Clifford algebras 11 Introduction 12 Clifford algebras 12 A special case 12 S^n as a projective line and the Vahlen group 12 Continued fractions for weakly Euclidean orders 12 Approximation in \mathbb{R}^3 via quaternions 12	14 14 15 16 17 18
6	Cont 6.1 6.2 6.3 6.4 6.5 6.6	tinued fractions for weakly Euclidean orders in definite Clifford algebras 11 Introduction 12 Clifford algebras 12 A special case 12 S^n as a projective line and the Vahlen group 12 Continued fractions for weakly Euclidean orders 12 Approximation in \mathbb{R}^3 via quaternions 12 6.6.1 Badly approximable triples from Hermitian forms 12	14 14 15 16 17 18 20 20
6	Cont 6.1 6.2 6.3 6.4 6.5 6.6	tinued fractions for weakly Euclidean orders in definite Clifford algebras11Introduction11Clifford algebras12A special case13S ⁿ as a projective line and the Vahlen group13Continued fractions for weakly Euclidean orders13Approximation in \mathbb{R}^3 via quaternions146.6.1Badly approximable triples from Hermitian forms156.6.2Experimentation and visualization for $\mathbb{Z}^3 \subseteq \mathbb{R}^3$ 15	14 15 16 17 18 20 25
6	Cont 6.1 6.2 6.3 6.4 6.5 6.6	tinued fractions for weakly Euclidean orders in definite Clifford algebras11Introduction11Clifford algebras11A special case11A special case11 S^n as a projective line and the Vahlen group11Continued fractions for weakly Euclidean orders11Approximation in \mathbb{R}^3 via quaternions126.6.1Badly approximable triples from Hermitian forms126.6.2Experimentation and visualization for $\mathbb{Z}^3 \subseteq \mathbb{R}^3$ 12Some examples of orders12	14 14 15 16 17 18 20 25 26
6	Cont 6.1 6.2 6.3 6.4 6.5 6.6 6.7 Near	tinued fractions for weakly Euclidean orders in definite Clifford algebras11Introduction11Clifford algebras11A special case11A special case11 S^n as a projective line and the Vahlen group11Continued fractions for weakly Euclidean orders11Approximation in \mathbb{R}^3 via quaternions126.6.1Badly approximable triples from Hermitian forms126.6.2Experimentation and visualization for $\mathbb{Z}^3 \subseteq \mathbb{R}^3$ 12Some examples of orders12vest integer continued fractions over norm-Euclidean number rings and simultaneous12	14 14 15 16 17 18 20 20 20 25 26
6	Cont 6.1 6.2 6.3 6.4 6.5 6.6 6.7 Near appr	Introduction for weakly Euclidean orders in definite Clifford algebras 11 Introduction	14 14 15 16 17 18 20 25 26 29

	7.2	$Simultaneous \ approximation \ over \ norm-Euclidean \ number \ fields \ via \ continued \ fractions 130$
	7.3	Experimentation in $\mathbb{Q}(\sqrt{2})$
8	Mise	cellaneous results on the discrete Markoff spectrum 136
	8.1	Introduction
	8.2	Markoff numbers and forms, Frobenius coordinates, and Cohn matrices
	8.3	A variation of Mcshane's identity
	8.4	Transcendence of limits of Markoff forms

Bibliography

144

Figures

Figure

3.1	$T\xi = \{\frac{1}{\xi}\}$	15
3.2	Invertible extension of the (slow) Gauss map on $(-\infty, 0) \times (1, \infty) \cup (-\infty, -1) \times (0, 1)$.	16
3.3	Distribution of the normalized error for almost every ξ	21
3.4	Visualizing S as first return of the geodesic flow to $\pi(\mathcal{C})$	23
3.5	∂V and translates (blue), $\partial (V^{-1})$ (red), and unit circle (black) for $d=1,2,3,7,11.$.	28
3.6	The numbers q_{n-1}/q_n and q_n/q_{n-1} , $1 \le n \le 10$, for 5000 randomly chosen z over	
	$\mathbb{Q}(\sqrt{-1})$ (left) and $\mathbb{Q}(\sqrt{-3})$ (right)	29
3.7	In the proof of Proposition 3.3.1, the assumption $k_n < 1$ with $a_n = \frac{1+\sqrt{-7}}{2}$ (left)	
	or $a_n = \frac{1+\sqrt{-11}}{2}$ (right) leads to restricted potential values for k_i with $i < n$ (green	
	regions)	36
3.8	Subdivision of the upper half-plane $\mathcal I$ (model circle) and ideal triangle $\mathcal I^*$ with ver-	
	tices $\{0, 1, \infty\}$ (model triangle). The continued fraction map T sends the circular	
	regions onto \mathcal{I} and the triangular regions onto \mathcal{I}^* .	38
3.9	Domain of the invertible extension, $\mathcal{I} \cup \overline{\mathcal{I}^*}$ (left) and $\mathcal{I}^* \cup \overline{\mathcal{I}}$ (right)	40
4.1	Partion of the line induced by reduced words of length three in the Coxeter generators	46
4.9	Crank of $T(x)$ with fixed points at 0, 1, and co	10
4.2	Graph of $T(x)$, with fixed points at 0, 1, and ∞ .	40
4.3	Approximating $x = 0.4189513796210592, x = bacabcacbcacacababac$	47
4.4	Two iterations of \widetilde{T} , red to green to blue	48

4.5	Partion of the plane induced by words of length 1, 2, 3, and 4 in the Coxeter gen-	
	erators (associated to the map T_B), or the orbit of the action of words from \mathcal{A}^S of	
	length 0, 1, 2, and 3 on the base quadruple R_B	50
4.6	Circles in $[0,1] \times [0,1]$ in the orbit of words of length ≤ 5 from \mathcal{A}^S acting on the base	
	quadruple R_B	51
4.7	A Descartes quadruple and its dual, the four "swaps" S_i , and the four "inversions"	
	S_i^{\perp}	52
4.8	The dual base quadruples R_A (red) and R_B (blue)	54
4.9	The regions A_i, A'_i, B_i, B'_i	55
4.10	Partition of the complex plane induced by words of length two in swap normal form	
	in the Möbius generators.	56
4.11	Approximating the irrational point $0.3828008104 + i0.2638108161$ using T_B	
	gives the normal form word $\mathfrak{s}_3^{\perp}\mathfrak{s}_2\mathfrak{s}_2^{\perp}\mathfrak{s}_3^{\perp}\mathfrak{s}_1\mathfrak{s}_1^{\perp}\mathfrak{s}_4\mathfrak{s}_2\mathfrak{s}_4\mathfrak{s}_1\mathfrak{s}_1^{\perp}\mathfrak{s}_3\mathfrak{s}_2\mathfrak{s}_3\mathfrak{s}_3^{\perp}\mathfrak{s}_4\mathfrak{s}_4^{\perp}\mathfrak{s}_2\mathfrak{s}_1\mathfrak{s}_2\cdots$.	57
4.12	Horoball covering of the ideal triangular face with vertices $\{0, 1, \infty\}$ with coordinates	
	(z,t)	60
4.13	Left: The Descartes quadruples (octahed on) when p/q first appears as a convergent	
	when inverting. Right: The result after mapping back to the fundamental octahedron	
	and sending $p/q \to \infty$ and ∞ to Q/q .	62
4.14	Left: The Descartes quadruples (octahedra) when p/q first appears as a convergent	
	when swapping. Right: The result after mapping back to the fundamental octahe-	
	dron and sending $p/q \to \infty$ and ∞ to Q/q .	63
4.15	At left, eight iterations of the map T , labelled in sequence (with $+$ and $-$ indicating	
	the orientation). At right, 100 iterations of T , with rainbow coloration indicating	
	time. In both pictures, the geodesic planes defining the octahedron are shown. $\ . \ .$	65
4.16	Top, left to right: regions $\mathcal{B}'_i \times B'_i$, $i = 1, 2, 3, 4$. Bottom, left to right: regions $\mathcal{B}_i \times B_i$,	
	i = 1, 2, 3, 4. Shown red×blue with subdivisions.	66
4.17	Graph of $f_B(z)$, the invariant density for T_B	68

4.18	Regions in the definition of T_A , boundaries given by the fixed circles of the \mathfrak{s}_i 72	2
4.19	Regions in the definition of T_B , boundaries given by the fixed circles of the \mathfrak{t}_i 73	3
4.20	Dual super-packings for the cubeoctahedron associated to $\mathbb{Q}(\sqrt{-2})$	4
4.21	Horoball covering of the ideal square face with vertices $\{0, -1/\sqrt{-2}, \sqrt{-2}, \infty\}$ with	
	coordinates (z, t)	9
4.22	A rational number p/q appearing as an approximation to z_0 for the first time with	
	respect to T_A (top) or T_B (bottom). In particular z_0 must be outside the labeled	
	disks (the interiors of A and R include infinity)	2
4.23	Result of applying γ to the configurations of Figure 4.22, with $\gamma(\infty)$ in the shaded	
	region. If p/q were an approximation to z_0 , then z_0 must be outside of the labeled	
	disks by definition of the algorithms associated to T_A and T_B	3
4.24	Antiprism reflection generators, $n = 3, 4, 69, \ldots 84$	4
5.1	Over $\mathbb{Q}(\sqrt{-1})$, we have the points $p_n/q_n = g_n(\infty)$ and $-q_{n-1}/q_n = g_n^{-1}(\infty)$, along with the unit circle and its image under g_n (black), circles of radius $1/C'$ and $C'/ q_n ^2$	
	(red), and the lines defining V and their images under g_n (blue)	4
5.2	The first 10,000 remainders z_n of $z = \sqrt{3}e^{2\pi i/5}$ over $\mathbb{Q}(i)$ (left) and the arcs $Z(H_n) \cap V$	
	(right)	5
5.3	10000 iterations of $T = T_B$ over $\mathbb{Q}(\sqrt{-1})$ on $z = \sqrt{7}(\cos(2\pi/5) + i\sin(2\pi/5))$ 105	7
5.4	10000 iterations of T_B over $\mathbb{Q}(\sqrt{-2})$ on $z = \sqrt{7}(\cos(2\pi/5) + i\sin(2\pi/5))$ 108	8
5.5	20,000 iterates of T for $z = \sqrt[3]{2} + \sqrt{\sqrt[3]{4} - n}$ over $\mathbb{Q}(\sqrt{-1})$ with $n = 4, 5, 6, 7, \ldots$ 112	1
5.6	10,000 iterates of T for $z = \sqrt[3]{2} + \sqrt{\sqrt[3]{4} - n}$ over $\mathbb{Q}(\sqrt{-3})$ with $n = 2, 3, 4, 5, \ldots$ 112	2
6.1	The ratios $q_{n-1}q_n^{-1}$, $1 \le n \le 10$, for 5000 randomly chosen points of $[-1/2, 1/2)^3$. All	
	of the points lie in the unit sphere, giving experimental evidence that the convergent	
	denominators are increasing	1
6.2	20000 remainders for $\sqrt{e/10}i + \sqrt{\pi/10}j + \sqrt{1 - (e + \pi)/10}$ lying on a finite collection	
	of spheres	5

xii

6.3	10000 remainders for $\sqrt{2}i + \sqrt{3}j$ all lying in the plane $x = 0$
6.4	Left are 10000 remainders for $\sqrt{2} + \sqrt{3}i + \sqrt{2}j$ all lying in the planes $x = \pm z$.
	Compare with 10000 remainders for $\pi + ei + \pi j$ on the right
7.1	Setup for nearest integer continued fractions over $\mathbb{Q}(\sqrt{2})$: $\mathcal{F}, \mathcal{F}^{-1}$ blue, $\mathbb{Z}[\sqrt{2}]$ green
	dots, $N(x) = 1$ red
7.2	Ratio of the denominators q_n/q_{n+1} for 5000 random numbers and $1 \le n \le 10. \ldots 135$
8.1	A plot of $(p/q, u/m)$ at depth ≤ 10 in the Farey tree (excluding $(0, 1/2)$ and $(1, 0)$). 140
8.2	Top: Part of the "half-topograph" (or tree of triples) of words in A and B giving
	Markoff forms (the fixed point forms) and Markoff numbers (one-third of the trace).
	Bottom: The continued fraction expansion of the roots of the fixed point forms from
	the \widetilde{A} , \widetilde{B} tree

Chapter 1

Some hyperbolic geometry

1.1 Introduction

In this chapter we recall some notions from hyperbolic geometry and the parameterization of pertinent geometric objects by algebraic ones. We are primarily interested in the Lie groups $G = SL_2(\mathbb{R}^r \times \mathbb{C}^s)$ and the arithmetic lattices (finite covolume discrete subgroups) $\Gamma = SL_2(\mathcal{O}_F)$ therein (where \mathcal{O}_F is the ring of integers in the number field F of signature (r, s)). The associated symmetric spaces G/K, $K \cong SO_2(\mathbb{R})^r \times SU_2(\mathbb{C})^s$, are products of hyperbolic two- and three-spaces, and $G/\{\pm 1\}^{r+s}$ is the group of orientation preserving isometries, acting on the left.

Hyperbolic *n*-space, \mathcal{H}^n , is the unique complete simply connected Riemannian *n*-manifold of constant negative sectional curvature -1 ([Lee97] Theorem 11.12), one of the prototypical model geometries (along with flat Euclidean space, curvature 0, and the round unit sphere, curvature +1). There are various models of hyperbolic space, summarized below.

- (hyperboloid) This model is (one sheet of, or the quotient by ± 1 of) the hypersurface in flat \mathbb{R}^{n+1} defined by $x_0^2 - \sum_{i=1}^n x_i^2 = 1$ with the metric $ds^2 = -dx_0^2 + \sum_{i=1}^n dx_i^2$ (which is positive definite when restricted to the hyperboloid). Its orientation preserving isometry group is $O_{\mathbb{R}}(n,1)^\circ$, the connected component of the identity of the real orthogonal group preserving the quadratic form in the definition of the metric above. The totally geodesic subspaces are the intersections of linear subspaces of \mathbb{R}^{n+1} with the hyperboloid.
- (conformal ball) Projecting from the point (-1, 0, ..., 0) to the hyperplane $x_0 = 0$ in the

hyperboloid model gives the ball $x_0 = 0$, $\sum_{i=1}^n x_i < 1$, with metric $ds^2 = \frac{4\sum_i dx_i^2}{(1-\sum_i x_i^2)^2}$. Totally geodesic subspaces are hemispheres orthogonal to the boundary of the ball. The model is conformal, i.e. the Euclidean angles and hyperbolic angles agree. The isometry group is generated by inversions in the codimension 1 hemispheres orthogonal to the boundary (i.e. inversive geometry in \mathbb{R}^n preserving the unit ball).

- (upper half-space) For this model, we take $\{(x_1, \ldots, x_n) \in \mathbb{R}^n : x_n > 0\}$ with the metric $ds^2 = \frac{\sum_i dx_i^2}{x_n^2}$. Totally geodesic subspaces are hemispheres and affine subspaces orthogonal to the boundary $x_n = 0$. The isometry group is generated by inversions in the hemispheres and reflections in the affine subspaces of codimension 1 orthogonal to the boundary (i.e. inversive geometry of \mathbb{R}^n preserving the upper half-space). Inversion (in \mathbb{R}^n) in the sphere of radius $\sqrt{2}$ centered at $(0, \ldots, 0, 1)$ exchanges the conformal ball and upper half-space models.
- (projective ball) Projecting from the origin in ℝⁿ⁺¹ to the plane x₀ = 1 in the hyperboloid model gives the projective ball model, x₀ = 1, ∑_{i=1}ⁿ x_i² < 1. Totally geodesic subspaces are (Euclidean) affine subspaces in the ball. This model is not conformal (angles are not preserved).

We will work primarily (exclusively?) with the upper half-space model since we are interested in specific discrete subgroups of the isometry group of this model.

As a prototype for our investigations (cf. chapter 3), we consider the approximation of a real number by rational numbers. Homogenizing the problem, we want to work with $P^1(\mathbb{Q}) = P^1(\mathbb{Z})$ inside of $P^1(\mathbb{R})$. $P^1(\mathbb{Q})$ is the orbit of the point $\infty = [1 : 0]$ under the action of the lattice $\Gamma = SL_2(\mathbb{Z}) \subseteq G = SL_2(\mathbb{R})$. The Euclidean algorithm on \mathbb{Z} leads to continued fractions and various questions in Diophantine approximation can be restated in terms of the geodesic flow on the modular surface $\Gamma \setminus G/K$, $K = SO_2(\mathbb{R})$.

We leave discussion of higher dimensional hyperbolic spaces and their " SL_2 " isometry groups to chapter 6. Some references for hyperbolic geometry, discrete subgroups of isometries, and their arithmetic include [Rat06], [Bea95], [EGM98], and [MR03].

1.2 Upper half-space model in dimensions two and three

In dimension two, we identify \mathcal{H}^2 with a subset of the complex plane and the ideal boundary with the real projective line

$$\mathcal{H}^2 = \{ z = x + iy \in \mathbb{C} : y > 0 \}, \ \partial \mathcal{H}^2 = P^1(\mathbb{R}).$$

With these coordinates, the metric is given by $ds^2 = \frac{dx^2 + dy^2}{y^2}$. Geodesics are vertical half-lines or semi-circles orthogonal to the real line.

In this model we have

$$\operatorname{Isom}(\mathcal{H}^2) = PGL_2(\mathbb{R}) = GL_2(\mathbb{R})/\mathbb{R}^{\times},$$

acting as fractional linear transformations

$$\begin{pmatrix} a & b \\ c & d \end{pmatrix} \cdot z = \begin{cases} \frac{az+b}{cz+d} & ad-bc > 0, \\ \frac{a\overline{z}+b}{c\overline{z}+d} & ad-bc < 0. \end{cases}$$

This action extends continuously to the ideal boundary.

In a similar fashion, we identify \mathcal{H}^3 with a subset of the Hamiltonians \mathbb{H} and the ideal boundary with the complex projective line

$$\mathcal{H}^2 = \{ \zeta = z + jt \in \mathbb{H} : z \in \mathbb{C}, t > 0 \}, \ \partial \mathcal{H}^3 = P^1(\mathbb{C}).$$

With these coordinates, the metric is given by $ds^2 = \frac{dx^2 + dy^2 + dt^2}{t^2}$ (with z = x + iy). Geodesics are vertical half-lines or semi-circles orthogonal to the complex plane.

In this model, the orientation preserving isometries are

$$\operatorname{Isom}^+(\mathcal{H}^3) = PGL_2(\mathbb{C}) = GL_2(\mathbb{C})/\mathbb{C}^{\times},$$

acting as fractional linear transformations

$$\begin{pmatrix} a & b \\ c & d \end{pmatrix} \cdot \zeta = (a\zeta + b)(c\zeta + d)^{-1}$$

Once again this action extends continuously to the ideal boundary (or vice versa, the so-called Poincaré extension).

One can continue this process, identifying a group of two-by-two matrices over definite Clifford algebras with the orientation preserving isometries in the upper half-space mode. This generalization will be discussed in chapter 6.

1.3 Binary quadratic and Hermitian forms

The points of \mathcal{H}^2 and \mathcal{H}^3 can be identified with zero sets of definite binary quadratic and Hermitian forms, for instance the map

$$G \to G/K, \ g \mapsto g^{\dagger}g,$$

where $G = SL_2(\mathbb{R})$ or $SL_2(\mathbb{C})$, $K = SO_2(\mathbb{R})$ or $SU_2(\mathbb{C})$, and \dagger is conjugate-transpose, identifies the symmetric space $G/K \cong \mathcal{H}^2$ or \mathcal{H}^3 with positive definite binary quadratic or Hermitian forms of determinant 1. Conversely, given a positive definite binary quadratic or Hermitian form

$$F(x,y) = ax^2 + bxy + cy^2 \text{ or } H(z,w) = Az\overline{z} + \overline{z}Bw + \overline{w}\overline{B}z + Cw\overline{w},$$
$$a, b, c \in \mathbb{R}, \ ac - b^2/4 > 0, \ A, C \in \mathbb{R}, \ B \in \mathbb{C}, \ \Delta := \det(H) = AC - B\overline{B} > 0,$$

we can associate the following point of \mathcal{H}^2 or \mathcal{H}^3

$$z = \frac{-b + \sqrt{b^2 - 4ac}}{2a} \in \mathcal{H}^2, \ \zeta = \frac{-B + j\sqrt{\Delta}}{A} \in \mathcal{H}^3$$

giving an SL_2 -equivariant bijection between definite forms (modulo real scalar multiplication) and \mathcal{H}^2 or \mathcal{H}^3

$$Z(F^g) = g^{-1} \cdot Z(F)$$
 or $Z(H^g) = g^{-1}Z(H)$,

where the right action on forms is given by change of variable

$$\begin{split} F^{g} &= g^{t} F g = \begin{pmatrix} \alpha^{2} a + \alpha \gamma b + \gamma^{2} c & \alpha \beta a + \frac{\alpha \delta + \beta \gamma}{2} b + \gamma \delta c \\ \alpha \beta a + \frac{\alpha \delta + \beta \gamma}{2} b + \gamma \delta c & \beta^{2} a + \beta \delta b + \delta^{2} c \end{pmatrix}, \\ H^{g} &= g^{\dagger} H g = \begin{pmatrix} |\alpha|^{2} a + \alpha \overline{\gamma} \overline{b} + \overline{\alpha} \gamma b + |\gamma|^{2} c & \overline{\alpha} \beta a + \beta \overline{\gamma} \overline{b} + \delta \overline{\alpha} b + \overline{\gamma} \delta c \\ \alpha \overline{\beta} a + \alpha \overline{\delta} \overline{b} + \overline{\beta} \gamma b + \gamma \overline{\delta} c & |\beta|^{2} a + \beta \overline{\delta} \overline{b} + \overline{\beta} \delta b + |\delta|^{2} c \end{pmatrix} \\ g &= \begin{pmatrix} \alpha & \beta \\ \gamma & \delta \end{pmatrix}, \end{split}$$

and the left action on the the zero set is the restriction of the isometric action to the boundary (linear fractional transformation),

$$g \cdot [z : w] = [az + bw : cz + dw]$$

An unoriented geodesic in \mathcal{H}^2 or \mathcal{H}^3 is determined by a pair of distinct points on the ideal boundary. This information can be encoded by an indefinite binary quadratic form

$$Q(x,y) = (\beta x - \alpha y)(\delta x - \gamma y), \ [\alpha : \beta], [\gamma : \delta] \in P^1(\mathbb{R}) \text{ or } P^1(\mathbb{C}),$$

whose zero set, Z(Q), consists of the two points at the boundary.

In \mathcal{H}^3 , geodesic planes are determined by their boundary, a copy of $P^1(\mathbb{R})$ in $P^1(\mathbb{C})$. This boundary can be described by the zero set Z(H) of an indefinite binary Hermitian form

$$\begin{split} H(z,w) &= (\bar{z}\ \bar{w}) \begin{pmatrix} A & B \\ \overline{B} & C \end{pmatrix} \begin{pmatrix} z \\ w \end{pmatrix}, \ A,C \in \mathbb{R}, \ B \in \mathbb{C}, \ \Delta := \det(H) = AC - |B|^2 < 0, \\ \\ Z(H) &= \{[z:w]:H(z,w) = 0\} = \begin{cases} \{[z:1]:|z+B/A|^2 = -\Delta/A^2\} & A \neq 0, \\ \{[1:w]:|w+\overline{B}/C|^2 = -\Delta/C^2\} & C \neq 0, \\ \\ \{[z:w]:z\overline{B}\bar{w} + \bar{z}Bw = 0\} & A = C = 0. \end{cases} \end{split}$$

Once again we have equivariance, $g^{-1}Z(H) = Z(H^g)$, where the left action on the zero set is the Möbius action and the right action on the form is change of variable.

1.4 The geodesic flow in \mathcal{H}^2 and \mathcal{H}^3

The unit tangent bundle of \mathcal{H}^2 , $T^1(\mathcal{H}^2) \subseteq \mathbb{C} \times \mathbb{C}$ can be identified with $PSL_2(\mathbb{R})$, explicitly by the simply transitive action

$$g \cdot (z, v) = \begin{pmatrix} a & b \\ c & d \end{pmatrix} \cdot (z, v) = \left(\frac{az+b}{cz+d}, \frac{v}{(cz+d)^2}\right) = \left(g \cdot z, \frac{dg}{dz} \cdot v\right),$$

after choosing a basepoint for the action ((i, i) for instance). The geodesic flow on $T^1(\mathcal{H}^2)$ is

$$\Phi_t(z,v) = \pm \left(\begin{array}{cc} a & b \\ c & d \end{array}\right) \left(\begin{array}{cc} e^{t/2} & 0 \\ 0 & e^{-t/2} \end{array}\right),$$

unit-speed flow for time t starting at z in the direction v. We can forget the tangent information and consider geodesic trajectories

$$\left(\begin{array}{cc} a & b \\ c & d \end{array}\right) \left(\begin{array}{cc} e^{t/2} & 0 \\ 0 & e^{-t/2} \end{array}\right) SO_2(\mathbb{R})$$

in $\mathcal{H}^2 \cong SL_2(\mathbb{R})/SO_2(\mathbb{R}).$

Similarly, in \mathcal{H}^3 the oriented orthonormal frame bundle can be identified with $PSL_2(\mathbb{C})$ via the derivative action (here $\eta \in \mathbb{C} + \mathbb{R}j$ is a tangent vector)

$$g \cdot (\zeta, \eta) = \left(g(\zeta), (a - g(\zeta)c)\eta(c\zeta + d)^{-1}\right) = \left(g(\zeta), (\zeta c + d)^{-1}\eta(c\zeta + d)^{-1}\right)$$

say with base point $(j, \{1, i, j\})$, the frame flow (parallel transport of a framed point $(p, \{e_1, e_2, e_3\})$ along the geodesic determined by (p, e_3)) can be written in a similar manner

$$\pm \left(\begin{array}{cc} a & b \\ c & d \end{array}\right) \left(\begin{array}{cc} e^{t/2} & 0 \\ 0 & e^{-t/2} \end{array}\right),$$

and we can forget the framing, looking only at points of \mathcal{H}^3

$$\left(\begin{array}{cc}a&b\\c&d\end{array}\right)\left(\begin{array}{c}e^{t/2}&0\\0&e^{-t/2}\end{array}\right)SU_2(\mathbb{C}).$$

Chapter 2

Badly approximable systems of linear forms and the Dani correspondence

2.1 Introduction

In this chapter, we first give a short discussion of the space of unimodular lattices in \mathbb{R}^n , $SL_n(\mathbb{Z})\backslash SL_n(\mathbb{R})$, a basic compactness criterion for subsets of lattices (Mahler's compactness criterion), and then translate between the notions of badly approximable systems of linear forms and bounded trajectories in the space of lattices (the Dani correspondence).

For more on the space of unimodular lattices, lattices in Lie groups, and arithmetic groups, see [Mor15], [Bor69], [Rag72], [PR94], [Sie89], [Ebe96].

2.2 The space of unimodular lattices and Mahler's compactness criterion

A (full) lattice $\Lambda \subseteq \mathbb{R}^n$ is a discrete additive subgroup isomorphic to \mathbb{Z}^n . A lattice Λ is unimodular if it has covolume one (e.g. the flat torus $\mathbb{R}^n/\mathbb{Z}^n$ has volume one). By choosing a \mathbb{Z} -basis for the lattice, up to change of basis, the space of unimodular lattices is the coset space $SL_n(\mathbb{Z})\backslash SL_n(\mathbb{R})$. In other words, we specify a unimodular lattice in \mathbb{R}^n by the rows of a matrix in $SL_n(\mathbb{R})$. This space is non-compact but has finite volume (equal to $\zeta(2) \cdot \ldots \cdot \zeta(n)$ when suitably normalized [Sie89]) induced by Haar measure on $SL_n(\mathbb{R})$.

In the sequel, we will make use of the following criterion for determining whether or not a subset of $SL_n(\mathbb{Z})\backslash SL_n(\mathbb{R})$ is bounded (has compact closure).

Theorem 2.2.1 (Mahler's compactness criterion, [Mah46] Theorem 2). A subset $\Omega \subseteq SL_n(\mathbb{R})$ is precompact modulo $SL_n(\mathbb{Z})$ if and only if the corresponding lattices (given by the \mathbb{Z} -span of the rows of $\omega \in \Omega$) contain no arbitrarily short vectors. In other words

$$\inf\{\|(x_1,\ldots,x_n)\omega\|_{\infty}:\omega\in\Omega,(x_1,\ldots,x_n)\in\mathbb{Z}^n\setminus\{0\}\}>0,$$

(where $||(x_1, \ldots, x_n)||_{\infty} = \max\{|x_i| : 1 \le i \le n\}$).

2.3 Compact quotients from anisotropic forms

Let $F(x) = F(x_1, ..., x_n)$ be an anisotropic integral *n*-ary form, i.e. F is a homogeneous polynomial in n variables such that

$$F(\mathbb{Z}^n) \subseteq \mathbb{Z}^n, \ F(x) \neq 0 \text{ if } 0 \neq x \in \mathbb{Z}^n,$$

with real and integral automorphism groups

$$Aut(F,\mathbb{R}) = \{g \in SL_n(\mathbb{R}) : F(xg) = F(x)\}, Aut(F,\mathbb{Z}) = SL_n(\mathbb{Z}) \cap Aut(F,\mathbb{R})\}$$

Lemma 2.3.1. If F is an anisotropic integral n-ary form, then

$$Aut(F,\mathbb{Z})\backslash Aut(F,\mathbb{R}) \subseteq SL_n(\mathbb{Z})\backslash SL_n(\mathbb{R})$$

is compact.

Proof. By Mahler's criterion, we need only show that

$$\inf\{\|xg\|: g \in Aut(F, \mathbb{R}), 0 \neq x \in \mathbb{Z}^n\} > 0,$$

since $Aut(F,\mathbb{R})$ is close in $SL_n(\mathbb{R})$. Because F is anisotropic and takes on integer values, we have

$$1 \le |F(x)| = |F(xg)|, \ 0 \ne x \in \mathbb{Z}^n, \ g \in SL_n(\mathbb{Z}).$$

Because F(x) is continuous, ||xg|| must be bounded away from zero and therefore $Aut(F, \mathbb{R})$ is compact modulo $SL_n(\mathbb{Z})$.

By restricting scalars, the proof above applies in other situations of interest, cf. 5.6.

2.4 Dani correspondence

Let

$$L = (L_1, \ldots, L_k) = (l_{ij})_{ij}$$

be a $(n-k) \times k$ matrix of real numbers whose columns represent the k linear forms $L_j(x) = \sum_{i=1}^{n-k} l_{ij}x_i$. We say that the system of linear forms is *badly approximable* if there exists C > 0 such that

$$\|x\|_{\infty}^{1/k} \cdot \|xL - p\|_{\infty} \ge C$$

for $p \in \mathbb{Z}^k$ and $0 \neq x \in \mathbb{Z}^{n-k}$.

Theorem 2.4.1 ([Dan85] Theorem 2.20). The system of linear forms L is badly approximable if and only if the trajectory

$$\left(\begin{array}{cc}I & L\\ 0 & I\end{array}\right) D_t \in SL_n(\mathbb{R})$$

is bounded modulo $SL_n(\mathbb{Z})$, where $D_t = diag(e^{-t}, \dots, e^{-t}, e^{\lambda t}, \dots, e^{\lambda t})$ and $\lambda = \frac{n-k}{k}$.

We will be interested in the case k = 1, n = 2, but with respect to other \mathbb{R} -algebras and discrete subrings, namely the spaces of (free, rank two, "unimodular") \mathcal{O}_F -modules and associated locally symmetric spaces

$$SL_2(\mathcal{O}_F) \setminus SL_2(F \otimes \mathbb{R}), \ SL_2(\mathcal{O}_F) \setminus SL_2(F \otimes \mathbb{R})/K, \ K \cong SO_2(\mathbb{R})^r \times SU_2(\mathbb{C})^s,$$

where F is a number field with r real and s conjugate pairs of complex embeddings, and \mathcal{O}_F is the ring of integers of F.

For example, the single linear form in one variable ξx ($\xi \in \mathbb{R}$ fixed) being badly approximable is the worst approximation behavior that a real number can have with respect to the rationals: there exists C > 0 such that

$$|\xi - p/q| \ge C/q^2$$
 for all $p/q \in \mathbb{Q}$.

$$SL_2(\mathbb{Z}) \begin{pmatrix} 1 & \xi \\ 0 & 1 \end{pmatrix} \begin{pmatrix} e^{-t} & 0 \\ 0 & e^t \end{pmatrix} SO_2(\mathbb{R})$$

on the modular surface. In other words, consider the geodesic ray $\{\xi + e^{-2t}i : t \ge 0\} \subseteq \mathcal{H}^2$ modulo the action of $SL_2(\mathbb{Z})$.

In this setting, we will say that a vector $z = (z_{\sigma})_{\sigma} \in \mathbb{R}^r \times \mathbb{C}^s$ (indexed by inequivalent $\sigma: F \to \mathbb{C}$) is badly approximable if there is a constant C > 0 such that

$$||q||_{\infty} ||qz - p||_{\infty} \ge C$$
 for all $p \in \mathcal{O}_F, 0 \neq q \in \mathcal{O}_F$,

where we identify $a \in F$ with the vector $(\sigma(a))_{\sigma} \in \mathbb{R}^r \times \mathbb{C}^s$ (i.e. via the homomorphism $F \to \mathbb{R}^r \times \mathbb{C}^s$ induced by the σ). Restricting scalars and applying Mahler's criterion gives the following.

Theorem 2.4.2 ([EGL16] Theorem 2.2). A subset $\Omega \subseteq SL_2(F \otimes \mathbb{R})$ is bounded modulo $SL_2(\mathcal{O}_F)$ if and only if the \mathcal{O}_F -modules $\omega \in \Omega$ (generated by the rows of ω) contain no arbitrarily short vectors, *i.e.*

$$\inf\{\|(x_1, x_2)\omega\|_{\infty} : (0, 0) \neq (x_1, x_2) \in (\mathcal{O}_F)^2\} > 0.$$

Applying the above version of Mahler's criterion gives the following version of the Dani correspondence.

Theorem 2.4.3 ([EGL16] Proposition 3.1). A vector $z = (z_{\sigma})_{\sigma} \in F \otimes \mathbb{R} \cong \mathbb{R}^r \times \mathbb{C}^s$ is badly approximable if and only if the trajectory

$$\omega_t(z) = \left\{ SL_2(\mathcal{O}_F) \left(\left(\begin{array}{cc} 1 & z_\sigma \\ 0 & 1 \end{array} \right) \left(\begin{array}{cc} e^{-t} & 0 \\ 0 & e^t \end{array} \right) \right)_\sigma \right\}$$

is bounded in $SL_2(\mathcal{O}_F) \setminus SL_2(F \otimes \mathbb{R})$.

Our applications of the above will be discussed in chapter 5. For completeness (and flavor), we provide proofs of the previous two results from [EGL16] when F is imaginary quadratic. **Theorem 2.4.4** (Mahler's compactness criterion). Let $\Omega \subseteq SL_2(\mathbb{C})$. The set $SL_2(\mathcal{O}) \cdot \Omega \subseteq$ $SL_2(\mathcal{O}) \setminus SL_2(\mathbb{C})$ is precompact if and only if

$$\inf\{\|Xg\|_2 : g \in \Omega, \ X = (x_1, x_2) \in \mathcal{O}^2 \setminus \{(0, 0)\}\} > 0,$$

where $||X||_2 = \sqrt{|x_1|^2 + |x_2|^2}$.

Proof of Mahler's criterion. Choose an integral basis for \mathcal{O} , say 1 and $\omega = \frac{D_F + \sqrt{D_F}}{2}$ for concretences, where D_F is the field disciminant,

$$D_F = \begin{cases} -d, & d \equiv 3 \mod 4\\ -4d, & d \equiv 1, 2 \mod 4 \end{cases}$$

We have a homomorphism

$$\phi: \mathbb{C} \to M_2(\mathbb{R}), \ z \mapsto \left(\begin{array}{cc} r & s \\ s \frac{D_F(1-D_F)}{4} & r+s D_F \end{array} \right), \ z = r+s\omega,$$

taking a complex number z to the matrix of multiplication by z in the our integral basis. This extends to a homomorphism

$$\Phi: SL_2(\mathbb{C}) \to SL_4(\mathbb{R}), \left(\begin{array}{cc} z_1 & z_2 \\ w_1 & w_2 \end{array}\right) \mapsto \left(\begin{array}{cc} \phi(z_1) & \phi(z_2) \\ \phi(w_1) & \phi(w_2) \end{array}\right),$$

with $\Phi(SL_2(\mathbb{C})) \cap SL_4(\mathbb{Z}) = \Phi(SL_2(\mathcal{O}))$. Hence we obtain a closed embedding

$$\Phi: SL_2(\mathcal{O}) \backslash SL_2(\mathbb{C}) \to SL_4(\mathbb{Z}) \backslash SL_4(\mathbb{R}).$$

One can easily verify that the bijection

$$\Psi: \mathbb{C}^2 \to \mathbb{R}^4, \ (a+b\omega, c+d\omega) \mapsto (a, b, c, d)$$

is $SL_2(\mathbb{C})$ -equivariant, i.e.

$$\Psi((a+b\omega, c+d\omega)g) = (a, b, c, d)\Phi(g), \ g \in SL_2(\mathbb{C}),$$

and that the norms $\|\Psi(\cdot)\|_2$, $\|\cdot\|_2$ are equivalent on \mathbb{C}^2 :

$$R_{+} \| \Psi(\cdot) \|_{2} \leq \| \cdot \|_{2} \leq R_{-} \| \Psi(\cdot) \|_{2},$$

where

$$R_{\pm} = \left(\frac{2}{1+|\omega|^2 \pm |1+\omega^2|}\right)^{1/2}$$

are the radii of the inscribed and circumscribed circles of the ellipse $|a + b\omega|^2 = 1$. Applying the standard version of Mahler's criterion (Theorem 2.2.1 above) to $\Phi(\Omega)$ gives the result.

Theorem 2.4.5 (Dani correspondence). For $z \in \mathbb{C}$, define

$$\Omega_z = \left\{ \begin{pmatrix} 1 & z \\ 0 & 1 \end{pmatrix} \begin{pmatrix} e^{-t} & 0 \\ 0 & e^t \end{pmatrix} : t \ge 0 \right\} \subseteq SL_2(\mathbb{C}).$$

The trajectories

$$SL_2(\mathcal{O}) \cdot \Omega_z \subseteq SL_2(\mathcal{O}) \backslash SL_2(\mathbb{C}),$$
$$\omega_z := SL_2(\mathcal{O}) \cdot \Omega_z \cdot SU_2(\mathbb{C}) \subseteq SL_2(\mathcal{O}) \backslash \mathcal{H}^3,$$

are precompact if and only if z is badly approximable.

Proof of Dani correspondence. For the proof, note that

$$\mathcal{O}^2 \cdot \Omega_z = \{ (e^{-t}q, e^t(p+qz)) : t \ge 0, \ (q,p) \in \mathcal{O}^2 \}.$$

Suppose z is badly approximable with $|q(qz+p)| \ge C'$ for all $p/q \in F$. If there exists $t \ge 0$ and $p/q \in F$ with $||(e^{-t}q, e^t(qz+p))||_2 < \sqrt{C'}$, then taking the product of the coordinates gives

$$|e^{-t}qe^{t}(qz+p)| = |q(qz+p)| < C',$$

a contradiction. Hence $\inf\{\|Xg\|_2 : g \in \Omega_z, X \in \mathcal{O}^2 \setminus \{(0,0)\}\} \geq \sqrt{C'}$ and $SL_2(\mathcal{O}) \cdot \Omega_z$ is precompact by Mahler's criterion.

If z is not badly approximable, then for every n > 0 there exists $p_n/q_n \in F$ such that $|q_n(q_nz+p_n)| < 1/n^2$. If t_n is such that $e^{-t_n}|q_n| = 1/n$, then $|e^{t_n}(q_nz+p_n)| = n|q_n(q_nz+p_n)| < 1/n$ and $||(e^{-t_n}q_n, e^{t_n}(q_nz+p_n))||_2 \le \sqrt{2}/n$. Therefore $SL_2(\mathcal{O}) \cdot \Omega_z$ is not contained in any compact set by Mahler's criterion. The result for ω_z follows from the invariance $||Xk||_2 = ||X||_2$ for $X \in \mathbb{C}^2$, $k \in$ $SU_2(\mathbb{C})$, and the fact that the projection $SL_2(\mathcal{O}) \setminus SL_2(\mathbb{C}) \to SL_2(\mathcal{O}) \setminus \mathcal{H}^3$ is proper and closed. \Box

Chapter 3

Continued fractions in \mathbb{R} and \mathbb{C}

3.1 Introduction

In this chapter we first give an overview of simple continued fractions on the real line including the relation to the geodesic flow on the modular surface (perhaps first exploited by E. Artin, [Art24]). There are innumerable books on continued fractions, including: [EW11], [Sch80], [Khi97], [Bil65], [Per54], and [Hen06]. Secondly, we give an overview of nearest integer continued fractions over the Euclidean imaginary quadratic fields, $\mathbb{Q}(\sqrt{-d})$, d = 1, 2, 3, 7, 11, which go back to A. Hurwitz [Hur87]. A few of the results are new, or at least proofs could not be found in the literature. Finally, we end with an quick summary of A. L. Schmidt's continued fractions over $\mathbb{Q}(\sqrt{-1})$ (cf. [Sch75a], [Sch82], [Nak88a], [Nak90]). These continued fractions are based on partitions of the plane associated to circle packings, which we reimagine and generalize in chapter 4.

3.2 Simple continued fractions

The integers are a Euclidean domain, with respect to the usual absolute value for instance. For a pair of integers $a, b \neq 0$ we have

$$a = ba_0 + r_0, \ 0 \le r_0 < |b|$$
$$b = r_0 a_1 + r_1, \ 0 \le r_1 < r_0$$
$$r_0 = r_1 a_2 + r_2, \ 0 \le r_2 < r_1$$

or written in matrices

$$\begin{pmatrix} a \\ b \end{pmatrix} = \begin{pmatrix} a_0 & 1 \\ 1 & 0 \end{pmatrix} \begin{pmatrix} b \\ r_0 \end{pmatrix}$$
$$= \begin{pmatrix} a_0 & 1 \\ 1 & 0 \end{pmatrix} \begin{pmatrix} a_1 & 1 \\ 1 & 0 \end{pmatrix} \begin{pmatrix} r_0 \\ r_1 \end{pmatrix}$$
$$= \begin{pmatrix} a_0 & 1 \\ 1 & 0 \end{pmatrix} \begin{pmatrix} a_1 & 1 \\ 1 & 0 \end{pmatrix} \begin{pmatrix} a_2 & 1 \\ 1 & 0 \end{pmatrix} \begin{pmatrix} r_1 \\ r_2 \end{pmatrix}$$

which if (a, b) = 1 gives

$$\left(\begin{array}{c}a\\b\end{array}\right) = \left(\begin{array}{c}a_0&1\\1&0\end{array}\right) \left(\begin{array}{c}a_1&1\\1&0\end{array}\right) \dots \left(\begin{array}{c}a_n&1\\1&0\end{array}\right) \left(\begin{array}{c}1\\0\end{array}\right)$$

when the algorithm stops. Thinking of this as an algorithm on rationals (dividing by b, r_0, r_1, \ldots in the first array or acting by fractional linear transformations in the second) we obtain an expression for a/b as a continued fraction

$$\frac{a}{b} = a_0 + \frac{1}{a_1 + \frac{1}{a_2 + \frac{1}{\dots \frac{1}{a_n}}}} =: [a_0; a_1, \dots, a_n].$$

Extending this to irrational numbers $\xi = \lfloor \xi \rfloor + \{\xi\} = a_0 + \xi_0$ gives a dynamical system

$$T: [0,1) \to [0,1), \ \xi \mapsto \{1/\xi\}$$

and infinite sequences

$$\xi_n = T^n \xi_0, \ a_{n+1} = \left\lfloor \frac{1}{T^n \xi_0} \right\rfloor, \ \begin{pmatrix} p_n & q_{n-1} \\ q_n & q_{n-1} \end{pmatrix} = \begin{pmatrix} a_0 & 1 \\ 1 & 0 \end{pmatrix} \begin{pmatrix} a_1 & 1 \\ 1 & 0 \end{pmatrix} \dots \begin{pmatrix} a_n & 1 \\ 1 & 0 \end{pmatrix}.$$

One can verify the following properties by induction or from the definitions:

$$q_n/q_{n-1} > 1 \ (n \ge 2), \ q_n \ge 2^{\frac{n-1}{2}} \ (n \ge 0), \ q_n \xi - p_n = (-1)^n \xi_0 \cdot \ldots \cdot \xi_n = \frac{(-1)^n \xi_n}{q_{n-1}\xi_n + q_n},$$

$$\frac{p_n}{q_n} = a_0 - \sum_{k=1}^n \frac{(-1)^k}{q_k q_{k-1}}, \ \xi = \frac{p_n + p_{n-1}\xi_n}{q_n + q_{n-1}\xi_n}, \ \frac{1}{q_{n+2}} \le |q_n\xi - p_n| \le \frac{1}{q_{n+1}}.$$

Hence for irrational ξ we have convergence

$$\xi = a_0 + \frac{1}{a_1 + \frac{1}{a_2 + \dots}} =: [a_0; a_1, a_2, \dots].$$

The branches of T^{-1} are all surjective and we have bijections

$$\mathbb{R} \setminus \mathbb{Q} \cong \mathbb{Z} \times \mathbb{N}^{\mathbb{N}}, \ \mathbb{Q} = \{ [a_0; a_1, \dots, a_n] : n \ge 0, \ a_n \ne 1 \text{ if } n \ge 1 \}.$$

T is the left shift on these sequences, $T([0; a_1, a_2, \ldots]) = [0; a_2, a_3, \ldots]$.



Figure 3.1: $T\xi = \{\frac{1}{\xi}\}$

3.2.1 Invertible extension and invariant measure

The map T is not invertible, but we can construct an invertible extension on $\mathcal{G} = (-\infty, -1) \times$ (0, 1)

$$\widetilde{T}: \mathcal{G} \to \mathcal{G}, \ \widetilde{T}(\eta, \xi) = (1/\eta - \lfloor 1/\xi \rfloor, 1/\xi - \lfloor 1/\xi \rfloor),$$

defined piecewise by Möbius transformations acting diagonally

$$\widetilde{T}(\eta,\xi) = (g \cdot \eta, g \cdot \xi), \quad \widetilde{T}^{-1}(\eta,\xi) = (h \cdot \eta, h \cdot \xi),$$
$$g = \begin{pmatrix} -\lfloor 1/\xi \rfloor & 1\\ 1 & 0 \end{pmatrix}, \quad h = \begin{pmatrix} 0 & 1\\ 1 & \lfloor -\eta \rfloor \end{pmatrix}.$$

We will think of $\mathcal{G} = (-\infty, -1) \times (0, 1)$ as a space of geodesics in hyperbolic 2-space, \mathcal{H}^2 , and the action of \widetilde{T} takes this space to itself piecewise by isometries $\operatorname{Isom}(\mathcal{H}^2) \cong PGL_2(\mathbb{R})$, where



Figure 3.2: Invertible extension of the (slow) Gauss map on $(-\infty, 0) \times (1, \infty) \cup (-\infty, -1) \times (0, 1)$.

 $PGL_2(\mathbb{R})$ acts on the upper half-plane by

$$\begin{pmatrix} a & b \\ c & d \end{pmatrix} \cdot z = \frac{az+b}{cz+d} \text{ or } \frac{a\overline{z}+b}{c\overline{z}+d}$$

according as the determinant ad - bc is positive or negative.

Let \mathcal{H}^2 have coordinates (x, y) with area $\frac{dxdy}{y^2}$. The top of the geodesic (η, ξ) (semi-circle from η to ξ with center on the real line x = 0) has coordinates $\left(\frac{\xi+\eta}{2}, \frac{\xi-\eta}{2}\right)$, so that hyperbolic area becomes $d\tilde{\mu}(\eta, \xi) := \frac{d\xi d\eta}{(\xi-\eta)^2}$ in these coordinates. This gives an isometry invariant measure on our space of geodesics, and since \tilde{T} is a bijection defined piecewise by isometries, $\tilde{\mu}$ is \tilde{T} -invariant. Pushing forward to the second coordinate gives a T-invariant measure on (0, 1)

$$d\mu(\xi) = d\xi \int_{-\infty}^{-1} \frac{d\eta}{(\xi - \eta)^2} = \frac{d\xi}{1 + \xi}$$

which we will normalize to a probability measure by dividing by log 2. This measure was known to Gauss, although more than 100 years passed before details and answers to some of Gauss' questions were given by Kuzmin [Kuz32]; see [Bre91] for some history.

3.2.2 Direct proof of ergodicity

(Following [Bil65].) For fixed $a_1, \ldots, a_n \in \mathbb{N}$, we have the cylinder set

$$\Delta_n = \{\psi(t) = \frac{p_n + p_{n-1}t}{q_n + q_{n-1}t} : 0 \le t < 1\}$$

which is the half-open interval between p_n/q_n and $\frac{p_n+p_{n-1}}{q_n+q_{n-1}}$ (oriented depending on the parity of n). These cylinder sets generate the Borel σ -algebra. If λ is Lebesgue measure, then we have (bar indicating conditional probability)

$$\lambda(T^{-n}[s,t) \mid \Delta_n) = \frac{\psi(t) - \psi(s)}{\psi(1) - \psi(0)} = (t-s)\frac{q_n(q_n + q_{n-1})}{(q_n + q_{n-1}s)(q_n + q_{n-1}t)} = (t-s)C,$$

where $1/2 \le C \le 2$. Hence there exists (a different) C > 0 such that

$$\frac{1}{C}\mu(A) \le \mu(T^{-n}A \mid \Delta_n) \le C\mu(A)$$

Considering T-invariant sets of positive measure, we have

$$T^{-1}A = A \Rightarrow \frac{1}{C}\mu(A) \le \mu(A \mid \Delta_n)$$

$$\mu(A) > 0 \Rightarrow \frac{1}{C}\mu(\Delta_n) \le \mu(\Delta_n \mid A)$$

$$\Rightarrow \frac{1}{C}\mu(B) \le \mu(B \mid A) \text{ for any measurable}B$$

$$A^c = B \Rightarrow \mu(A^c) = 0, \ \mu(A) = 1.$$

Hence μ is ergodic.

3.2.3 Consequences of ergodicity

We can apply the following ergodic theorem to various functions to obtain almost everywhere statistics for continued fractions. References for any ergodic theory used here include [Wal82], [Bil65], [EW11].

Theorem 3.2.1. Supose (X, T, μ) is a measure preserving system and $f \in L^1(\mu)$, then the limit

$$\lim_{N \to \infty} \frac{1}{N} \sum_{n=0}^{n-1} f \circ T^n = f^*$$

exists almost everywhere and

$$\int_X f d\mu = \int_X f^* d\mu$$

(i.e. f^* is the conditional expectation of f with respect to the σ -algebra of T-invariant sets). In particular, if the system is ergodic then f^* is constant almost everywhere,

$$f^* = \int_X f d\mu.$$

For instance:

• If f is the indicator of the interval $(\frac{1}{k+1}, \frac{1}{k})$, we get

$$\mathbb{P}(a_n = k) = \lim_{N \to \infty} \frac{1}{N} |\{i : a_i = k, \ 1 \le i \le N\}| = \frac{1}{\log 2} \int_{\frac{1}{k+1}}^{\frac{1}{k}} \frac{d\xi}{1+\xi} = \frac{1}{\log 2} \log\left(\frac{(k+1)^2}{k(k+2)}\right) + \frac{1}{\log 2} \log\left(\frac{(k+1)^2}{$$

k	1	2	3	4	5	6	7
$\mathbb{P}(a_n = k)$	41.58%	16.99%	9.31%	5.89%	4.06%	2.97%	2.27%

• Taking f to be $\sum_k \log k \cdot \mathbf{1}_{(\frac{1}{k+1},\frac{1}{k})},$ we get (almost everywhere)

$$\lim_{N \to \infty} \left(\prod_{n=1}^{N} a_n \right)^{1/N} = \exp\left(\lim_{N \to \infty} \frac{1}{N} \sum_{n=1}^{N} \log a_n \right) = \exp\left(\frac{1}{\log 2} \sum_k \int_{\frac{1}{k+1}}^{\frac{1}{k}} \frac{\log k}{1+\xi} d\xi \right)$$
$$= \prod_k \left(\frac{(k+1)^2}{k(k+2)} \right)^{\frac{\log k}{\log 2}} = 2.6854520010\dots$$

• With $f_M = \sum_{k \leq M} k \cdot \mathbf{1}_{(\frac{1}{k+1}, \frac{1}{k})}$ and taking $M \to \infty$ we get

$$\lim_{N \to \infty} \frac{1}{N} \sum_{n=1}^{\infty} a_n = \infty.$$

With a little more work we can show

$$\lim_{n \to \infty} \frac{1}{n} \log q_n = \frac{1}{\log 2} \cdot \frac{\pi^2}{12} \text{ and } \lim_{n \to \infty} \frac{1}{n} \log |\xi - p_n/q_n| = -\frac{1}{\log 2} \cdot \frac{\pi^2}{6}.$$

We have

$$\begin{pmatrix} \xi \\ 1 \end{pmatrix} = \begin{pmatrix} p_n & p_{n-1} \\ q_n & q_{n-1} \end{pmatrix} \begin{pmatrix} 1 \\ T^n \xi \end{pmatrix}, \ T^n \xi = (-1)^{n-1} \frac{q_n \xi - p_n}{q_{n-1} \xi - p_{n-1}},$$

so that

$$\prod_{k=0}^{n-1} T^k \xi = (-1)^n (q_{n-1}\xi - p_{n-1}) = |q_{n-1}\xi - p_{n-1}|.$$

Hence, from the list of properties in the first section, we have

$$\frac{1}{q_{n+1}} \le \prod_{k=0}^{n-1} T^k \xi \le \frac{1}{q_n}.$$

Taking logarithms and letting $n \to \infty$ we get

$$-\lim_{n \to \infty} \frac{1}{n} \log q_{n+1} \le \lim_{n \to \infty} \sum_{k=0}^{n-1} \log T^k \xi \le -\lim_{n \to \infty} \frac{1}{q_n} \log q_n,$$

so that

$$\lim_{n \to \infty} \frac{1}{n} \log q_n = -\frac{1}{\log 2} \int_0^1 \frac{\log \xi}{1+\xi} d\xi = \frac{1}{\log 2} \sum_{k=0}^\infty (-1)^{k+1} \int_0^1 \xi^k \log \xi \, d\xi$$
$$= \frac{1}{\log 2} \sum_{k=1}^\infty \frac{(-1)^{k+1}}{k^2} = \frac{1}{\log 2} \cdot \frac{\pi^2}{12}.$$

Since $\frac{1}{q_nq_{n+2}} \le |\xi - \frac{p_n}{q_n}| \le \frac{1}{q_nq_{n+1}}$, the above gives

$$\lim_{n \to \infty} \frac{1}{n} \log |\xi - p_n/q_n| = -\frac{1}{\log 2} \cdot \frac{\pi^2}{6}.$$

Finally, a result on the normalized error $\theta_n(\xi) = q_n |q_n \xi - p_n|$ (assuming \widetilde{T} is ergodic).

Proposition 3.2.2 ([Hen06] Theorem 4.1). For μ almost every ξ , we have

$$\lim_{N \to \infty} \frac{1}{N} |\{1 \le n \le N : \theta_n(\xi) \le t\}| = \begin{cases} \frac{t}{\log 2} & 0 \le t \le 1/2 \\ 1 + \frac{1 - t + \log t}{\log 2} & 1/2 \le t \le 1 \\ 1 & t \ge 1 \end{cases}$$

Proof. Note that

$$\widetilde{T}^{n}(-\infty,\xi) = (-q_{n}/q_{n-1},T^{n}\xi) = (-[a_{n};a_{n-1},\ldots,a_{1}],[0;a_{n+1},a_{n+2},\ldots])$$
$$\theta_{n}(\xi) = \frac{1}{1/T^{n}\xi + q_{n-1}/q_{n}} = \frac{1}{[a_{n+1};a_{n+2},\ldots] + [0;a_{n},\ldots,a_{1}]},$$

so that $\theta_n(\xi) \leq t$ iff $\frac{1}{1/\xi'-1/\eta'} \leq t$ where $(\eta',\xi') = \widetilde{T}^n(-\infty,\xi)$. Let $\mathcal{G}(c) = \{(\eta,\xi) \in \mathcal{G} : 1/\xi - 1/\eta \geq c\}$ Then for $\epsilon > 0$ and n large, we have

$$\widetilde{T}^n(\eta,\xi) \in \mathcal{G}(1/t+\epsilon) \Rightarrow \widetilde{T}^n(-\infty,\xi) \in \mathcal{G}(1/t) \Rightarrow \widetilde{T}^n(\eta,\xi) \in \mathcal{G}(1/t-\epsilon).$$

The measure of $\mathcal{G}(c)$ with respect to $\frac{d\xi d\eta}{(\xi - \eta)^2 \log 2}$ for $c \ge 1$ is

$$\begin{cases} \frac{1}{\log 2} \left(1 - \frac{1}{c} + \log 2 - \log c \right) & 1 \le c \le 2\\ \frac{1}{c \log 2} & c \ge 2 \end{cases}$$

,

which gives the result when t = 1/c. See Figure 3.3 for a graph of the distribution.

3.2.4 Geodesic flow on the modular surface and continued fractions

(Following chapter 9 of [EW11] in spirit if not detail.) The group $SL_2(\mathbb{R})$ acts transitively on the upper half-plane by fractional linear transformations, and the stabilizer of z = i is $SO_2(\mathbb{R})$. Hence, as a homogeneous space, $\mathcal{H}^2 \cong SL_2(\mathbb{R})/SO_2(\mathbb{R})$. Moreover $SL_2(\mathbb{R})$ acts as orientation perserving isometries and in fact $PSL_2(\mathbb{R}) \cong \text{Isom}^+(\mathcal{H}^2)$.



Figure 3.3: Distribution of the normalized error for almost every ξ .

We can identify $PSL_2(\mathbb{R})$ with $T^1(\mathcal{H}^2)$, the unit tangent bundle, as follows. Let $(z, v) \in$ $T^1(\mathcal{H}^2) \subseteq \mathcal{H}^2 \times \mathcal{C}$, where z = x + yi is in the upper half-plane and v has norm 1 in the hyperbolic metric, i.e. if $v = v_1 + v_2 i$, then $\langle v, v \rangle_z := v_1 v_2 / y^2 = 1$. Define the derivative action of $SL_2(\mathbb{R})$ on $T^1(\mathcal{H}^2)$ by

$$g \cdot (z, v) = (g(z), g'(z)v) = \left(\frac{az+b}{cz+d}, \frac{v}{(cz+d)^2}\right)$$

One can verify that this action is isometric and transitive with kernel ± 1 , so that we get the identification $PSL_2(\mathbb{R}) \cong T^1(\mathcal{H}^2)$.

Under the NAK decomposition, with

$$p = \begin{pmatrix} \sqrt{y} & x/\sqrt{y} \\ 0 & 1/\sqrt{y} \end{pmatrix} \in NA, \ k = \begin{pmatrix} \cos(\theta/2) & \sin(\theta/2) \\ -\sin(\theta/2) & \cos(\theta/2) \end{pmatrix} \in K,$$

we have $pk(i,i) = (x + yi, yie^{i\theta})$.

Associating the identity with the point $(z, v) = (i, i) \in T^1(\mathcal{H}^2)$, the geodesic flow Φ_t : $T^1(\mathcal{H}^2) \to T^1(\mathcal{H}^2)$ is right multiplication by $\begin{pmatrix} e^{t/2} & 0 \\ 0 & e^{-t/2} \end{pmatrix}$. It takes a point (z, v) along a unit speed geodesic in the direction v for time t, as can be checked along the imaginary axis and extended via the isometric action.

The group $\Gamma = SL_2(\mathbb{Z})$ is a discrete subgroup of $SL_2(\mathbb{R})$ and the quotient $\Gamma \setminus SL_2(\mathbb{R})$ has finite volume, equivalently, the quotient $\Gamma \setminus \mathcal{H}^2$ has finite hyperbolic area (namely $\pi/3$). We identify $\Gamma \setminus SL_2(\mathbb{R})$ with the unit tangent bundle of the modular surface.

We now want to relate continued fractions (or its invertible extension) with the geodesic flow on the modular surface. There is a slight complication because continued fractions are defined over $GL_2(\mathbb{Z})$. Consider the following subsets of $T^1(\mathcal{H}^2)$

$$\mathcal{C}^{+} = \{(z, v) \in \overrightarrow{\eta\xi} : (\eta, \xi) \in \mathcal{G}, \ z \in i\mathbb{R}\},\$$
$$\mathcal{C}^{-} = \{(z, v) \in \overrightarrow{(-\eta)(-\xi)} : (\eta, \xi) \in \mathcal{G}, \ z \in i\mathbb{R}\},\$$
$$\mathcal{C} = \mathcal{C}^{+} \cup \mathcal{C}^{-},\$$

i.e. those points and directions of the intersection of elements of \mathcal{G} with the imaginary axis (and their reflected images). Let $\pi : T^1(\mathcal{H}^2) \to T^1(\Gamma \setminus \mathcal{H}^2)$ be the natural projection.

Proposition 3.2.3. If $(z, v) = (\eta, \xi) \in \pi(\mathcal{C}^+)$, the next return of the geodesic flow to $\pi(\mathcal{C})$ is in $\pi(\mathcal{C}^-)$ with coordinates $-\widetilde{T}(\eta, \xi)$, and similarly for $(z, v) = (-\eta, -\xi) \in \pi(\mathcal{C}^-)$, the next return of the geodesic flow to $\pi(\mathcal{C})$ is in $\pi(\mathcal{C}^+)$ with coordinates $\widetilde{T}(\eta, \xi)$. In other words, the map

$$S: \mathcal{G} \cup -\mathcal{G}, \ S(\eta, \xi) = -\widetilde{T}(\eta, \xi), \ S(-\eta, -\xi) = \widetilde{T}(\eta, \xi),$$

is the first return of the geodesic flow to the cross section $\pi(\mathcal{C})$.

Proof. Proof by picture. Applying -1/z to $(\eta, \xi) \in (-\infty, -1) \times (0, 1)$ gives $(-1/\eta, -1/\xi) \in (0, 1) \times (-\infty, -1)$. Following the geodesic flow to the next intersection with $\pi(\mathcal{C})$ corresponds to translation by $\lfloor 1/\xi \rfloor$ where we end up in $-\mathcal{G}$ with coordinates $(\lfloor 1/\xi \rfloor - 1/\eta, \lfloor 1/\xi \rfloor - 1/\xi) = -\widetilde{T}(\eta, \xi)$. Similarly for the other case.

To complete the picture of how continued fractions fit into the geodesic flow, we need to consider the return time $r(\eta, \xi)$, $r(-\eta, -\xi)$ to the cross section. Here is a more general construction, the special flow under a function.

Proposition 3.2.4. Suppose (X, T, μ) is a measure preserving system, and $f : X \to (0, \infty)$ is μ -measurable. Let $X_f = \{(x, s) : 0 \le s < f(x)\}$ and for each $t \ge 0$ let $\phi_t : X_f \to X_f$ be defined by

$$\phi_t(x,s) = \left(T^n x, s+t - \sum_{k=0}^{n-1} f(T^k x)\right)$$



Figure 3.4: Visualizing S as first return of the geodesic flow to $\pi(\mathcal{C})$.

where n is the least non-negative integer such that $0 \leq s + t < \sum_{k=0}^{n} f(T^{k}x)$. Let $\mu_{f} = \mu \times \lambda$ restricted to X_{f} . Then:

- μ_f is ϕ_t -invariant for all $t \ge 0$,
- $\mu_f(X_f) < \infty \Leftrightarrow f \in L^1(X,\mu),$
- If $\mu_f(X_f) < \infty$, then (X, T, μ) is ergodic \Leftrightarrow the flow $\{\phi_t\}_{t \ge 0}$ is ergodic.

Let $r : \mathcal{G} \cup -\mathcal{G} \to (0, \infty)$ be the return time of the associated $(z, v) \in \pi(\mathcal{C})$ to $\pi(\mathcal{C}), X_r = (\mathcal{G} \cup -\mathcal{G})_r, S(\eta, \xi)$ as above, and ϕ the special flow associated to r. Let $\Sigma = SL_2(\mathbb{Z}) \setminus SL_2(\mathbb{R}) / SO_2(\mathbb{R})$ be the modular surface, $T^1(\Sigma) = SL_2(\mathbb{Z}) \setminus SL_2(\mathbb{R})$ its unit tangent bundle, and Φ the geodesic flow (right multiplication by diag $(e^{t/2}, e^{-t/2})$). We have

Proposition 3.2.5. The following diagram commutes

$$\begin{array}{ccc} X_r & \xrightarrow{\phi_t} & X_r \\ & & \downarrow \\ & & \downarrow \\ T^1(\Sigma) & \xrightarrow{\Phi_t} & T^1(\Sigma) \end{array}$$

where the arrows on the left and right are $((y, x), s) \mapsto \Phi_s(z, v)$. Moreover, the measure μ_r for the special flow under the return time (constructed from Gauss measure and Lebesgue measure) is the pullback (up to a multiplicative constant) of the Φ -invariant measure $\frac{d\theta dx dy}{y^2}$ on the unit tangent bundle (Haar measure on $PSL_2(\mathbb{R})$).
24

Proof. That the above commutes follows from our earlier discussion, so we now consider the measures involved. We have two coordinate systems on $PSL_2(\mathbb{R}) \cong T^1(\mathcal{H}^2)$, thinking of a point z = x + yi and direction $v = iye^{i\theta}$ at z,

$$g(x, y, \theta) = \begin{pmatrix} \sqrt{y} & x/\sqrt{y} \\ 0 & 1/\sqrt{y} \end{pmatrix} \begin{pmatrix} \cos(\theta/2) & \sin(\theta/2) \\ -\sin(\theta/2) & \cos(\theta/2) \end{pmatrix},$$

discussed above, and another obtained by associating to (z, v) the geodesic (η, ξ) it determines and the distance/time one must go from the "top" of the geodesic

$$g(\eta,\xi,t) = \frac{1}{\sqrt{|\eta-\xi|}} \begin{pmatrix} \max\{\xi,\eta\} & \min\{\xi,\eta\} \\ 1 & 1 \end{pmatrix} \begin{pmatrix} 0 & 1 \\ -1 & 0 \end{pmatrix}^{\epsilon} \begin{pmatrix} e^{t/2} & 0 \\ 0 & e^{-t/2} \end{pmatrix},$$

with $\epsilon = 0$ if $\xi > \eta$ and $\epsilon = 1$ otherwise. One can think of this as taking the geodsic $\overrightarrow{0\infty}$ with marked point *i*, mapping it to the geodesic $\overrightarrow{\eta\xi}$ with marked point at the top, and flowing for the required time. We have an invariant measure on each of these

$$d\mu dt = \frac{d\eta d\xi}{(\xi - \eta)^2} dt, \ \frac{dxdy}{y^2} d\theta$$

and we would like to show that these are the same (perhaps up to a constant). Assume $\xi > \eta$ (the other case being similar). We have

$$\frac{1}{\sqrt{|\eta-\xi|}} \begin{pmatrix} \xi & \eta \\ 1 & 1 \end{pmatrix} \begin{pmatrix} e^{t/2} & 0 \\ 0 & e^{-t/2} \end{pmatrix} \cdot (i,i) = (x+yi, iye^{i\theta})$$

where

$$x = \frac{\xi e^t + \eta e^{-t}}{e^t + e^{-t}}, \ y = \frac{\xi - \eta}{e^t + e^{-t}}, \ \theta = \arctan\left(\frac{1}{\sinh t}\right).$$

Some computation shows

$$\frac{dxdyd\theta}{y^2} = \left(\frac{e^t + e^{-t}}{\xi - \eta}\right)^2 \begin{vmatrix} \frac{e^t}{e^t + e^{-t}} & \frac{e^{-t}}{e^t + e^{-t}} & *\\ \frac{1}{e^t + e^{-t}} & \frac{-1}{e^t + e^{-t}} & *\\ 0 & 0 & \frac{-1}{\cosh t} \end{vmatrix} = 2\frac{d\eta d\xi dt}{(\xi - \eta)^2},$$

as desired.

We can compute the return time from the above,

$$r(\pm\eta,\pm\xi) = \frac{1}{2}\log\left(\frac{1-\eta\lfloor 1/\xi\rfloor}{1-\xi\lfloor 1/\xi\rfloor}\right),\,$$

and its integral must be

$$\int_{\mathcal{G}\cup-\mathcal{G}} r(\eta,\xi) \frac{d\xi d\eta}{(\xi-\eta)^2} = \frac{\pi^2}{3}$$

since the total volume of $T^1(\Sigma)$ is $2\pi^2/3$. (I couldn't compute this integral, and neither could Mathematica, but it agrees numerically within the estimated error.)

3.2.5 Mixing of the geodesic flow

Mixing of the geodesic flow on the modular surface is implied by the following "decay of matrix coefficients" theorem, applied to the Hilbert space $\mathcal{H} = L_0^2(SL_2(\mathbb{Z}) \setminus SL_2(\mathbb{R}))$ (L^2 functions f with $\int f = 0$) and $g = \text{diag}(e^{t/2}, e^{-t/2})$.

Theorem 3.2.6 (Howe-Moore, cf. [Mor15], §11.2, or [Zim84], Theorem 2.2.20). Suppose ρ : $SL_2(\mathbb{R}) \to \mathcal{U}(\mathcal{H})$ is a unitary representation with no non-zero fixed vectors, i.e. $\{v \in \mathcal{H} : \rho(G)v = v\} = \{0\}$. Then for all $v, w \in \mathcal{H}, \langle \rho(g)v, w \rangle \to 0$ as $g \to \infty$ (i.e. for any $\epsilon > 0$, there exists $K \subseteq G$ compact such that for $g \notin K, \langle \rho(g)v, w \rangle < \epsilon$).

Proof. We first note that any $g \in SL_2(\mathbb{R})$ can be written uniquely as KA^+K (Cartain decomposition) where K is $SO_2(\mathbb{R})$ and A^+ is the collection of diag (a, a^{-1}) , a > 0. (Proof: diagonalize the quadratic form $||gx||^2$.) Let $g_n \to \infty$, $g_n = k_n a_n l_n$, and for $v, w \in \mathcal{H}$, let \tilde{v} , \tilde{w} be weak limits of $\rho(l_n)v, \rho(k_n^{-1})w$. We have

$$\begin{aligned} \langle \rho(k_n a_n l_n) v, w \rangle &- \langle \rho(a_n) \tilde{v}, \tilde{w} \rangle \\ &= \langle \rho(a_n) (\rho(l_n) v - \tilde{v}), \rho(k_n^{-1}) w \rangle + \langle \rho(a_n), \rho(k_n^{-1}) w - \tilde{w} \rangle \to 0. \end{aligned}$$

Therefore we may assume $g_n = a_n = \text{diag}(t_n, 1/t_n)$ with $t_n \to \infty$.

Let

$$u_s^+ = \begin{pmatrix} 1 & s \\ 0 & 1 \end{pmatrix}, \ u_s^- = \begin{pmatrix} 1 & 0 \\ s & 1 \end{pmatrix}.$$

Then

$$a_n^{-1}u_s^+a_n = u_{s/t_n^2}^+ \to 1, \ a_n u_s^- a_n^{-1} = u_{s/t_n^2}^- \to 1.$$

Let $v, w \in \mathcal{H}$, and let E be a weak limit of $\rho(a_n)$ (diagonal argument on an orthonormal basis). We want to show that E = 0 (in particular, $\langle \rho(a_n)v, w \rangle \rightarrow \langle Ev, w \rangle = 0$). We have

$$\langle \rho(u_s^+)Ev, w \rangle = \lim_n \langle \rho(u_s^+)\rho(a_n)v, w \rangle = \lim_n \langle \rho(a_n)\rho(a_n^{-1}u_s^+a_n)v, w \rangle = \langle Ev, w \rangle,$$

and similarly $\langle \rho(u_s^-)E^*v, w \rangle = \langle E^*v, w \rangle$. Also, $\langle \rho(a_n)v, \rho(a_m)v \rangle = \langle \rho(a_m^{-1})v, \rho(a_n^{-1})w \rangle$ (since A is Abelian), so that $EE^* = E^*E$, i.e. E is normal. If $E \neq 0$, then $EE^* \neq 0$ and there is some $v \in \mathcal{H}$ such that $w = E^*Ev = EE^*v \neq 0$. This w is invariant under $\{u_s^{\pm}\}_s$, which generates $SL_2(\mathbb{R})$, contradicting the assumption that π has no non-zero invariant vectors. Hence E = 0.

3.3 Nearest integer continued fractions over the Euclidean imaginary quadratic fields

A natural generalization of continued fractions to complex numbers over appropriate discrete subrings \mathcal{O} of \mathbb{C} , in particular over $\mathbb{Z}[\sqrt{-1}]$ and $\mathbb{Z}\begin{bmatrix}\frac{1+\sqrt{-3}}{2}\end{bmatrix}$, was introduced by A. Hurwitz, [Hur87]. Let $K = K_d = \mathbb{Q}(\sqrt{-d}), d > 0$ a square-free integer, be an imaginary quadratic field and $\mathcal{O} = \mathcal{O}_d$ the ring of integers of K. For d = 1, 2, 3, 7, 11 the \mathcal{O}_d are Euclidean with respect to the usual norm $|z|^2 = z\overline{z}$, noting that the collection of disks $\{z \in \mathbb{C} : |z - r| < 1\}_{r \in \mathcal{O}}$ cover the plane (Figure 3.5), and in fact are the only d for which \mathcal{O}_d is Euclidean with respect to any function (cf. [Lem95] §4). Consider the open Voronoï cell for $\mathcal{O}_d \subseteq \mathbb{C}$, the collection of points closer to zero than to any other lattice point, along with a subset \mathcal{E} of the boundary, so that we obtain a strict fundamental domain for the additive action of \mathcal{O} on \mathbb{C} ,

$$V = V_d = \{ z \in \mathbb{C} : |z| < |z - r|, \ r \in \mathcal{O} \} \cup \mathcal{E}, \ \mathcal{E} \subseteq \partial V.$$

For the Euclidean values of d, and only for these values, V_d is contained in the open unit disk. The regions V_d are rectangles for d = 1, 2 and hexagons for d = 3, 7, 11; see Figure 3.5. For $z \in \mathbb{C}$, we denote by $\lfloor z \rfloor \in \mathcal{O}$ and $\{z\} \in V$ the nearest integer and remainder, uniquely satisfying

$$z = \lfloor z \rfloor + \{z\}.$$

We now restrict ourselves to Euclidean K to describe the continued fraction algorithm and applications.

We have an almost everywhere defined map $T = T_d : V_d \to V_d$ given by $T(z) = \{1/z\}$. For $z \in \mathbb{C}$ define sequences $a_n \in \mathcal{O}, z_n \in V$, for $n \ge 0$:

$$a_{0} = \lfloor z \rfloor, z_{0} = z - a_{0} = \{z\},$$
$$a_{n} = \lfloor \frac{1}{z_{n-1}} \rfloor, \ z_{n} = \left\{ \frac{1}{z_{n-1}} \right\} = \frac{1}{z_{n-1}} - a_{n} = T^{n}(z_{0})$$

In this way, we obtain a continued fraction expansion for $z \in \mathbb{C}$,

$$z = a_0 + \frac{1}{a_1 + \frac{1}{a_2 + \dots}} =: [a_0; a_1, a_2, \dots],$$

where the expansion is finite for $z \in K$. The convergents to z will be denoted by

$$\frac{p_n}{q_n} = [a_0; a_1, \dots, a_n],$$

where p_n , q_n are defined by

$$\begin{pmatrix} p_n & p_{n-1} \\ q_n & q_{n-1} \end{pmatrix} = \begin{pmatrix} a_0 & 1 \\ 1 & 0 \end{pmatrix} \cdots \begin{pmatrix} a_n & 1 \\ 1 & 0 \end{pmatrix}.$$

Here are a few easily verified algebraic properties that will be used below:

$$q_n z - p_n = (-1)^n z_0 \dots z_n, \ z = \frac{p_n + z_n p_{n-1}}{q_n + z_n q_{n-1}},$$
$$z - \frac{p_n}{q_n} = \frac{(-1)^n}{q_n^2 (z_n^{-1} + q_{n-1}/q_n)}, \ \frac{q_n}{q_{n-1}} = a_n + \frac{q_{n-2}}{q_{n-1}}.$$

The first equality proves convergence $p_n/q_n \to z$ for irrational z and gives a rate of convergence exponential in n. A useful parameter is $\rho = \rho_d$, the radius of the smallest circle around zero containing V_d ,

$$\rho_d = \frac{\sqrt{1+d}}{2}, \ d = 1, 2, \ \rho_d = \frac{1+d}{4\sqrt{d}}, \ d = 3, 7, 11.$$

We note that $|a_n| \ge 1/\rho_d$ for $n \ge 1$, which is easily verified for each d.

Taking the transpose of the matrix expression above, we have the equality

$$\frac{q_n}{q_{n-1}} = a_n + \frac{1}{a_{n-1} + \frac{1}{\dots + \frac{1}{a_1}}} \stackrel{alg.}{=} [a_n; a_{n-1}, \dots, a_1]$$

as rational numbers (indicated by the overset "alg."), but this does not hold at the level of continued fractions, i.e. the continued fraction expansion of q_n/q_{n-1} is not necessarily $[a_n; a_{n-1}, \ldots, a_1]$. See Figure 3.6 for the distribution of q_{n-1}/q_n , for 5000 random numbers and $1 \le n \le 10$, over $\mathbb{Q}(\sqrt{-1})$ and $\mathbb{Q}(\sqrt{-3})$. The bounds $|q_{n+2}/q_n| \ge 3/2$ are proved in [Hen06] and [Dan17] for d = 1 and 3 respectively.



Figure 3.5: ∂V and translates (blue), $\partial(V^{-1})$ (red), and unit circle (black) for d = 1, 2, 3, 7, 11.

Monotonicity of the denominators q_n was shown by Hurwitz [Hur87] for d = 1, 3, Lunz [Lun36] for d = 2, and stated without proof in [Lak73] for d = 1, 2, 3, 7, 11. As this is a desirable property to establish, we outline the proof for the cases d = 7, 11 in an appendix. The proofs are unenlightening and follow the outline for the simpler cases d = 1, 3 in [Hur87].

Proposition 3.3.1. For any $z \in \mathbb{C}$, the denominators of the convergents p_n/q_n are strictly increasing in absolute value, $|q_{n-1}| < |q_n|$.

Proof. See appendix 3.3.1.

For each of the Euclidean imaginary quadratic fields K there is a constant C > 0 such that



Figure 3.6: The numbers q_{n-1}/q_n and q_n/q_{n-1} , $1 \le n \le 10$, for 5000 randomly chosen z over $\mathbb{Q}(\sqrt{-1})$ (left) and $\mathbb{Q}(\sqrt{-3})$ (right).

for any $z \in \mathbb{C}$ there are infinitely many solutions $p/q \in K$, (p,q) = 1 to

$$|z - p/q| \le C/|q|^2,\tag{\dagger}$$

by a pigeonhole argument for instance (cf. [EGM98] Chapter 7, Proposition 2.6). The smallest such C are $1/\sqrt{3}$, $1/\sqrt{2}$, $1/\sqrt[4]{13}$, $1/\sqrt[4]{8}$, and $2/\sqrt{5}$ for d = 1, 2, 3, 7, 11 respectively (for references, see the Introduction to [Vul95b], and simple geometric proofs for d = 1, 2 can be found in chapter 4 of this document). We can obtain rational approximations with a specific C satisfying inequality (†) using the nearest integer algorithms described above. The best constants coming from the nearest integer convergents, $\sup_{z,n} \{|q_n|^2|z - p_n/q_n|\}$, can be found in Theorem 1 of [Lak73].

Proposition 3.3.2. For $z \in \mathbb{C} \setminus K$, the convergents p_n/q_n satisfy

$$|z - p_n/q_n| \le \frac{1}{(1/\rho - 1)|q_n|^2},$$

i.e. we can take $p/q = p_n/q_n$ and $C = \frac{\rho}{1-\rho}$ in the inequality (†).

Proof. Using simple properties of the algorithm and the bounds $1/z_n \in V^{-1}$, $|q_{n-1}/q_n| \leq 1$, we have

$$|z - p_n/q_n| = \frac{1}{|q_n|^2 |z_n^{-1} + q_{n-1}/q_n|} \le \frac{1}{|q_n|^2 (1/\rho - 1)}.$$

We say z is badly approximable over K if there is a C' > 0 such that

$$|z - p/q| \ge C'/|q|^2$$

for all $p/q \in K$, i.e. z is badly approximable if the exponent of two on |q| is the best possible in the inequality (†). In chapter 5, we will show that the badly approximable numbers are characterized by the boundedness of the partial quotients in the nearest integer continued fraction expansion, analogous to the well-known fact for simple continued fractions over the real numbers. To this end, we need to be able to compare any rational approximation of z to those coming from the continued fraction expansion. While the convergents are not necessarily the best approximations (e.g. [Lak73]), they aren't so bad.

Lemma 3.3.3. There are effective constants $\alpha = \alpha_d > 0$ such that for any irrational z with convergents p_n/q_n and rational p/q with $|q_{n-1}| < |q| \le |q_n|$ we have

$$|q_n z - p_n| \le \alpha |qz - p|.$$

Proof. Write p/q in terms of the convergents p_n/q_n , p_{n-1}/q_{n-1} for some $s, t \in \mathcal{O}$

$$\begin{pmatrix} p \\ q \end{pmatrix} = \begin{pmatrix} p_n & p_{n-1} \\ q_n & q_{n-1} \end{pmatrix} \begin{pmatrix} s \\ t \end{pmatrix} = \begin{pmatrix} p_n s + p_{n-1} t \\ q_n s + q_{n-1} t \end{pmatrix}$$

If s = 0, then $p/q = p_{n-1}/q_{n-1}$, impossible by the assumption $|q_{n-1}| < |q|$. If t = 0, then $p/q = p_n/q_n$ and the result is clear with $\alpha = 1$. We may therefore assume $|s|, |t| \ge 1$. We have

$$\left|z - \frac{p}{q}\right| \ge \left|\left|\frac{p_n}{q_n} - \frac{p}{q}\right| - \left|z - \frac{p_n}{q_n}\right|\right| = \left|\left|\frac{t}{qq_n}\right| - \left|z - \frac{p_n}{q_n}\right|\right|,$$

noting that $t = (-1)^n (pq_n - p_n q)$ by inverting the matrix relating p, q, s, and t.

Define δ by $|t| = \delta |q_n|^2 |z - p_n/q_n|$, so that

$$\frac{1}{|\delta - |q/q_n||}|qz - p| \ge |q_n z - p_n|$$

If $\delta > 1$, then we have our α . A lower bound for δ is

$$\delta = \frac{|t|}{|q_n|^2 |z - p_n/q_n|} \ge |t| \inf_{z,n} \{ (|q_n| |q_n z - p_n|)^{-1} \}.$$

The infimum above is calculated in [Lak73], Theorem 1, where it is found to be

$$\begin{pmatrix}
1 & d = 1, \\
\sqrt{\frac{486 - \sqrt{3}}{786}} = 0.78493... & d = 2,
\end{cases}$$

$$\inf_{z,n} \{ (|q_n||q_n z - p_n|)^{-1} \} = \begin{cases} \sqrt{\frac{7 + \sqrt{21}}{7}} = 1.28633\dots & d = 3, \\ \sqrt{\frac{2093 - 9\sqrt{21}}{2408}} = 0.92307\dots & d = 7, \end{cases}$$

$$\sqrt{\frac{30-8\sqrt{5}-5\sqrt{11}+3\sqrt{55}}{50}} = 0.59627\dots d = 11.$$

The smallest integers of norm greater than one in \mathcal{O}_d have absolute values of $\sqrt{2}$ (for d = 1, 2, 7) and $\sqrt{3}$ (for d = 3, 11). Multiplying these potential values of |t| by the above constants gives values of δ greater than one, so that $|\delta - |q/q_n|| \ge |\delta - 1|$ is bounded away from zero. Hence we are left to explore those rationals p/q with |t| = 1.

For general t we have

$$qz - p = q \frac{p_n + z_n p_{n-1}}{q_n + z_n q_{n-1}} - p = (q_n s + tq_{n-1}) \frac{p_n + z_n p_{n-1}}{q_n + z_n q_{n-1}} - (p_n s + tp_{n-1})$$
$$= \frac{(-1)^n (sz_n - t)}{q_n + z_n q_{n-1}}$$

and

$$q_n z - p_n = \frac{(-1)^n z_n}{q_n + z_n q_{n-1}},$$

and we want $\alpha > 0$ such that

$$|q_n z - p_n| \le \alpha |qz - p|.$$

Substituting the above we have

$$|q_n z - p_n| \le \alpha |qz - p| \iff \frac{|z_n|}{|q_n + z_n q_{n-1}|} \le \alpha \frac{|z_n s - t|}{|q_n + z_n q_{n-1}|}$$
$$\iff \frac{1}{|s - t/z_n|} \le \alpha.$$

If $|s - t/z_n| < 1/2$ and |t| = 1, then $s/t = a_{n+1}$ since $s/t \in \mathcal{O}$ is the nearest integer to $1/z_n$. However (with $|q_{n-1}| < |q| \le |q_n|$, $q = sq_n + tq_{n-1}$),

$$\left|\frac{q_{n+1}}{q_n}\right| = \left|a_{n+1} + \frac{q_{n-1}}{q_n}\right| = \left|\frac{s}{t} + \frac{q_{n-1}}{q_n}\right| = \left|s + t\frac{q_{n-1}}{q_n}\right| = \left|\frac{q}{q_n}\right| \le 1,$$

and we obtain a contradiction if $|s - t/z_n| < 1/2$ and |t| = 1. Hence when |t| = 1 we can take $\alpha = 2$.

In summary, we can take

$$\alpha_d = \begin{cases} 2.41421\dots & d=1\\ 9.08592\dots & d=2\\ 2 & d=3\\ 3.27419\dots & d=7\\ 30.51490\dots & d=11 \end{cases}$$

taking the maximum of 2 (covering the case |t| = 1) and the bound on $\frac{1}{\delta - |q/q_n|}$ for |t| > 1.

No attempt was made to optimize the value of α in the lemma. The above result for d = 1, 3and $\alpha = 1$ is contained in Theorem 2 of [Lak73]. Another proof for d = 1 and $\alpha = 5$ is Theorem 5.1 of [Hen06], and a proof for d = 3, $\alpha = 2$ can be found in [Dan17].

3.3.1 Appendix: Monotonicity of denominators for d = 7, 11

The purpose of this appendix is to prove Proposition 3.3.1 for d = 7, 11. Proofs for d = 1, 3 are found in [Hur87] and a proof for d = 2 in [Lun36] (§VII, Satz 11). Monotonicity for d = 7, 11 was stated in [Lak73] without proof (for reasons obvious to anyone reading what follows). All of the proofs follow the same basic outline, with d = 11 being the most tedious.

Proof of Proposition 3.3.1. For the purposes of this proof, define $k_n = q_n/q_{n-1}$; we will show $|k_n| > 1$. We will also use the notation $B_t(r)$ for the ball of radius t centered at $r \in \mathcal{O}$. When n = 1 we have $|k_1| = |a_1| \ge 1/\rho > 1$. The recurrence $k_n = a_n + 1/k_{n-1}$ is immediate from the definitions. We argue by contradiction following [Hur87]. Suppose n > 1 is the smallest value for which $|k_n| \le 1$ so that $|k_i| > 1$ for $1 \le i < n$. Then

$$|a_n| = |k_n - 1/k_{n-1}| < 2.$$

For small values of a_i , those for which $(a_i+V)\cap V^{-1}\cap (\mathbb{C}\setminus V^{-1})\neq \emptyset$, the values of a_{i+1} are restricted (cf. Figure 3.5). More generally, there are arbitrarily long "forbidden sequences" stemming from these small values of a_i . We will use some of the forbidden sequences that arise in this fashion to show that the assumption $k_n < 1$ leads to a contradiction.

• (d = 7) The only allowed values of a_n for which $|a_n| < 2$ are $a_n = \frac{\pm 1 \pm \sqrt{-7}}{2}$. By symmetry, we suppose $a_n = \frac{1+\sqrt{-7}}{2} =: \omega$ without loss of generality. It follows that $k_n \in B_1(\omega) \cap B_1(0)$. Subtracting $\omega = a_n$, we see that $1/k_{n-1} \in B_1(0) \cap B_1(-\omega)$. Applying 1/z then gives $k_{n-1} \in B_1(\omega - 1) \setminus B_1(0)$. Since $k_{n-1} = a_{n-1} + 1/k_{n-2}$, the only possible values for a_{n-1} are $\omega, \omega - 1, \omega - 2, 2\omega - 1$, and $2\omega - 2$. One can verify that the two-term sequences

a_i	a_{i+1}	
$\omega - 2$	ω	
$\omega - 1$	ω	
ω	ω	

are forbidden, so that $a_{n-1} = 2\omega - 1$ or $2\omega - 2$. We now have either $k_{n-1} \in B_1(2\omega - 1) \cap B_1(\omega - 1)$ if $a_{n-1} = 2\omega - 1$, or $k_{n-1} \in B_1(2\omega - 2) \cap B_1(\omega - 1)$ if $a_{n-1} = 2\omega - 2$. Subtracting a_{n-1} and applying 1/z shows that $a_{n-2} = 2\omega - 1$ or $2\omega - 2$ if $a_{n-1} = 2\omega - 1$, and $a_{n-2} = 2\omega$ or $2\omega - 1$ if $a_{n-1} = 2\omega - 2$.

Continuing shows that for $i \leq n-1$

$$k_i \in (B_1(2\omega - 2) \cup B_1(2\omega - 1) \cup B_1(2\omega)) \cap (B_1(\omega - 1) \cup B_1(\omega)),$$

the green region on the left in Figure 3.7. This is impossible; for instance $k_1 = a_1 \in \mathcal{O}$ but the region above contains no integers.

• (d = 11) The only allowed values of a_n for which $|a_n| < 2$ are $\frac{\pm 1 \pm \sqrt{-11}}{2}$. By symmetry, we suppose $a_n = \frac{1+\sqrt{-11}}{2} =: \omega$ without loss of generality. Hence $k_n \in B_1(0) \cap B_1(\omega)$. Subtracting a_n and applying 1/z shows that $k_{n-1} \in B_{1/2}(\frac{\omega-1}{2}) \setminus B_1(0)$ and $a_{n-1} = \omega - 1$ or ω . If $a_{n-1} = \omega - 1$, subtracting, applying 1/z and using the three-term forbidden sequences

a_i	a_{i+1}	a_{i+2}	
$\omega - 1$	$\omega - 1$	ω	
ω	$\omega - 1$	ω	
$\omega + 1$	$\omega - 1$	ω	

shows that $a_{n-2} = 2\omega$ or $2\omega - 1$, with $k_{n-2} \in B_1(2\omega) \cap B_1(\omega)$ or $B_1(2\omega - 1) \cap B_1(\omega)$ respectively.

If $a_{n-1} = \omega$, subtracting and applying 1/z gives $a_{n-2} = \omega - 2$ or $\omega - 1$ with $k_{n-2} \in B_{1/2}(\frac{\omega-2}{2}) \cap B_1(\omega-2)$ or $(B_{1/2}(\frac{\omega-2}{2}) \cap B_1(\omega-1)) \setminus B_{1/2}(\frac{\omega-1}{2})$ respectively. If $a_{n-2} = \omega - 1$, subtracting a_{n-2} , applying 1/z, and using the three-term forbidden sequences

a_i	a_{i+1}	a_{i+2}
$\omega - 1$	$\omega - 1$	ω
$\omega - 2$	$\omega - 1$	ω

shows that $a_{n-3} = 2\omega - 2$ or $2\omega - 1$ with $k_{n-3} \in B_1(2\omega - 2) \cap B_1(\omega - 1)$ or $B_1(2\omega - 1) \cap B_1(\omega - 1)$ respectively. If $a_{n-2} = \omega - 2$, subtracting a_{n-2} and applying 1/z shows that $a_{n-3} = \omega$ or $\omega + 1$, with $k_{n-3} \in (B_1(\omega) \cap B_{1/2}(\frac{\omega+1}{2})) \setminus B_1(0)$ or $B_1(\omega + 1) \cap B_{1/2}(\frac{\omega+1}{2})$ respectively. If $a_{n-3} = \omega + 1$, we loop back into a symmetric version of a case previously considered (namely $k_{n-4} \in B_{1/2}(\frac{\omega-2}{2}) \setminus B_1(0)$ and $a_{n-4} = \omega - 1$ or $\omega - 2$). If $a_{n-3} = \omega$, subtracting a_{n-3} , applying 1/z, and using the forbidden sequences

	a_{i+1}	a_{i+2}	a_{i+3}	a_{i+4}
$\omega - 1$	ω	$\omega - 2$		
ω	ω	$\omega - 2$	ω	
$\omega + 1$	ω	$\omega - 2$	ω	ω

shows that $a_{n-4} = 2\omega$ or $2\omega - 1$ with $k_{n-4} \in B_1(2\omega) \cap B_1(\omega)$ or $B_1(2\omega - 1) \cap B_1(\omega)$ resepectively.

Continuing, we find k_i for $i \leq n-1$ restricted to the region depicted on the right in Figure 3.7. This region contains no integers, contradicting $k_1 \in \mathcal{O}$.

3.4 An overview of A. L. Schmidt's continued fractions

Here we give a quick summary of Asmus Schmidt's continued fraction algorithm [Sch75a], its ergodic theory [Sch82], and further results of Hitoshi Nakada concerning these [Nak88a], [Nak90], [Nak88b]. Define the following matrices in $PGL(2, \mathbb{Z}[i])$:

$$V_{1} = \begin{pmatrix} 1 & i \\ & \\ 0 & 1 \end{pmatrix}, V_{2} = \begin{pmatrix} 1 & 0 \\ & \\ -i & 1 \end{pmatrix}, V_{3} = \begin{pmatrix} 1 - i & i \\ & \\ -i & 1 + i \end{pmatrix},$$



Figure 3.7: In the proof of Proposition 3.3.1, the assumption $k_n < 1$ with $a_n = \frac{1+\sqrt{-7}}{2}$ (left) or $a_n = \frac{1+\sqrt{-11}}{2}$ (right) leads to restricted potential values for k_i with i < n (green regions).

$$E_{1} = \begin{pmatrix} 1 & 0 \\ 1 - i & i \end{pmatrix}, E_{2} = \begin{pmatrix} 1 & -1 + i \\ 0 & i \end{pmatrix}, E_{3} = \begin{pmatrix} i & 0 \\ 0 & 1 \end{pmatrix}, C = \begin{pmatrix} 1 & -1 + i \\ 1 - i & i \end{pmatrix}.$$

Note that

$$S^{-1}V_i S = V_{i+1}, \ S^{-1}E_i S = V_{i+1}, \ S^{-1}CS = C \text{ (indices modulo 3)}$$

where $S = \begin{pmatrix} 0 & -1 \\ 1 & -1 \end{pmatrix}$ is elliptic of order three, and
 $\tau \circ m \circ \tau = m^{-1}$

for the Möbius transformations m induced by $\{V_i, E_i, C\}$ (here τ is complex conjugation).

In [Sch75a] Schmidt uses infinite words in these letters to represent complex numbers as infinite products $z = \prod_n T_n, T_n \in \{V_i, E_i, C\}$ in two different ways. Let $M_N = \prod_{n=1}^N T_n$. We have regular chains

$$\det M_N = \pm 1 \Rightarrow T_{n+1} \in \{V_i, E_i, C\}, \ \det M_N = \pm i \Rightarrow T_{n+1} \in \{V_i, C\},$$

representing z in the upper half-plane \mathcal{I} (the model circle) and **dually regular chains**

$$\det M_N = \pm i \Rightarrow T_{n+1} \in \{V_i, E_i, C\}, \ \det M_N = \pm 1 \Rightarrow T_{n+1} \in \{V_i, C\},$$

representing $z \in \{0 \le x \le 1, y \ge 0, |z - 1/2| \ge 1/2\} =: \mathcal{I}^*$ (the model triangle). The model circle is a disjoint union of four triangles and three circles, and the model triangle is a disjoint union of three triangles and one circle (see Figure 3.8):

$$\mathcal{I} = \mathcal{V}_1 \cup \mathcal{V}_2 \cup \mathcal{V}_3 \cup \mathcal{E}_1 \cup \mathcal{E}_2 \cup \mathcal{E}_3 \cup \mathcal{C}$$
$$\mathcal{I}^* = \mathcal{V}_1^* \cup \mathcal{V}_2^* \cup \mathcal{V}_3^* \cup \mathcal{C}^*$$

where

$$\mathcal{V}_i = v_i(\mathcal{I}), \ \mathcal{E}_i = e_i(\mathcal{I}^*), \ \mathcal{C} = c(\mathcal{I}^*), \ \mathcal{V}_i^* = v_i(\mathcal{I}^*), \ \mathcal{C}^* = c(\mathcal{I}),$$

(lowercase letters indicating the Möbius transformation associated to the corresponding matrix).



Figure 3.8: Subdivision of the upper half-plane \mathcal{I} (model circle) and ideal triangle \mathcal{I}^* with vertices $\{0, 1, \infty\}$ (model triangle). The continued fraction map T sends the circular regions onto \mathcal{I} and the triangular regions onto \mathcal{I}^* .

By considering $z = \prod_n T_n$ we obtain rational approximations $p_i^{(N)}/q_i^{(N)}$ to z by

$$M_N \begin{pmatrix} 1 & 0 & 1 \\ & & \\ 0 & 1 & 1 \end{pmatrix} = \begin{pmatrix} p_1^{(N)} & p_2^{(N)} & p_3^{(N)} \\ q_1^{(N)} & q_2^{(N)} & q_3^{(N)} \end{pmatrix}$$

(which are the orbits of $\infty, 0, 1$ under the partial products $m_N = t_1 \circ \cdots \circ t_N$).

The shift map T on $X = \mathcal{I} \cup \mathcal{I}^* = \{\text{chains, dual chains}\}$ maps X to itself via Möbius transformations, specifically (mapping $\mathcal{V}_i, \mathcal{C}^*$ onto \mathcal{I} and $\mathcal{V}_i^*, \mathcal{E}_i, \mathcal{C}$ onto \mathcal{I}^*)

$$T(z) = \begin{cases} v_i^{-1}z & z \in \mathcal{V}_i \cup \mathcal{V}_i^* \\ e_i^{-1}z & z \in \mathcal{E}_i \\ c^{-1}z & z \in \mathcal{C} \cup \mathcal{C}^* \end{cases}$$

The shift $T: X \to X$ is shown to be ergodic ([Sch82], Theorem 5.1) with respect to the following probability measure

$$\tilde{f}(z) = \begin{cases} \frac{1}{2\pi^2} (h(z) + h(sz) + h(s^2 z)) & z = x + yi \in \mathcal{I} \\ \frac{1}{2\pi} \frac{1}{y^2} & z = x + yi \in \mathcal{I}^* \end{cases}$$

where

$$h(z) = \frac{1}{xy} - \frac{1}{x^2} \arctan\left(\frac{x}{y}\right).$$

By inducing to $X \setminus \bigcup_i (\mathcal{V}_i \cup \mathcal{V}_i^*)$ Schmidt gives "faster" convergents $\hat{p}_{\alpha}^{(n)}/\hat{q}_{\alpha}^{(n)}$ and a sequence of exponents e_n (1 for E_i , C, and the return time k for V_i^k). He then gives results analogous to

those of simple continued fractions via the pointwise ergodic theorem, including the arithmetic and geometric mean of the exponents which exist for almost every z ([Sch82] theorem 5.3)

$$\lim_{n \to \infty} \left(\prod_{i=1}^{n} e_i \right)^{1/n} = 1.26 \dots, \ \lim_{n \to \infty} \frac{1}{n} \sum_{i=1}^{n} e_i = 1.6667 \dots$$

In [Nak88a], Nakada constructs an invertible extension of T on a space of geodesics in two copies of three-dimensional hyperbolic space. In one copy we take geodesics from $\overline{\mathcal{I}}^*$ to \mathcal{I} and in the other the geodesics from $\overline{\mathcal{I}}$ to \mathcal{I}^* where the overline indicates complex conjugation. The regions are shown in Figure 3.9. The extension acts as Schmidt's T depending on the second coordinate. Nakada doesn't provide a second proof of ergodicity, but quotes Schmidt's result. Also in [Nak88a], results about the the density of Gaussian rationals p/q that appear as convergents and satisfy $|z - p/q| < c/|q|^2$ are obtained. For instance ([Nak88a], theorem 7.3), for almost every $z \in X$ and $0 < c < \frac{1}{1+1\sqrt{2}}$ it holds that

$$\lim_{N \to \infty} \frac{1}{N} \# \{ p/q \in \mathbb{Q}(i) : p/q = p_i^{(n)}/q_i^{(n)}, \ 1 \le n \le N, \ i = 1, 2, 3, \ |z - p/q| < c/|q|^2 \} = \frac{c^2}{\pi}.$$

In [Nak90], Main Theorem, Nakada describes the rate of convergence of Schmidt's convergents. Namely for almost every z

$$\lim_{n \to \infty} \frac{1}{n} \log |q_i^{(n)}| = \frac{E}{\pi}, \ \lim_{n \to \infty} \frac{1}{n} \log \left| z - \frac{p_i^{(n)}}{q_i^{(n)}} \right| = -\frac{2E}{\pi}, \ E = \sum_{k=0}^{\infty} \frac{(-1)^k}{(2k+1)^2}.$$

For reference, the relationship between the Super-Apollonian Möbius generators $\{\mathfrak{s}_i, \mathfrak{s}_i^{\perp}\}$ (see chapter 4 for definitions) and Schmidt's $\{v_i, e_i, c\}$ are

$$\begin{split} & \mathfrak{s}_1 = c^2 \circ \tau, \ \mathfrak{s}_2 = e_1^2 \circ \tau, \ \mathfrak{s}_3 = e_2^2 \circ \tau, \ \mathfrak{s}_4 = e_3^2 \circ \tau, \\ & \mathfrak{s}_1^{\perp} = 1 \circ \tau, \ \mathfrak{s}_2^{\perp} = v_1^2 \circ \tau, \ \mathfrak{s}_3^{\perp} = v_2^2 \circ \tau, \ \mathfrak{s}_4^{\perp} = v_3^2 \circ \tau. \end{split}$$

Schmidt gave similar algorithms over $\mathbb{Q}(\sqrt{-2})$ and $\mathbb{Q}(\sqrt{-11})$ in [Sch11] and [Sch78].



Figure 3.9: Domain of the invertible extension, $\mathcal{I} \cup \overline{\mathcal{I}^*}$ (left) and $\mathcal{I}^* \cup \overline{\mathcal{I}}$ (right).

Chapter 4

Continued fractions for some right-angled hyperbolic Coxeter groups

4.1 Introduction

In this chapter we discuss approximation algorithms in the plane based on circle packings coming from ideal, right-angled, two-colorable polyhedra in \mathcal{H}^3 (i.e. hyperbolic polyhedra all of whose vertices lie on $\partial \mathcal{H}^3$, whose faces are two-colorable with respect to incidence at edges, and whose faces meet only at right angles). After this general construction, we discuss the simpler example of billiards in the ideal hyperbolic triangle (restriction to the real line of the octahedral and cubeoctahedral examples of 4.4 and 4.5) and some of its properties. We then discuss two examples in some detail (octahedral and cubeoctrahedral continued fractions), with applications to Diophantine approximation over $\mathbb{Q}(\sqrt{-1})$ and $\mathbb{Q}(\sqrt{-2})$. These algorithms are variations on those of A. L. Schmidt [Sch75a], [Sch11], although precisely relating them is difficult (e.g. there doesn't seem to be a simple intertwining map). Note that the discussion of "Super-Apollonian" continued fractions has substantial overlap with our [CFHS17].

4.2 A general construction

4.2.1 Right-angled hyperbolic Coxeter groups from two-colorable polyhedra

Let Π be a polyhedron whose faces can be two-colored, i.e. the faces of Π can be partitioned into two sets $S = \{s_i\}$ and $T = \{t_j\}$ such that no two faces in S share an edge and no two faces in T share an edge. We consider right-angled hyperbolic Coxeter groups of the form

$$\Gamma = \langle s_1, \dots, s_m, t_1, \dots, t_n | s_i^2 = t_i^2 = [s_i, t_j] = 1, s_i \sim t_j \rangle_{z_i}$$

where $s_i \sim t_j$ if the faces s_i and t_j share an edge. In other words, we are considering reflections in the sides of an ideal (all vertices at infinity) right-angled hyperbolic polyhedron whose faces can be two-colored. Every combinatorial type of polyhedron whose faces can be two-colored is realizable as an ideal right-angled hyperbolic polyhedron. This is a consequence of the so-called Koebe-Andreev-Thurston theorem, that every connected simple planar graph can be realized as the tangency graph of disks in the plane. See [KN17] for a discussion in this context.

The ideal boundaries of the planes defining the faces of Π consist of oriented circles $Z(s_i)$, $Z(t_j)$ with the property that the interiors of Z(s), $s \in S$ are disjoint, the interiors of Z(t), $t \in T$ are disjoint, together they cover the sphere, and if Z(s) and Z(t) intersect, they do so at right angles.

4.2.2 A pair of dynamical systems

Denote the interiors of Z(s), Z(t), by C(s), C(t). The sets

$$P(t) := \overline{C(t)} \cap \left(S^2 \setminus \bigcup_{s \in S} C(s)\right), \ P(s) = \overline{C(s)} \cap \left(S^2 \setminus \bigcup_{t \in T} C(t)\right)$$

have polygonal boundary with vertices on Z(t) and Z(s) respectively (the overline indicates closure, so as to include the vertices in the closed sets P(s), P(t)). We have two partitions (excepting shared vertices of the polygonal faces P)

$$S^{2} = (\bigcup_{s \in S} C(s)) \cup (\bigcup_{t \in T} P(t)) = (\bigcup_{s \in S} P(s)) \cup (\bigcup_{t \in T} C(t))$$

of the sphere, the first \mathcal{P}_S making the faces in S open disks and the faces in T closed polygons, and the second \mathcal{P}_T making the faces in T open disks and the faces in S closed polygons. We think of \mathcal{P}_S and \mathcal{P}_T "preferring" S and T respectively. See Figures 4.18 and 4.19 for an example.

We will use these partitions to define a pair of dynamical systems, "dual" in the sense that their invertible extensions are inverses to one another. Let $s_i, t_j \in PGL_2(\mathbb{C}) \rtimes \langle \mathfrak{c} \rangle \cong \text{Isom}(H^3)$, where \mathfrak{c} is complex conjugation, be a realization of Γ as a discrete group of hyperbolic isometries in the upper half-space model. Define two dynamical systems on $S^2 = P^1(\mathbb{C}) = \mathbb{C} \cup \{\infty\}$ as follows

$$\phi_S(z) = \begin{cases} s(z) & z \in C(s) \\ t(z) & z \in P(t) \end{cases}, \ \phi_T(w) = \begin{cases} s(w) & w \in P(s) \\ t(w) & w \in C(t) \end{cases}$$

(Note that ϕ_S , ϕ_T are well-defined at the vertices, which are fixed points.)

The invertible extensions Φ_S and Φ_T are defined on a collection of geodesics (pairs of distinct points of $P^1(\mathbb{C})$) as follows. For each $s \in S$ and each $t \in T$, define the dual regions

$$C^*(s) = \left(\cup_{t'}C(t')\right) \cup \left(\cup_{t'}P(t')\right), \ P^*(s) = \left(\cup_{t'}C(t')\right) \cup \left(\cup_{t'}P(t')\right),$$
$$C^*(t) = \left(\cup_{s'}C(s')\right) \cup \left(\cup_{s'}P(s')\right), \ P^*(t) = \left(\cup_{s'}C(s')\right) \cup \left(\cup_{s'}P(s')\right),$$

where the primed index in the definition of $C^*(s)$ is over all $t \in T$ such that the regions C(t'), P(t'), do not intersect C(s) (and similarly for the three other dual regions). The invertible extensions are defined on the pairs (thought of as geodesics going from w in one of the dual regions to z in one of the circular or polygonal regions for Φ_S and vice versa for Φ_T)

$$(w, z) \in \mathcal{G} = (\bigcup_{s \in S} C^*(s) \times C(s)) \cup (\bigcup_{t \in T} P^*(t) \times P(t))$$
$$= (\bigcup_{s \in S} P^*(s) \times P(s)) \cup (\bigcup_{t \in T} C^*(t) \times C(t)).$$

On \mathcal{G} , we have (acting diagonally, extending ϕ_S in the second coordinate and ϕ_T in the first coordinate respectively)

$$\Phi_S(w, z) = (r(w), \phi_S(z)) \text{ where } \phi_S(z) = r(z), \ r \in S \cup T,$$

$$\Phi_T(w, z) = (\phi_T(w), q(z)) \text{ where } \phi_T(w) = q(w), \ q \in S \cup T.$$

The reader can convince themselves that the Φ are bijections on the stated domains (see Figure 4.16 for an example of the domain for the invertible extension in the octahedral case). Moreover, the extensions Φ_S and Φ_T are inverse to one another.

4.2.3 Symbolic dynamics

The maps ϕ and Φ are one- and two-sided subshifts of finite type on the alphabet $S \cup T$. The sequences in question come from the two "obvious" normal forms for elements of Γ . If we consider ϕ_S , then the sequence

$$(\phi_S(z), \phi_S^2(z), \phi_S^3(z), \ldots) = (r_1(z), r_2r_1(z), r_3r_2r_1(z), \ldots), r_n \in S \cup T,$$

has the following properties.

- r_n ≠ r_{n+1}, i.e. no words of the form s² or t² appear (inversions in the boundary circles of C(s) or P(t) ensure you do not repeat a digit).
- If r_n and r_{n+1} commute (i.e. their fixed circles are orthogonal), then r_n ∈ S and r_{n+1} ∈ T.
 Geometrically, this comes from the fact that we are taking C(s) and P(t) in the definition,
 i.e. we prefer S over T in the definition.

Similarly, the map ϕ_T will produce a sequence

$$(\phi_T(w), \phi_T^2(w), \phi_T^3(w), \ldots) = (q_1(w), q_2q_1(w), q_3q_2q_1(w), \ldots), \ q_n \in S \cup T,$$

satisfying the opposite convention for the commutation relations: if q_n and q_{n+1} commute, then $q_n \in T$ and $q_{n+1} \in S$.

Hence, to a point $w \in \mathbb{C}$ or $z \in \mathbb{C}$, we encode the pair (w, z) as

$$(\ldots q_3 q_2 q_1, r_1 r_2 r_3 \ldots),$$

and one sees that Φ_S and Φ_T are the left- and right-shift on the bi-infinite word (concatenating the q- and r-strings), and that this bi-infinite word (read left to right) is in the normal form described above for the r-sequence (i.e. s before t when they commute and no instances of s^2 or t^2).

4.2.4 Invariant measures and ergodicity

The maps Φ are bijections defined piecewise by isometries, and therefore preserve the isometry invariant measure $d\eta = |z - w|^{-4} du dv dx dy$ (where w = u + iv, z = x + iy) on the space of geodesics \mathcal{G} . [This is simple to verify or derive as was done for \mathcal{H}^2 in 3.2.1] This pushforward of this measure onto the first or second coordinate gives explicit invariant measures for the continued fraction maps ϕ . See 4.4.6 and 4.5.1 for examples.

The measure preserving systems described in this chapter are almost certainly ergodic, but I have not been able to prove it. The frame flow and geodesic flow on the finite volume quotient $\Gamma \setminus \mathcal{H}^3$ are ergodic, but the systems described in this section seem to live in some hybrid of cross-sections of flows on the two infinite volume quotients $\langle s \in S \rangle \setminus \mathcal{H}^3$ and $\langle t \in T \rangle \setminus \mathcal{H}^3$. A. L. Schmidt gives a direct argument for ergodicity of his continued fraction map in [Sch82].

4.3 Billiards in the ideal triangle

Consider the group $\Gamma = \langle \mathfrak{a}, \mathfrak{b}, \mathfrak{c} \rangle \subseteq PGL_2(\mathbb{R}) \cong \text{Isom}(\mathcal{H}^2)$, generated by reflections in the walls of the ideal hyperbolic triangle with vertices $\{0, 1, \infty\}$:

$$\mathfrak{a}(x) = -x, \ \mathfrak{b}(x) = \frac{x}{2x-1}, \ \mathfrak{c}(x) = 2-x.$$

We have the short exact sequence and isomorphisms

$$1 \to \Gamma \to PGL_2(\mathbb{Z}) \to PGL_2(\mathbb{Z}/(2)) \to 1,$$
$$PGL_2(\mathbb{Z}) = \Gamma \rtimes S_3, \ \Gamma \cong \mathbb{Z}/(2) * \mathbb{Z}/(2) * \mathbb{Z}/(2).$$

Reduced words in these generators index the triangles in the associated tessellation, and the words of length n partition the line into $3 \cdot 2^{n-1}$ subintervals. The partition by words of length m > nrefines the partition by words of length n. Irrational x are then uniquely coordinatized by infinite words in the alphabet $\{a, b, c\}$. See Figure 4.1.

The expansion of x as an infinite word in $\{\mathfrak{a}, \mathfrak{b}, \mathfrak{c}\}$ is produced by a dynamical system T: $P^1(\mathbb{R}) \to P^1(\mathbb{R})$:

$$T(x) = \begin{cases} \mathfrak{a}(x) = -x & x \in [-\infty, 0] \\ \mathfrak{b}(x) = \frac{x}{2x-1} & x \in [0, 1] \\ \mathfrak{c}(x) = 2 - x & x \in [1, \infty]. \end{cases}$$



Figure 4.1: Partion of the line induced by reduced words of length three in the Coxeter generators.

There are three neutral fixed points, $\frac{0}{1}$, $\frac{1}{1}$, and $\frac{1}{0}$, to which rational points descend in finitely many steps depending on the parity of the numerator and denominator (even/odd, odd/odd, odd/even).



Figure 4.2: Graph of T(x), with fixed points at 0, 1, and ∞ .

If x is irrational and $\mathfrak{m} \in {\mathfrak{a}, \mathfrak{b}, \mathfrak{c}}$ is defined by $T^n x = \mathfrak{m}(T^{n-1}x)$, then we obtain three sequences of rational convergents by following the inverse orbit of ${0, 1, \infty}$,

$$\lim_{n \to \infty} \mathfrak{m}_1 \mathfrak{m}_2 \dots \mathfrak{m}_n x_0 = x, \ x_0 = 0, 1, \infty,$$

i.e. we keep track of the the vertices of the triangles through which the geodesic $\overrightarrow{\infty x}$ passes. This algorithm is a slow, reflective, version of the usual continued fraction algorithm. The convergents can be constructed starting from $(1/1, \pm 1/0, 0/1)$, according as x is positive or negative, and taking mediants, moving through the Farey tree. Applying \mathfrak{a} , \mathfrak{b} , or \mathfrak{c} updates the first, second, or third

position by taking the mediant $\frac{p}{q} \oplus \frac{r}{s} := \frac{p+q}{r+s}$ of the other two entries.

For example, the first 20 convergents of the random number

 $x = 0.4189513796210592\ldots, x = bacabcacbcacacababac\ldots,$

are (mediants in italics, see Figure 4.3):

(1/1, 1/0, 0/1), (1/1, 1/2, 0/1), (1/3, 1/2, 0/1), (1/3, 1/2, 2/5), (3/7, 1/2, 2/5), (3/7, 5/12, 2/5), (3/7, 5/12, 2/5), (3/7, 5/12, 2/5), (3/7, 5/12, 2/5), (3/7, 5/12, 2/5), (3/7, 5/12, 2/5), (3/7, 5/12, 2/5), (13/31, 31/74, 18/43), (13/31, 31/74, 44/105), (75/179, 31/74, 18/43), (13/31, 31/74, 44/105), (75/179, 31/74, 44/105), (75/179, 31/74, 106/253), (137/327, 31/74, 168/401), (199/475, 31/74, 168/401), (139/475, 367/876, 168/401), (535/1277, 367/876, 168/401), (535/1277, 703/1678, 168/401), (871/2079, 703/1678, 168/401), (871/2079, 703/1678, 1574/3757)

 $= (0.4189514\dots, 0.4189511\dots, 0.4189512\dots).$



Figure 4.3: Approximating x = 0.4189513796210592..., $\mathfrak{x} = \mathfrak{bacabcaccacababac}...$

The invertible extension \widetilde{T} of T is defined on the space of geodesics \mathcal{G} that intersect the

fundamental triangle, acting on \overrightarrow{yx} depending on x:

$$\widetilde{T}(y,x) = \begin{cases} (\mathfrak{a}(y),\mathfrak{a}(x)) & x \in [-\infty,0] \\ (\mathfrak{b}(y),\mathfrak{b}(x)) & x \in [0,1] \\ (\mathfrak{c}(y),\mathfrak{c}(x)) & x \in [1,\infty]. \end{cases}$$

 \widetilde{T} associates to the geodesic $\overline{y}\overline{x}$ a bi-infinite word $\mathfrak{y}^{-1}\mathfrak{x}$ in $\{\mathfrak{a}, \mathfrak{b}, \mathfrak{c}\}$, which we will relate to the geodesic flow in $\Gamma \setminus \mathcal{H}^2$.



Figure 4.4: Two iterations of \widetilde{T} , red to green to blue

The measure $d\eta(y, x) = (x - y)^{-2} dx dy$ is isometry-invariant on the space of geodesics in the hyperbolic plane. Since \widetilde{T} is a bijection defined piecewise by isometries, $\eta|_{\mathcal{G}}$ is \widetilde{T} -invariant. Pushing forward to the second coordinate gives an *infinite* T-invariant measure

$$d\mu(x) = \begin{cases} \frac{dx}{-x} & x < 0, \\ \frac{dx}{x(1-x)} & 0 < x < 1, \\ \frac{dx}{x-1} & x > 1, \end{cases}$$

We will see that $(\mathcal{G}, \widetilde{T}, \eta|_{\mathcal{G}})$ is ergodic (hence also the ergodicity of $(P^1(\mathbb{R}), T, \mu)$).

The word $\mathfrak{y}^{-1}\mathfrak{x}$ associated to the geodesic \overrightarrow{yx} records the sequence of collisions with the walls of the triangle in $\Gamma \setminus \mathcal{H}^2$, and \widetilde{T} is the first-return of the geodesic flow (billiards in the triangle) to the cross-section defined by points/directions on the boundary. The return time is integrable with respect to $d\eta(y, x)$. For instance, a geodesic $(y, x) \in [-\infty, 0] \times [1, \infty]$, has return time

$$r(y,x) = \frac{1}{2} \log \left(\frac{x(1-y)}{y(1-x)} \right).$$

and the integral is

$$\frac{1}{2} \int_{-\infty}^{0} \int_{1}^{\infty} \log\left(\frac{x(1-y)}{y(1-x)}\right) \frac{dxdy}{(y-x)^2} = \frac{\pi^2}{6}.$$

It is well-known (since E. Hopf [Hop71]) that the geodesic flow on finite volume hyperbolic surfaces is ergodic, from which it follows that \tilde{T} and T are ergodic.

4.4 Octahedral ("Super-Apollonian") continued fractions over $\mathbb{Q}(\sqrt{-1})$

[For the convenience of the author, we follow the notation of [CFHS17] which contains more on our "Super-Apollonian" continued fractions and its relationship with Apollonian circle packings and the Super-Apollonian group of [GLM⁺05], [GLM⁺06a], [GLM⁺06b].] In this section we describe the right-angled continued fractions generated by reflections in the sides of the ideal regular rightangled octahedron with vertices

$$0, 1, \infty, i, 1+i, \frac{1}{1-i}.$$

This reflection group Γ is generated by

$$\mathfrak{s}_{1} = \frac{(1+2i)\bar{z}-2}{2\bar{z}-1+2i}, \ \mathfrak{s}_{2} = \frac{\bar{z}}{2\bar{z}-1}, \ \mathfrak{s}_{3} = -\bar{z}+2, \ \mathfrak{s}_{4} = -\bar{z},$$
$$\mathfrak{s}_{1}^{\perp} = \bar{z}, \ \mathfrak{s}_{2}^{\perp} = \bar{z}+2i, \ \mathfrak{s}_{3}^{\perp} = \frac{\bar{z}}{-2i\bar{z}+1}, \ \mathfrak{s}_{4}^{\perp} = \frac{(1-2i)\bar{z}+2i}{-2i\bar{z}+1+2i}$$

and fits into the short exact sequence

$$1 \to \Gamma \to PGL_2(\mathbb{Z}[i]) \rtimes \langle \bar{z} \rangle \to PGL_2(\mathbb{Z}[i]/(2)) \to 1,$$

with

$$[PGL_2(\mathbb{Z}[i]) \rtimes \langle \bar{z} \rangle : \Gamma] = 48, \ PGL_2(\mathbb{Z}[i]) \rtimes \langle \bar{z} \rangle = \Gamma \rtimes \text{Bin. Oct.}$$

4.4.1 Digression on the Apollonian super-packing and the Super-Apollonian group

While this subsection isn't necessary for understanding the rest of the chapter, it gives convenient language to discuss the continued fraction algorithms we present in the sequel since the partitions of the approximation algorithm come from circle packings.



Figure 4.5: Partion of the plane induced by words of length 1, 2, 3, and 4 in the Coxeter generators (associated to the map T_B), or the orbit of the action of words from \mathcal{A}^S of length 0, 1, 2, and 3 on the base quadruple R_B .



Figure 4.6: Circles in $[0,1] \times [0,1]$ in the orbit of words of length ≤ 5 from \mathcal{A}^S acting on the base quadruple R_B .

A *Descartes quadruple* is a collection of four mutually tangent circles in the plane (ordered and oriented so that the interiors do not overlap), and its dual quadruple consists of the four mutually tangent circles passing orthogonally through its points of tangency. The curvatures (oriented inverse radii) of the circles satisfy

$$2(c_1^2 + c_2^2 + c_3^2 + c_4^2) = (c_1 + c_2 + c_3 + c_4)^2,$$

i.e. they are zeros of the quadratic form

We can move between "adjacent" quadruples with four *swaps* (fix three circles and replace the fourth by its inversion in the disjoint dual circle) and four *inversions* (fix one circle and replace the other three with their inversions in the fixed circle). See Figure 4.7 for an example of these operations.



Figure 4.7: A Descartes quadruple and its dual, the four "swaps" S_i , and the four "inversions" S_i^{\perp} .

The super-Apollonian group $\mathcal{A}^S \subseteq O_D(\mathbb{Z}) \subseteq GL_4(\mathbb{Z})$ is the group generated by the swaps

and inversions $(S_i^\perp:=S_i^t)$

$$S_{1} = \begin{pmatrix} -1 & 2 & 2 & 2 \\ 0 & 1 & 0 & 0 \\ 0 & 0 & 1 & 0 \\ 0 & 0 & 0 & 1 \end{pmatrix}, S_{2} = \begin{pmatrix} 1 & 0 & 0 & 0 \\ 2 & -1 & 2 & 2 \\ 0 & 0 & 1 & 0 \\ 0 & 0 & 0 & 1 \end{pmatrix}, S_{3} = \begin{pmatrix} 1 & 0 & 0 & 0 \\ 0 & 1 & 0 & 0 \\ 2 & 2 & -1 & 2 \\ 0 & 0 & 0 & 1 \end{pmatrix}, S_{4} = \begin{pmatrix} 1 & 0 & 0 & 0 \\ 0 & 1 & 0 & 0 \\ 0 & 0 & 1 & 0 \\ 2 & 2 & 2 & -1 \end{pmatrix}.$$

 \mathcal{A}^{S} is a **right-angled** hyperbolic Coxeter group with presentation

$$\langle S_i, S_i^{\perp}, 1 \le i \le 4 : S_i^2 = (S_i^{\perp})^2 = [S_i, S_j^{\perp}] = 1, i \ne j \rangle.$$

The Super-Apollonian group acts on Descartes quadruples where the four circles are coordinatized by indefinite binary Hermitian forms of determinant -1 (called augmented curvature center or ACC coordinates in [GLM⁺06a]). To the circle

$$0 = A|z|^2 + B\overline{z} + \overline{B}z + C, \ A, C \in \mathbb{R}, \ B \in \mathbb{C}, \ AC - |B|^2 = -1$$

we associate the quadruple of real numbers

$$(C, A, B_1, B_2),$$

the co-curvature, curvature, x-coordinate of the curvature-center, and y-coordinate of the curvaturecenter respectively. Four circles (given by the columns of the 4×4 matrix R) form a Descartes quadruple if and only if they satisfy

$$R^{t}DR = \begin{pmatrix} 0 & -4 & 0 & 0 \\ -4 & 0 & 0 & 0 \\ 0 & 0 & 2 & 0 \\ 0 & 0 & 0 & 2 \end{pmatrix}.$$

The Super-Apollonian group acts by left-multiplication on such R as swaps and inversions. We choose two convenient "base quadruples," shown in Figure 4.8, for later use:

$$R_B = \begin{pmatrix} 0 & 0 & 0 & -1 \\ 2 & 0 & 0 & 1 \\ 0 & 2 & 0 & 1 \\ 2 & 2 & 2 & 1 \end{pmatrix}, R_A = R_B^{\perp} = \begin{pmatrix} 2 & 2 & 1 & 2 \\ 0 & 2 & 1 & 0 \\ 2 & 0 & 1 & 0 \\ 0 & 0 & -1 & 0 \end{pmatrix}$$



Figure 4.8: The dual base quadruples R_A (red) and R_B (blue).

4.4.2 A pair of dynamical systems

The pair of dynamical systems on $P^1(\mathbb{C})$ associated to the two-colorable octahedron with faces defined by $S := \{\mathfrak{s}_i\}$ and $S^{\perp} := \{\mathfrak{s}_i^{\perp}\}$ are as follows. Let B_i , B'_i , be the open circular and closed triangular regions of the plane coming from S^{\perp} (A_i and A'_i defined similarly, see Figure 4.9), and define

$$T_B(z) = \begin{cases} \mathfrak{s}_i z \quad z \in B'_i, \\ \mathfrak{s}_i^{\perp} z \quad z \in B_i, \end{cases}, \quad T_A(w) = \begin{cases} \mathfrak{s}_i w \quad w \in A_i, \\ \mathfrak{s}_i^{\perp} w \quad w \in A'_i. \end{cases}$$

Each of T_A and T_B has six fixed points, the points of tangency $\{0, 1, \infty, i, i + 1, 1/(1-i)\}$.

Iteration of the map T_B with input z produces a word $\mathfrak{z} = \mathfrak{m}_1 \cdots \mathfrak{m}_n \cdots$ in the Möbius generators $S \cup S^{\perp}$, where \mathfrak{m}_n is defined by $T_B^n z = \mathfrak{m}_n(T_B^{n-1}z)$. (Similarly for T_A .) We take the word to be finite for $z \in \mathbb{Q}(i)$, ending when a fixed point is reached (cf. 4.4.3). As discussed in 4.2.3, the words produced will be in "normal form," i.e.

- the words $(\mathfrak{s}_i^{\perp})^2$, \mathfrak{s}_i^2 do not appear, and
- if \mathfrak{m}_n and \mathfrak{m}_{n+1} commute, then $\mathfrak{m}_n \in S^{\perp}$ and $\mathfrak{m}_{n+1} \in S$ (for T_B) or $\mathfrak{m}_n \in S$ and $\mathfrak{m}_{n+1} \in S^{\perp}$ (for T_A).

The collection of length n words in normal form partitions the plane into a collection of $9 \cdot 5^{n-1} - 1$ triangles and circles, which we call Farey circles and triangles following Schmidt [Sch75a].



Figure 4.9: The regions A_i , A'_i , B_i , B'_i .

Specifically, we associate to each word an open circular or closed triangular region, the notation being

$$F_B(\mathfrak{m}) = \mathfrak{m}_1 \cdots \mathfrak{m}_{n-1} B_i$$
 (circular), $\mathfrak{m}_1 \cdots \mathfrak{m}_{n-1} B'_i$ (triangular),

for $\mathfrak{m} = \mathfrak{m}_1 \cdots \mathfrak{m}_n$ with $\mathfrak{m}_n = \mathfrak{s}_i^{\perp}$ (circular) or \mathfrak{s}_i (triangular). This definition is set up so that the words of length one correspond to the eight regions of the base quadruple. See Figure 4.10 for the partition and labels on words of length two and Figure 4.5 for the first four partitions.



 $\mathfrak{s}_2^\perp \mathfrak{s}_1^\perp$

Figure 4.10: Partition of the complex plane induced by words of length two in swap normal form in the Möbius generators.

If z is rational, then $z = \mathfrak{z}v$ for a unique $v \in \{0, 1, \infty, i, 1 + i, \frac{1}{1-i}\}$ matching z in parity as described in Theorem 4.4.1. If z is not rational, then \mathfrak{z} is an infinite word with the property that

$$z = \bigcap_{n \ge 1} F(\mathfrak{m}_1 \cdots \mathfrak{m}_n).$$

This gives a bijection $z \leftrightarrow \mathfrak{z}$, under which T_B is the left shift operator, $T_B(\mathfrak{m}_1\mathfrak{m}_2\cdots) = \mathfrak{m}_2\mathfrak{m}_3\cdots$.



Figure 4.11: Approximating the irrational point 0.3828008104...+i0.2638108161... using T_B gives the normal form word $\mathfrak{s}_3^{\perp}\mathfrak{s}_2\mathfrak{s}_2^{\perp}\mathfrak{s}_3^{\perp}\mathfrak{s}_1\mathfrak{s}_1^{\perp}\mathfrak{s}_4\mathfrak{s}_2\mathfrak{s}_4\mathfrak{s}_1\mathfrak{s}_1^{\perp}\mathfrak{s}_3\mathfrak{s}_2\mathfrak{s}_3\mathfrak{s}_3^{\perp}\mathfrak{s}_4\mathfrak{s}_4^{\perp}\mathfrak{s}_2\mathfrak{s}_1\mathfrak{s}_2\cdots$.

Before discussing the invertible extension(s) and invariant measure(s), we consider some arithmetic properties in the next few subsections.

4.4.3 A Euclidean algorithm

When applied to rational points p/q, iterating T_A and T_B performs a "slow, reflective" Euclidean algorithm, reducing the height of p/q until one of the vertices of the fundamental octahedron is reached.

Theorem 4.4.1. Under the dynamical systems T_A and T_B , every Gaussian rational $z \in \mathbb{Q}(i)$ reaches one of the fixed points in finite time. The fixed point reached is determined by the "parity" of the numerator and denominator of z = p/q, i.e. one of the six equivalence classes under the equivalence relation

$$\frac{p}{q} \sim \frac{r}{s} \iff ps \equiv qr \pmod{2}.$$

Proof. The extended Bianchi¹ group $B[-1] = PGL_2(\mathbb{Z}[i]) \rtimes \langle \mathfrak{c} \rangle$ is a maximal discrete subgroup of $\operatorname{Isom}(\mathcal{H}^3) \cong PSL_2(\mathbb{C}) \rtimes \langle \mathfrak{c} \rangle$, where \mathfrak{c} is complex conjugation. The group Γ is the kernel of the

¹ after Luigi Bianchi, [Bia92]

surjective map

$$B[-1] = PGL_2(\mathbb{Z}[i]) \rtimes \langle \mathfrak{c} \rangle \to PGL_2(\mathbb{Z}[i]/(2))$$

since Γ is in the kernel and both are of index 48 in B[-1] (we have $[B[-1] : \Gamma] = 48$ comparing fundamental domains and $|PGL_2(\mathbb{Z}[i]/(2))| = 48$ by direct computation). Hence Γ preserves parity, e.g.

$$\mathfrak{s}_1\left(\frac{p}{q}\right) = \frac{(1+2i)\bar{p} - 2\bar{q}}{2\bar{p} + \bar{q}(-1+2i)} \equiv \frac{p}{q} \mod 2.$$

Termination in finite time follows from the following version(s) of the Euclidean algorithm in $\mathbb{Z}[i]$. "Homogenizing" T_A and T_B gives dynamical systems on pairs $(0,0) \neq (p,q) \in \mathbb{Z}[i] \times \mathbb{Z}[i]$ that terminate when $p/q \in \{0, 1, \infty, i, 1 + i, \frac{1}{1-i}\}$, i.e. act on pairs via complex conjugation and the matrices implied in the definitions of the $\mathfrak{s}_i, \mathfrak{s}_i^{\perp}$. For instance,

$$T_B(p,q) := (\bar{p}, 2\bar{p} - \bar{q})$$
 for $\frac{p}{q} \in B'_2$, where $T_B(p/q) = \mathfrak{s}_2(p/q) = \frac{p/q}{2(\overline{p/q}) - 1}$.

We'll consider the case of T_B , noting that the proof for T_A is nearly identical:

$$(p,q) \mapsto \begin{cases} s_1(p,q) = ((1+2i)\bar{p} - 2\bar{q}, 2\bar{p} + (2i-1)\bar{q}) & p/q \in B'_1 \setminus \{i, 1+i, \frac{1}{1-i}\} \\ s_2(p,q) = (\bar{p}, 2\bar{p} - \bar{q}) & p/q \in B'_2 \setminus \{0, 1, \frac{1}{1-i}\} \\ s_3(p,q) = (2\bar{q} - \bar{p}, \bar{q}) & p/q \in B'_3 \setminus \{1, 1+i, \infty\} \\ s_4(p,q) = (-\bar{p}, \bar{q}) & p/q \in B'_4 \setminus \{0, i, \infty\} \\ s_4(p,q) = (\bar{p}, \bar{q}) & p/q \in B_1 \setminus \{0, 1, \infty\} \\ s_2^{\perp}(p,q) = (\bar{p} + 2i\bar{q}, \bar{q}) & p/q \in B_2 \setminus \{i, i+1, \infty\} \\ s_3^{\perp}(p,q) = (\bar{p}, \bar{q} - 2i\bar{p}) & p/q \in B_3 \setminus \{0, i, \frac{1}{1-i}\} \\ s_4^{\perp}(p,q) = ((1-2i)\bar{p} + 2i\bar{q}, (2i+1)\bar{q} - 2i\bar{p}) & p/q \in B_4 \setminus \{1, 1+i, \frac{1}{1-i}\}. \end{cases}$$

The inequalities defining the regions B_i , B'_i show that |q| is decreased whenever s_1 , s_2 , s_3^{\perp} , or s_4^{\perp} is applied. For example, when applying s_1 , the fact that p/q is in the triangle $B'_1 \setminus \{i, 1+i, \frac{1}{1-i}\}$ (or the circle A_1 which the inequalities define) shows that |q| is decreased when applying s_1 , as follows:

$$\frac{p}{q} \in B_1' \setminus \left\{ i, 1+i, \frac{1}{1-i} \right\} \Longrightarrow \left| \frac{p}{q} - \frac{1+2i}{2} \right|^2 < \frac{1}{4} \quad \Longrightarrow \quad |2\bar{p} + (1-2i)\bar{q}| < |q|.$$

Similarly, application of s_3 and s_2^{\perp} both decrease |p|. Applying s_4 maps B'_4 onto $B'_1 \cup B'_2 \cup B_4 \cup B'_3$ (from which one of |p|, |q| will be decreased as just discussed). Finally, s_1^{\perp} maps B_1 onto one of the other seven regions. Hence the algorithm terminates.

4.4.4 Geometry of the first approximation constant for $\mathbb{Q}(i)$

The purpose of this section and the next is to relate the Super-Apollonian continued fraction algorithm to classical statements of Diophantine approximation. In this section, we give the first value of the Lagrange spectrum for complex approximation by Gaussian rationals. With this as a point for comparison, in the second section we describe the quality of the approximations obtained by the algorithm.

The "good" rational approximations to an irrational $z \in \mathbb{C}$

$$|z - p/q| \le C/|q|^2$$

are determined by the collection of horoballs

$$B_C(p/q) = \{(z,t) \in H^3 : |z - p/q|^2 + (t - C/|q|^2)^2 \le C^2/|q|^4\},\$$
$$B_C(\infty) = \{(z,t) \in H^3 : t \ge 1/2C\},\$$

through which the geodesic \overrightarrow{wz} passes (or through which any geodesic \overrightarrow{wz} eventually passes). Over $\mathbb{Q}(i)$, the smallest value of C with the property that every irrational z has infinitely many rational approximations satisfying the above inequality was determined by Ford in [For25]. Here we give a short proof of this fact using the geometry of the ideal octahedron.

Proposition 4.4.2. Every $z \in \mathbb{C} \setminus \mathbb{Q}(i)$ has infinitely many rational approximations $p/q \in \mathbb{Q}(i)$ such that

$$\left|z - \frac{p}{q}\right| \le \frac{C}{|q|^2}, \quad C = \frac{1}{\sqrt{3}} \simeq 0.577350269\dots$$

The constant $1/\sqrt{3}$ is the smallest possible, as witnessed by $z = \frac{1+\sqrt{-3}}{2}$.

Proof. The value of C for which the horoballs with parameter C based at the ideal vertices $\{0, 1, \infty, i, 1+i, 1/(1-i)\}$ cover the boundary of the fundamental octahedron is easily found to be
$1/\sqrt{3}$ (one need only cover the face with vertices $\{0, 1, \infty\}$, see Figure 4.12). Hence as we follow the geodesic $\overrightarrow{\infty z}$ through the tesselation by octahedra, at least one of the six vertices of each octahedron satisfies the above inequality with $C = 1/\sqrt{3}$, one or two as it enters and one or two as it exits (some of which may coincide). This gives the smallest value of C for which the inequality above has infinitely many solutions for all irrational z, noting the the geodesic from $e^{-\pi i/3}$ to $e^{\pi i/3}$ passes orthogonally through the "centers" of the opposite faces of the octahedra through which it passes.

The sequence of octahedra we consider in our continued fraction algorithm are not necessarily along the geodesic path, but we do capture all rationals with $|z - p/q| < C/|q|^2$ with $C = 1/(1 + 1/\sqrt{2})$ as detailed in the next section.



Figure 4.12: Horoball covering of the ideal triangular face with vertices $\{0, 1, \infty\}$ with coordinates (z, t).

4.4.5 Quality of rational approximation

To any complex number we associate six sequences of Gaussian rational approximations by following the inverse orbit of the six vertices of the fundamental octahedron (six points of tangency of our base quadruples R_A , R_B). Namely, if $z = \prod_{i=1}^{\infty} \mathfrak{z}_i = \prod_{i=1}^{\infty} \mathfrak{w}_i = w$ in the two codings, then the convergents $p_{n,\alpha}^A/q_{n,\alpha}^A$, $p_{n,\alpha}^B/q_{n,\alpha}^B$ are given by

$$\frac{p_{n,\alpha}^A}{q_{n,\alpha}^A} = \left(\prod_{i=1}^n \mathfrak{w}_i\right)(\alpha), \ \frac{p_{n,\alpha}^B}{q_{n,\alpha}^B} = \left(\prod_{i=1}^n \mathfrak{z}_i\right)(\alpha), \ \alpha \in \{0, 1, \infty, i, i+1, 1/(1-i)\},$$

with the property that

$$\lim_{n \to \infty} \frac{p_{n,\alpha}^A}{q_{n,\alpha}^A} = w, \ \lim_{n \to \infty} \frac{p_{n,\alpha}^B}{q_{n,\alpha}^B} = z$$

for all α and $w \in \mathbb{C} \setminus \mathbb{Q}(i), z \in \mathbb{C} \setminus \mathbb{Q}(i)$.

The following theorem is more or less equivalent to a statement about approximation by Schmidt's continued fractions ([Sch75a], Theorem 2.5), where it is stated without proof.

Theorem 4.4.3. If p/q is such that

$$|z_0 - p/q| < \frac{C}{|q|^2}, \quad C = \frac{\sqrt{2}}{1 + \sqrt{2}} = 0.585786437...,$$

then p/q is a convergent to z_0 (with respect to both T_A and T_B). Moreover, the constant C is the largest possible.

Proof. Note that the Apollonian super-packings associated to the root quadruples R_A , R_B , are invariant under the action of $PSL_2(\mathbb{Z}[i]) \rtimes \langle \mathfrak{c} \rangle$. Consider the quadruple where p/q first appears as a convergent to z_0 , and let $\gamma(z) = \frac{-Qz+P}{-qz+p} \in PSL_2(\mathbb{Z}[i])$ take this quadruple to the base quadruple (say R_B) with p/q mapping to infinity and infinity mapping to Q/q. For any value of C, the disk of radius $C/|q|^2$ centered at p/q gets mapped by γ to the exterior of the disk of radius 1/C centered at Q/q:

$$w = \frac{-Qz + P}{-qz + p} \Rightarrow |z - p/q| = \frac{1}{|w - Q/q||q|^2},$$
$$\frac{C}{|q|^2} \ge |z - p/q| = \frac{1}{|w - Q/q||q|^2} \Rightarrow |w - Q/q| \ge 1/C$$

Consider the ways in which p/q can first appear as a convergent to z_0 in the sequence of partitions of the plane.

We might invert into a circle containing z₀. In particular, then, p/q is in the interior of the circle of inversion (since it is its first appearance as a convergent). In this case, by the discussion in 4.4.2, all z inside the circle also include this inversion in their expansion. Therefore, all z inside the circle will have p/q as a convergent. Our goal is to show that the circle of radius 1/|q|² around p/q is contained in the circle of inversion.

Under γ above (perhaps after applying some binary tetrahedral symmetry of the base quadruple), the circles A, B, get mapped to A', B' as in the figures below, with $Q/q = \gamma(\infty)$ lying in the triangle inside B' as shown. The exterior of a disk of radius one centered at Q/q does not meet the interior of B', hence, applying γ^{-1} , the disk of radius $1/|q|^2$ centered at p/q does not meet B. Therefore p/q is a convergent to any z with $|z - p/q| < 1/|q|^2$. See Figure 4.13.



Figure 4.13: Left: The Descartes quadruples (octahedon) when p/q first appears as a convergent when inverting. Right: The result after mapping back to the fundamental octahedron and sending $p/q \rightarrow \infty$ and ∞ to Q/q.

We might swap into a triangle containing z₀ producing p/q as a point of tangency on an edge of this triangle. In the image, z₀ is in the quadrangle with sides formed by A, B, C, D; this is the union of two triangles. The dotted circles in the figures indicate the two possible swaps associated to the initial creation of p/q as a convergent. In this case, any z in the indicated quadrangle will have p/q as a convergent. We aim to show that a circle of radius C/|q|² around p/q is contained in this region.

Under γ above (perhaps after applying some binary tetrahedral symmetry of the base quadruple), the circles A, B, C, D, and E are mapped to A', B' C', D' and E', the three



Figure 4.14: Left: The Descartes quadruples (octahedra) when p/q first appears as a convergent when swapping. Right: The result after mapping back to the fundamental octahedron and sending $p/q \to \infty$ and ∞ to Q/q.

circles tangent to p/q are mapped to the lines in the second picture, with $Q/q = \gamma(\infty)$ lying in the intersection of the disks defined by B' and E'. The exterior of any circle of radius $1/C = 1 + 1/\sqrt{2}$ centered inside E' avoids the interiors of A', B', C', and D'. Applying γ^{-1} shows that the disk of radius $C/|q|^2$ around p/q does not meet A, B, C, or D, so that p/q is a convergent to any z with $|z - p/q| < C/|q|^2$. See Figure 4.22.

4.4.6 Invertible extension and invariant measures

In this section, we discuss the invertible extension T of T_B , with the property that T^{-1} extends T_A , along with an invariant measure for T. In what follows we adhere to the convention of using w, \mathfrak{w} as coordinates for T_A and z, \mathfrak{z} as coordinates for T_B since we will be using both throughout.

The space of oriented geodesics in \mathcal{H}^3 , identified with pairs in $P^1(\mathbb{C}) \times P^1(\mathbb{C}) \setminus diag$. carries an isometry invariant measure

$$|z - w|^{-4} du \, dv \, dx \, dy, \ w = u + iv, \ z = x + iy.$$

We restrict this measure to geodesics in the set

$$\mathcal{G} = \left(\bigcup_{i} A_{i} \times B_{i}\right) \bigcup \left(\bigcup_{i \neq j} A_{i}' \times B_{j}\right) \bigcup \left(\bigcup_{i \neq j} A_{i} \times B_{j}'\right) \bigcup \left(\bigcup_{i,j} A_{i}' \times B_{j}'\right),$$

consisting of geodesics between disjoint A and B regions of the base quadruples (see Figure 4.9). In what follows we use A coordinates for w in the first coordinate and B coordinates for z in the second coordinate. Define $T : \mathcal{G} \to \mathcal{G}$ by

$$T(w,z) = \begin{cases} (\mathfrak{s}_i w, \mathfrak{s}_i z) & z \in B_i, \ \mathfrak{z} = \mathfrak{s}_i \dots \\ (\mathfrak{s}_i^{\perp} w, \mathfrak{s}_i^{\perp} z) & z \in B'_i, \ \mathfrak{z} = \mathfrak{s}_i^{\perp} \dots \end{cases}$$

where \mathfrak{z} is the Möbius transformation corresponding to z as described in 4.4.2. Equivalently,

$$T^{-1}(w,z) = \begin{cases} (\mathfrak{s}_i w, \mathfrak{s}_i z) & w \in A_i, \ \mathfrak{w} = \mathfrak{s}_i \dots \\ (\mathfrak{s}_i^{\perp} w, \mathfrak{s}_i^{\perp} z) & w \in A'_i, \ \mathfrak{w} = \mathfrak{s}_i^{\perp} \dots \end{cases}$$



Figure 4.15: At left, eight iterations of the map T, labelled in sequence (with + and - indicating the orientation). At right, 100 iterations of T, with rainbow coloration indicating time. In both pictures, the geodesic planes defining the octahedron are shown.

In other words, T is applying T_B diagonally depending on the second coordinate and T^{-1} is applying T_A diagonally depending on the first coordinate. In terms of the shifts on pairs $(\mathfrak{w},\mathfrak{z}) = (\prod_i \mathfrak{w}_i, \prod_i \mathfrak{z}_i)$ corresponding to (w, z), we have

$$T(w,z) = (\mathfrak{z}_1\mathfrak{w},\prod_{i=2}^{\infty}\mathfrak{z}_i) = (\mathfrak{z}_1w,T_B(z)), \ T^{-1}(w,z) = (\prod_{i=2}^{\infty}\mathfrak{w}_i,\mathfrak{w}_1\mathfrak{z}) = (T_A(w),\mathfrak{w}_1z)$$

See Figure 4.15 for a visualization of the invertible extension.

Define the following regions (see Figure 4.16):

$$\mathcal{A}_{i} = B_{i} \cup (\cup_{j \neq i} B_{j}')$$
$$\mathcal{A}_{i}' = (\cup_{j} B_{j}') \cup (\cup_{j \neq i} B_{j})$$
$$\mathcal{B}_{i} = A_{i} \cup (\cup_{j \neq i} A_{j}')$$
$$\mathcal{B}_{i}' = (\cup_{i} A_{i}') \cup (\cup_{j \neq i} A_{j})$$

Here \mathcal{A}_i consists of the *B* regions not intersecting A_i , and so on.

Theorem 4.4.4. The function $T : \mathcal{G} \to \mathcal{G}$ is a measure-preserving bijection. Consequently, the push-forward of this measure onto the first or second coordinate gives invariant measures μ_A and

 μ_B for T_A and T_B . Specifically, we have

$$d\mu_A(w) = f_A(w) \, du \, dv = \begin{cases} f_{A_i}(u, v) \, du \, dv = du \, dv \int_{\mathcal{A}_i} |z - w|^{-4} \, dx \, dy, & w \in A_i \\ f_{A'_i}(u, v) \, du \, dv = du \, dv \int_{\mathcal{A}'_i} |z - w|^{-4} \, dx \, dy, & w \in A'_i \end{cases}$$

and

$$d\mu_B(z) = f_B(z) \, dx \, dy = \begin{cases} f_{B_i}(x, y) \, dx \, dy = dx \, dy \int_{\mathcal{B}_i} |z - w|^{-4} \, du \, dv, & z \in B_i \\ f_{B'_i}(x, y) \, dx \, dy = dx \, dy \int_{\mathcal{B}'_i} |z - w|^{-4} \, du \, dv, & z \in B'_i \end{cases}$$

See Figure 4.17 for a graph of the invariant density f_B .



Figure 4.16: Top, left to right: regions $\mathcal{B}'_i \times \mathcal{B}'_i$, i = 1, 2, 3, 4. Bottom, left to right: regions $\mathcal{B}_i \times \mathcal{B}_i$, i = 1, 2, 3, 4. Shown red×blue with subdivisions.

Proof. Note that $\mathcal{G} = \bigcup_i (A_i \times \mathcal{A}_i \cup A'_i \times \mathcal{A}'_i) = \bigcup_i (\mathcal{B}_i \times B_i \cup \mathcal{B}'_i \times B'_i)$, which can be seen by in Figure 4.16. Since

$$T(\mathcal{B}_i \times B_i) = A'_i \times \mathcal{A}'_i,$$
$$T(\mathcal{B}'_i \times B'_i) = A_i \times \mathcal{A}_i,$$

we immediately get that T is a bijection. As T is a bijection defined piecewise by isometries, it preserves the measure described above.

Computing the relevant integrals in Theorem 4.4.4 gives $\pi/4$ times hyperbolic area on the triangular regions B'_i :

$$f_B(x,y) = \begin{cases} \frac{\pi}{4(\frac{1}{4}-d^2)^2} & z \in B'_1, \ d^2 = \left(x-\frac{1}{2}\right)^2 + (y-1)^2 \\ \frac{\pi}{4(\frac{1}{4}-d^2)^2} & z \in B'_2, \ d^2 = \left(x-\frac{1}{2}\right)^2 + y^2 \\ \frac{\pi}{4(1-x)^2} & z \in B'_3 \\ \frac{\pi}{4x^2} & z \in B'_4 \end{cases}$$

and on the circular regions B_i we have

$$f_B(x,y) = \begin{cases} H(x,y) & z \in B_1 \\ H(x,1-y) & z \in B_2 \\ G(x,y) & z \in B_3 \\ G(1-x,y) & z \in B_4 \end{cases},$$

where

$$\begin{split} H(x,y) &= h(x,y) + h(1-x,y) + h(x^2 - x + y^2, y), \\ G(x,y) &= h(x,y^2 - y + x^2) + h(x^2 - x + y^2, y^2 - y + x^2) + h(x^2 - x + (1-y)^2, y^2 - y + x^2), \\ h(x,y) &= \frac{\arctan(x/y)}{4x^2} - \frac{1}{4xy}. \end{split}$$

Furthermore, we have the relationship

$$f_A(w) = f_B(\rho w) = f_B(\mathfrak{d} w)$$

where ρ is rotation by $\pi/2$ around 1/(1-i) and \mathfrak{d} is the isometry switching opposite faces of the octahedron

$$\mathfrak{d} = rac{ar{z}-1+i}{(1-i)ar{z}+i} = \mathfrak{d}^{-1}, \ \mathfrak{s}_i^\perp = \mathfrak{d}\mathfrak{s}_i\mathfrak{d},$$

(the Möbius equivalent of the "duality" operator from [GLM⁺06a]). The measures μ_A , μ_B have S_3 symmetry on each of the A_i , A'_i , B_i , B'_i . For instance the isometries permuting $\{0, 1, \infty\}$ on A'_1 , B_1 , preserve the measure (generators shown for a transposition and three-cycle)

$$(0,1) \sim -\bar{z} + 1, \ (0,1,\infty) \sim \frac{-1}{z-1}.$$

,

The measures also have S_4 symmetry on \mathbb{C} , permuting the pairs $\{A_i, A'_i\}, \{B_i, B'_i\}$ (transpositions (i, i + 1) shown)

$$(1,2) \sim \bar{z} + i, \ (2,3) \sim \frac{1}{\bar{z}}, \ (3,4) \sim -\bar{z} + 1.$$

The total measure assigned to each of the A_i , A'_i , B_i , B'_i , is $\pi^2/4$, so to normalize μ_A , μ_B , we divide by $2\pi^2$. All of the above should be compared with Nakada's extension of Schmidt's system, described in chapter 3.4.



Figure 4.17: Graph of $f_B(z)$, the invariant density for T_B .

The following lemmas compare the measures of some Farey circles and triangles via the involutions \perp and $^{-1}$, simplifying computations. For use in the proofs, we note the equality of

regions

$$\mathfrak{s}_{i}^{\perp} \mathcal{B}_{i} = A_{i}' = \mathfrak{d} B_{i}',$$

$$\mathfrak{s}_{i} \mathcal{B}_{i}' = A_{i} = \mathfrak{d} B_{i},$$

$$\mathfrak{s}_{i} \mathcal{A}_{i} = B_{i}' = \mathfrak{d} A_{i}',$$

$$\mathfrak{s}_{i}^{\perp} \mathcal{A}_{i}' = B_{i} = \mathfrak{d} A_{i}.$$
(4.1)

Lemma 4.4.5. For \mathfrak{m} , \mathfrak{n} , in normal form (associated to T_B and T_A respectively), there are equalities of measure

$$\mu_B(F_B(\mathfrak{m})) = \mu_B(F_B(\mathfrak{m}^{\perp})), \quad \mu_A(F_A(\mathfrak{n})) = \mu_A(F_A(\mathfrak{n}^{\perp}))$$

Proof. Consider the case where $\mathfrak{m} = \mathfrak{s}_i \dots \mathfrak{s}_j$ so that $F_B(\mathfrak{m}) = \mathfrak{ms}_j B'_j \subseteq B'_i$ and $F_B(\mathfrak{m}^{\perp}) = \mathfrak{m}^{\perp}\mathfrak{s}_i^{\perp}B_i \subseteq B_j$ (all other cases are analogous) and let $\omega = |z - w|^{-4} du dv dx dy$ be the invariant form on the space of geodesics. Note that $\mathfrak{m}^{\perp} = \mathfrak{dm}^{-1}\mathfrak{d}$. We have

$$\mu_B(F_B(\mathfrak{m})) = \int_{\mathcal{B}'_i} \int_{F_B(\mathfrak{m})} \omega = \int_{\mathcal{B}'_i} \int_{\mathfrak{ms}_j B'_j} \omega$$

while

$$\begin{split} \mu_B(F_B(\mathfrak{m}^{\perp})) &= \int_{\mathcal{B}_j} \int_{F(\mathfrak{m}^{\perp})} \omega = \int_{\mathcal{B}_j} \int_{\mathfrak{m}^{\perp} \mathfrak{s}_i^{\perp} B_i} \omega = \int_{\mathcal{B}_j} \int_{\mathfrak{dm}^{-1} \mathfrak{ds}_i^{\perp} B_i} \omega = \int_{\mathfrak{m} \mathfrak{dB}_j} \int_{\mathfrak{ds}_i^{\perp} B_i} \omega \\ &= \int_{\mathfrak{m} \mathfrak{dB}_j} \int_{\mathfrak{s}_i \mathfrak{dB}_i} \omega = \int_{\mathfrak{m} \mathfrak{s}_j B'_j} \int_{\mathfrak{s}_i A_i} \omega = \int_{\mathfrak{m} \mathfrak{s}_j B'_j} \int_{\mathcal{B}'_i} \omega. \end{split}$$

Here we are using the fact that $\omega = |z - w|^{-4} du dv dx dy$ is the invariant form and the relations in (4.1).

Lemma 4.4.6. For \mathfrak{m} , \mathfrak{n} in normal form (associated to T_B and T_A respectively), there are equalities of measure

$$\mu_B(F_B(\mathfrak{m})) = \mu_A(F_A(\mathfrak{m}^{-1})), \quad \mu_A(F_A(\mathfrak{n})) = \mu_B(F_B(\mathfrak{n}^{-1})).$$

Proof. Consider the case where $\mathfrak{m} = \mathfrak{s}_i \dots \mathfrak{s}_j$ so that $F_B(\mathfrak{m}) = \mathfrak{m}\mathfrak{s}_j B'_j \subseteq B'_i$ and $F_A(\mathfrak{m}^{-1}) = \mathfrak{m}^{-1}\mathfrak{s}_i A_i \subseteq A_j$ (all other cases are analogous) and let $\omega = |z - w|^{-4} du dv dx dy$ be the invariant

form on the space of geodesics. Then

$$\mu_B(F(\mathfrak{m})) = \int_{\mathcal{B}'_i} \int_{F_B(m)} \omega = \int_{\mathcal{B}'_i} \int_{\mathfrak{ms}_j B'_j} \omega = \int_{\mathfrak{s}_i A_i} \int_{\mathfrak{m} \mathcal{A}_j} \omega = \int_{\mathfrak{m}^{-1} \mathfrak{s}_i A_i} \int_{\mathcal{A}_j} \omega$$

while

$$\mu_A(F(\mathfrak{m}^{-1})) = \int_{F_A(\mathfrak{m}^{-1})} \int_{\mathcal{A}_j} \omega = \int_{\mathfrak{m}^{-1}\mathfrak{s}_i A_i} \int_{\mathcal{A}_j} \omega.$$

Here we are using the fact that $\omega = |z - w|^{-4} du dv dx dy$ is the invariant form and the relations in (4.1).

4.5 Cubeoctahedral continued fractions over $\mathbb{Q}(\sqrt{-2})$

In this section, we repeat some of the arguments of the last section for the ideal right-angled regular cubeoctahedron with vertices

$$0, 1, \infty, \frac{-1}{\sqrt{-2}}, \frac{1+\sqrt{-2}}{2}, \frac{\sqrt{-2}-1}{\sqrt{-2}}, \sqrt{-2}, 1+\sqrt{-2}, \frac{1}{1-\sqrt{-2}}, \frac{\sqrt{-2}}{1+\sqrt{-2}}, \frac{-1+\sqrt{-2}}{1+\sqrt{-2}}, \frac{2}{1-\sqrt{-2}}.$$

Hyperbolic 3-space can be tessellated by this polyhedron, and the tessellation is invariant under the group $GL_2(\mathbb{Z}[\sqrt{-2}])$. The associated continued fractions have applications to Diophantine approximation over the field $\mathbb{Q}(\sqrt{-2})$.

4.5.1 Dual pair of dynamical systems, invertible extension, and invariant measures

At the end of the introduction to [Sch11], Schmidt remarks that there is an ergodic theory for his continued fractions over $\mathbb{Q}(\sqrt{-2})$ similar to that over $\mathbb{Q}(i)$ in [Sch82]. Here is a version of that in terms of inversions in eight circles (with cubic tangency) and six 'dual' circles (with octahedral tangency).

Consider the following 14 Möbius transformations defined over $\mathbb{Z}[\sqrt{-2}]$

$$\mathfrak{s}_1 = \frac{(1+2\sqrt{-2})\bar{z}-4}{2\bar{z}-1+2\sqrt{-2}}, \ \mathfrak{s}_2 = \frac{\bar{z}}{2\bar{z}-1}, \ \mathfrak{s}_3 = -\bar{z}+2,$$

$$\mathfrak{s}_{4} = -\bar{z}, \ \mathfrak{s}_{5} = \frac{(1+2\sqrt{-2})\bar{z}-2}{4\bar{z}-1+2\sqrt{-2}}, \ \mathfrak{s}_{6} = \frac{(3+2\sqrt{-2})\bar{z}-4}{4\bar{z}-3+2\sqrt{-2}},$$
$$\mathfrak{t}_{1} = \bar{z}+2\sqrt{-2}, \ \mathfrak{t}_{2} = \bar{z}, \ \mathfrak{t}_{3} = \frac{(5-2\sqrt{-2})\bar{z}+4\sqrt{-2}}{-4\sqrt{-2}\bar{z}+5+2\sqrt{-2}}, \ \mathfrak{t}_{4} = \frac{(3-2\sqrt{-2})\bar{z}+2\sqrt{-2}}{-4\sqrt{-2}\bar{z}+3+2\sqrt{-2}},$$
$$\mathfrak{t}_{5} = \frac{(3-2\sqrt{-2})\bar{z}+4\sqrt{-2}}{-2\sqrt{-2}\bar{z}+3+2\sqrt{-2}}, \ \mathfrak{t}_{6} = \frac{\bar{z}}{-2\sqrt{-2}\bar{z}+1}, \ \mathfrak{t}_{7} = \frac{(1-2\sqrt{-2})\bar{z}+2\sqrt{-2}}{-2\sqrt{-2}\bar{z}+1+2\sqrt{-2}}, \ \mathfrak{t}_{8} = \frac{3\bar{z}+2\sqrt{-2}}{-2\sqrt{-2}\bar{z}+3},$$

inversions in the circles whose interiors are A_i and B_i (see Figures 4.18, 4.19). These generate a discrete group Γ of isometries of hyperbolic three-space (of finite covolume), reflections in the sides of an ideal, right-angled cubeoctahedron. The cubeoctahedral reflection group Γ is the kernel of the map

$$PGL_2(\mathbb{Z}[\sqrt{-2}]) \rtimes \langle \mathfrak{c} \rangle \to PGL_2(\mathbb{Z}[\sqrt{-2}]/(2)),$$

(\mathfrak{c} complex conjugation), similar to the Gaussian situation of the previous section.

We define two dynamical systems, $T_A, T_B : \mathbb{C} \to \mathbb{C}$

$$T_A(w) = \begin{cases} \mathfrak{s}_i w & w \in A_i \\ \mathfrak{t}_i w & w \in A'_i \end{cases}$$
$$T_B(z) = \begin{cases} \mathfrak{t}_i z & z \in B_i \\ \mathfrak{s}_i z & z \in B'_i \end{cases}$$

and use these to coordinatize the plane in two different ways using the alphabet $\{\mathfrak{s}_j, \mathfrak{t}_j\}$. We write $w = \mathfrak{w} = \prod_{i=1}^{\infty} \mathfrak{w}_i$ if $T_A^i(w) = \mathfrak{w}_i T_A^{i-1} w$ and similarly $z = \mathfrak{z} = \prod_{i=1}^{\infty} \mathfrak{z}_i$ if $T_B^i(z) = \mathfrak{z}_i T_B^{i-1} z$. With these coordinates, T_A , T_B are the shift maps. [The words are in "normal form" as described in 4.2.3] Both T_A and T_B have 12 fixed points (the vertices of the cubeoctahedron) and rational points go to them in finite time, depending once again on their character modulo 2 (12 coprime pairs from $\mathbb{Z}[\sqrt{-2}]/(2)$). See Figure 4.20 for the A and B "super-packings."

We now define an invertible extension T of T_A and T_B (T will extend T_B and T^{-1} will extend T_A), defined on a subset of the space of geodesics of hyperbolic 3-space $P^1(\mathbb{C}) \times P^1(\mathbb{C}) \setminus \text{diag.}$ Define



Figure 4.18: Regions in the definition of T_A , boundaries given by the fixed circles of the \mathfrak{s}_i .



Figure 4.19: Regions in the definition of T_B , boundaries given by the fixed circles of the t_i .



Figure 4.20: Dual super-packings for the cube octahedron associated to $\mathbb{Q}(\sqrt{-2}).$

the following groupings of the regions A_i, A'_i, B_i, B'_i

$$\mathcal{A}_{i} = (\cup_{B'_{j} \cap A_{i} = \emptyset} B'_{j}) \cup (\cup_{B_{j} \cap A_{i} = \emptyset} B_{j})$$
$$\mathcal{A}'_{i} = (\cup_{B'_{j} \cap A'_{i} = \emptyset} B'_{j}) \cup (\cup_{B_{j} \cap A'_{i} = \emptyset} B_{j})$$
$$\mathcal{B}_{i} = (\cup_{A'_{j} \cap B_{i} = \emptyset} A'_{j}) \cup (\cup_{A_{j} \cap B_{i} = \emptyset} A_{j})$$
$$\mathcal{B}'_{i} = (\cup_{A'_{j} \cap B'_{i} = \emptyset} A'_{j}) \cup (\cup_{A_{j} \cap B'_{i} = \emptyset} A_{j})$$

(i.e. for each A region \mathcal{A} is the union of all the B regions disjoint from it, similarly for the \mathcal{B}). The space of geodesics on which T will be defined is

$$\mathcal{G} = (\cup_i \mathcal{A}_i \times A_i) \cup (\cup_i \mathcal{A}'_i \times A'_i) = (\cup_i \mathcal{B}_i \times B_i) \cup (\cup_i \mathcal{B}'_i \times B'_i)$$

with

$$T(w,z) = (\mathfrak{z}_1 \prod_{j=1}^{\infty} \mathfrak{w}_j, \prod_{j=2}^{\infty} \mathfrak{z}_j) = \begin{cases} (\mathfrak{t}_i w, \mathfrak{t}_i z) = (\mathfrak{t}_i w, T_B(z)) & z \in B_i \\ (\mathfrak{s}_i w, \mathfrak{s}_i z) = (\mathfrak{s}_i w, T_B(z)) & z \in B'_i \end{cases}$$

or equivalently

$$T^{-1}(w,z) = (\prod_{j=2}^{\infty} \mathfrak{w}_j, \mathfrak{w}_1 \prod_{j=1}^{\infty} \mathfrak{z}_j) = \begin{cases} (\mathfrak{s}_i w, \mathfrak{s}_i z) = (T_A(w), \mathfrak{s}_i z) & w \in A_i \\ (\mathfrak{t}_i w, \mathfrak{t}_i z) = (T_A(w), \mathfrak{t}_i z) & w \in A'_i \end{cases}$$

One can verify that T is a bijection

$$T(\mathcal{B}_i \times B_i) = A'_i \times \mathcal{A}'_i,$$
$$T(\mathcal{B}'_i \times B'_i) = A_i \times \mathcal{A}_i$$

(noting $\mathcal{G} = \bigcup_i (A_i \times \mathcal{A}_i \cup A'_i \times \mathcal{A}'_i) = \bigcup_i (\mathcal{B}_i \times B_i \cup \mathcal{B}'_i \times B'_i)$). The space of geodesics has an isometry invariant measure $|z - w|^{-4} du dv dx dy$ (w = u + iv, z = x + iy) and since T is a bijection defined piecewise by isometries, this measure is T-invariant on \mathcal{G} . The pushforward of this measure onto the first and second coordinates will provide invariant measures μ_A , μ_B for T_A and T_B

$$d\mu_A = f_A(w)dudv = \begin{cases} dudv \int_{\mathcal{A}_i} |z - w|^{-4} dxdy & w \in A_i \\ dudv \int_{\mathcal{A}'_i} |z - w|^{-4} dxdy & w \in A'_i \end{cases},$$

$$d\mu_B = f_B(z)dxdy = \begin{cases} dxdy \int_{\mathcal{B}_i} |z - w|^{-4}dudv & w \in B_i \\ dxdy \int_{\mathcal{B}'_i} |z - w|^{-4}dudv & w \in B'_i \end{cases}.$$

Computing these integrals on the triangular (A'_i) and quadrangular (B'_i) regions gives a multiple of hyperbolic area

$$f_A(w) = \begin{cases} \frac{\pi}{4(v-\sqrt{2})^2} & w \in A_1' \\ \frac{\pi}{4v^2} & w \in A_2' \end{cases},$$

$$f_A(w) = \frac{\pi\rho^2}{(\rho^2 - d^2)^2}, \ d, \rho = \begin{cases} d = |w - (1/2 + 5\sqrt{-2}/8)|, \rho = \sqrt{2}/8 & w \in A_3' \\ d = |w - (1/2 + 3\sqrt{-2}/8)|, \rho = \sqrt{2}/8 & w \in A_4' \\ d = |w - (1 + 3\sqrt{-2}/4)|, \rho = \sqrt{2}/4 & w \in A_5' \\ d = |w - 3\sqrt{-2}/4|, \rho = \sqrt{2}/4 & w \in A_6' \\ d = |w - (1 + \sqrt{-2}/4)|, \rho = \sqrt{2}/4 & w \in A_7' \\ d = |w - \sqrt{-2}/4|, \rho = \sqrt{2}/4 & w \in A_8' \end{cases}$$

$$f_B(z) = \begin{cases} \frac{\pi}{4(x-1)^2} & z \in B'_3 \\ \frac{\pi}{4x^2} & z \in B'_4 \end{cases},$$

$$f_B(z) = \frac{\pi\rho^2}{(\rho^2 - d^2)^2}, \ d, \rho = \begin{cases} d = |z - (1/2 + \sqrt{-2})|, \rho = 1/2 & w \in B'_1 \\ d = |z - 1/2|, \rho = 1/2 & w \in B'_2 \\ d = |z - (1/2 + \sqrt{-2})|, \rho = 1/4 & w \in B'_5 \\ d = |z - (1/2 + \sqrt{-2})|, \rho = 1/4 & w \in B'_6 \end{cases},$$

with the mass of each piece being $\mu_A(A'_i) = \pi^2/4$, $\mu_B(B'_i) = \pi^2/2$.

On the circular regions A_i , B_i , we have

$$f_A(w) = \begin{cases} G(u, v) & w \in A_4 \\ G(\mathfrak{m}w) \|\mathfrak{m}w\| & w \in A_i \end{cases},$$
$$f_B(z) = \begin{cases} H(x, y) & z \in B_2 \\ H(\mathfrak{n}z) \|\mathfrak{n}z\| & z \in B_i \end{cases},$$

76

where

$$h(a,b) = \frac{\arctan(a/b)}{4a^2} - \frac{1}{4ab},$$

$$H(a,b) = h(a,b) + h(1-a,b) + h(a^2 - a + b^2, b),$$

$$G(a,b) = H(b/\sqrt{2}, a/\sqrt{2}) + H((b-1)/\sqrt{2}, a/\sqrt{2})$$

and the \mathfrak{m} and \mathfrak{n} are chosen to take $A_i \times A_i$ or $\mathcal{B}_i \times B_i$ to $\mathcal{A}_4 \times A_4$ or $\mathcal{B}_2 \times B_2$. The integrals are obtained by noting that for any Möbius transformation \mathfrak{m}

$$\phi(z) = \int_{\mathfrak{m}R} |z - w|^{-4} du dv \Rightarrow \phi(\mathfrak{m}z) ||\mathfrak{m}z|| = \int_R |z - w|^{-4} du dv.$$

The following act as permutations of $A_i \times A_i$ or $\mathcal{B}_i \times B_i$ and generate the symmetries (octahedral/cubic):

$$\mathfrak{m}_{(1326)} = \frac{1}{w - \sqrt{-2}}, \ \mathfrak{m}_{(165)} = \frac{-1}{w - 1}, \ \mathfrak{m}_{(12)} = \bar{w} + \sqrt{-2}, \ \mathfrak{m}_{(34)(56)} = -\bar{w} + 1,$$
$$\mathfrak{n}_{(3574)} = \frac{1}{z - \sqrt{-2}}, \ \mathfrak{n}_{(167)} = \mathfrak{n}_{(458)} = \frac{-1}{z - 1}, \ \mathfrak{n}_{(12)(34)(57)(68)} = \bar{z} + \sqrt{-2}, \ \mathfrak{n}_{(58)(67)} = -\bar{z} + 1.$$

The mass of the circular regions is $\mu_A(A_i) = \pi^2/2$, $\mu_B(B_i) = \pi^2/4$. Hence the total masses are $\mu_A(\mathbb{C}) = 5\pi^2$, $\mu_B(\mathbb{C}) = 11\pi^2/2$.

The "Apollonian" structure of a group isomorphic to the group generated by the "swaps" \mathfrak{s}_i (and similar groups for the other Euclidean imaginary quadratic fields) is explored in [Sta18].

4.5.2 Geometry of the first approximation constant for $\mathbb{Q}(\sqrt{-2})$

Recall that the "good" rational approximations to an irrational $z \in \mathbb{C}$

$$|z - p/q| \le C/|q|^2$$

are determined by the collection of horoballs

$$B_C(p/q) = \{(z,t) \in H^3 : |z - p/q|^2 + (t - C/|q|^2)^2 \le C^2/|q|^4\},\$$
$$B_C(\infty) = \{(z,t) \in H^3 : t \ge 1/2C\},\$$

through which the geodesic $\overrightarrow{\infty z}$ passes (or through which any geodesic \overrightarrow{wz} eventually passes). Over $\mathbb{Q}(\sqrt{-2})$, the smallest value of C with the property that every irrational z has infinitely many rational approximations satisfying the above inequality was determined by Perron in [Per33]. Here we give a short proof of this fact using the geometry of the ideal cubeoctahedron.

Proposition 4.5.1. Every $z \in \mathbb{C} \setminus \mathbb{Q}(\sqrt{-2})$ has infinitely many rational approximations $p/q \in \mathbb{Q}(\sqrt{-2})$ such that

$$\left|z - \frac{p}{q}\right| \le \frac{C}{|q|^2}, \quad C = \frac{1}{\sqrt{2}}.$$

The constant $1/\sqrt{2}$ is the smallest possible, as witnessed by $z = \frac{1+i}{\sqrt{2}}$.

Proof. $GL_2(\mathbb{Z}[\sqrt{-2}])$ acts transitively on the faces of the same shape, so we only need cover the triangular face with vertices $\{0, 1, \infty\}$ and the square face that we identify with the ideal quadrilateral with vertices $\{0, -1/\sqrt{-2}, \sqrt{-2}, \infty\}$. The value of C needed for the former is $1/\sqrt{3}$ and for the latter is $1/\sqrt{2}$ (see Figure 4.21).

The point not covered by the open horoballs of parameter $1/\sqrt{2}$ on the square face above the imaginary axis is $(-1/\sqrt{-2}, 1/\sqrt{2})$. An exceptional geodesic entering orthogonally at this point has radius $1/\sqrt{2}$ and exits orthogonally through the opposite square face at the point $(\frac{2}{3} + \frac{i}{\sqrt{2}}, \frac{1}{3\sqrt{2}})$. This geodesic has irrational endpoints $\frac{\pm 1+i}{\sqrt{2}}$. Once again, shrinking the horoballs allows this geodesic through, so that $C = 1/\sqrt{2}$ is optimal.

4.5.3 Quality of approximation

We now consider the quality of approximation of the algorithms above. The 12 sequences of convergents (orbits of the 12 vertices of the cubeoctahedron) to a number z_0 include all good rational approximations as described in the next theorem. The proof follows the same reasoning used in 4.4.3, but the constant disagrees with that given (without proof) in [Sch11]. There ([Sch11] Theorem 2.4) the constant is $C = \frac{2\sqrt{2}}{1+\sqrt{17}}$, smaller than the constant we give below. While our algorithms and those of [Sch11] are both based on the same circle packings, we keep track of



Figure 4.21: Horoball covering of the ideal square face with vertices $\{0, -1/\sqrt{-2}, \sqrt{-2}, \infty\}$ with coordinates (z, t).

more approximations (12 sequences of convergents) than he does (the 5 sequences of convergents necessary to describe the Farey sets of the partition), which explains the discrepancy.

Theorem 4.5.2. If p/q is such that

$$|z_0 - p/q| < \frac{C}{|q|^2}, \quad C = \frac{2\sqrt{2}}{1 + \sqrt{2} + \sqrt{3}} = 0.682162754\dots,$$

then p/q is a convergent to z_0 (with respect to both T_A and T_B). Moreover, the constant C is the largest possible.

Proof. The proof follows the reasoning of 4.4.3. We consider when a rational p/q appears for the first time when approximating $z_0 \in \mathbb{C}$, send p/q to infinity and the boundary circles for the cubeoctahedron to the 'basic' configuration, and see what happens. For brevity, we focus on the worst case scenario when p/q arises from 'swapping' into a polygonal region in two different ways, as in Figure 4.22. By sending p/q to ∞ by an appropriate $\gamma \in B[-2] = GL_2(\mathbb{Z}[\sqrt{-2}]) \rtimes \langle \mathfrak{c} \rangle$ (i.e. send our approximating cubeoctahedron back to the fundamental cubeoctahedron via the inverse of T_A^n or T_B^n if p/q is an *n*th convergent to z_0 , then permuting the fundamental cubeoctahedron) we can send the arrangments of Figure 4.22 to the base configurations of Figure 4.23. A neighborhood of p/q of the form

$$\left|z - \frac{p}{q}\right| \le \frac{C}{|q|^2}$$

(say containing z_0) is now a neighborhood of infinity of the form

$$|z - \gamma(\infty)| \ge \frac{1}{C}.$$

The point $\gamma(\infty)$ is constrained to lie in the shaded region of Figure 4.23, say by considering the black circle and the circles A and R respectively. Since we swapped into the polygonal regions, z_0 must not be in the disks

A, B, C, D (for
$$T_A$$
), R, S, T, U, V, W (for T_B),

of Figure 4.22 (oriented with the labels in the interior). After applying γ , this says that $\gamma(z_0)$ must be exterior to the disks with the same labels in Figure 4.23. The quantity

$$\frac{1}{C} = \frac{1}{2} + \sqrt{\left(\frac{1}{2}\right)^2 + \left(\frac{1}{2\sqrt{2}}\right)^2} + \frac{1}{2\sqrt{2}}$$

is the diameter of the union of the disk C (respectively U) with the disk bounded by the black circle in Figure 4.23. This is the radius of the smallest circle centered in the shaded region whose interior avoids the labeled disks (i.e. forcing $\gamma(z_0)$ to be in the appropriate neighborhood of infinity).

4.6 Antiprisms and questions

The construction and examples of the previous sections raise some questions. For instance:

- Given a two-colorable polyhedron, is there a natural field of definition for the associated reflection group? If so, what is it?
- Does the associated continued fraction algorithm have any arithmetic significance? If so, what is it?
- The three main examples of this chapter (ideal triangle, octahedron, cubeoctahedron) are all 2-congruence subgroups. Does this have any significance?

For example, some circles defining reflection groups for regular antiprisms are shown in Figure 4.24. The circles have (center, radius) equal to (where the antiprism consists of two *n*-gons and 2n triangles, and $1 \le k \le n$):

$$(0,1), \left(\frac{\zeta_n^k}{1+\sin(\pi/n)}, \frac{\sin(\pi/n)}{1+\sin(\pi/n)}\right) \text{ red circles}, \\ \left(0, \frac{\cos(\pi/n)}{1+\sin(\pi/n)}\right), \left(\frac{\zeta_{2n}^k}{\cos(\pi/n)}, \tan(\pi/n)\right) \text{ blue circles}.$$

These coordinates are somewhat arbitrary (i.e. whatever I could conveniently write down), but they are defined over the cyclotomic field $\mathbb{Q}(\zeta_{2n})$.

However, the n = 3 case is the ideal right-angled octahedron, and the n = 4 case is the ideal right-angled cubeoctahedron cut in half. This is interesting because we know the octahedron can be



Figure 4.22: A rational number p/q appearing as an approximation to z_0 for the first time with respect to T_A (top) or T_B (bottom). In particular z_0 must be outside the labeled disks (the interiors of A and R include infinity).



Figure 4.23: Result of applying γ to the configurations of Figure 4.22, with $\gamma(\infty)$ in the shaded region. If p/q were an approximation to z_0 , then z_0 must be outside of the labeled disks by definition of the algorithms associated to T_A and T_B .



Figure 4.24: Antiprism reflection generators, n = 3, 4, 69.

defined over $\mathbb{Q}(\sqrt{-1})$ and the cubeoctahedron can be defined over $\mathbb{Q}(\sqrt{-2})$ where the orbit of the vertices gives the enitire field in a systematic way. So, somehow, the fields $\mathbb{Q}(\sqrt{-1})$ and $\mathbb{Q}(\sqrt{-2})$ can be "embedded" into these cyclotomic fields. At the very least this gives an injection of sets, but perhaps it preserves some other structure, such as the heights of the points.

Finally, we note a general construction for constructing ideal polyhedral cell structures on \mathcal{H}^3 that are invariant under the Bianchi groups $PSL_2(\mathcal{O}_K)$, K imaginary quadratic (cf. [Yas10] for the construction, applications, and some data). For the five Euclidean imaginary quadratic fields, this complex consists of one orbit of a single polyhedron (the octahedron, cubeoctahedron, tetrahedon, triangular prism, and truncated tetrahedron for -d = 1, 2, 3, 7, 11 and $K = \mathbb{Q}(\sqrt{-d})$), which is right-angled in the first two examples.

Chapter 5

Characterizations and examples of badly approximable numbers

5.1 Introduction

This chapter gives characterizations and examples of badly approximable vectors in various settings, starting with a discussion of the prototypical one-dimensional case in the next section. Much of this chapter comes from [Hin18a] and [Hin18b]. Here is an outline, followed by a general discussion.

- (5.2) Badly approximable real numbers $\xi = [a_0; a_1, a_2, \ldots]$ over \mathbb{Q} are characterized by bounded partial quotients $|a_n| \leq M$, or (more or less equivalently) bounded geodesic trajectories on the modular surface, $\{\xi + e^{-t}i : t \geq 0\} \mod SL_2(\mathbb{Z})$. Obvious examples of badly approximable numbers are given by quadratic irrationals, whose partial quotients are eventually periodic, and whose associated trajectory is asymptotic to a closed (periodic) geodesic on the modular surface, namely the projection of the geodesic with ideal endpoints ξ , ξ' (prime denoting Galois conjugate). There is also another, simpler, argument of Liouville (concerning rational approximations to algebraic numbers) showing that quadratic irrationalities are badly approximable.
- (5.3) In this section we show that complex numbers badly approximable over the Euclidean imaginary quadratic fields are characterized by bounded partial quotients in their nearest integer continued fraction expansion.
- (5.4) Here we show that badly approximable complex numbers over $\mathbb{Q}(\sqrt{-1})$ and $\mathbb{Q}(\sqrt{-2})$

are characterized by orbits bounded away from fixed points under the octahedral and cubeoctahedral continued fraction algorithms of chapter 4.

- (5.5) Here we state a version of Mahler's criterion and the Dani correspondence (cf. chapter
 2) characterizing badly approximable vectors. See [EGL16] for proofs, or 2.4 for details in the imaginary quadratic case.
- (5.6) In this section we give specific examples of badly approximable vectors associated to compact totally geodesic subspaces of the "modular" spaces

$$\Gamma \backslash G/K = SL_2(\mathcal{O}_F) \backslash (\mathcal{H}^2)^r \times (\mathcal{H}^3)^s.$$

These have simple number theoretic descriptions in terms of totally indefinite anisotropic rational binary quadratic and Hermitian forms. These examples are natural generalizations of real quadratic irrationals from the one-dimensional situation of 5.2.

- (5.7) While the previous section discussed the quadratic/Hermitian examples from the point of view of the Dani correspondence, here we give arguments using nearest integer continued fractions when F is a Euclidean imaginary quadratic field.
- (5.8) Here we discuss the Hermitian examples from the point of view of right-angled continued fractions over $\mathbb{Q}(\sqrt{-1})$ and $\mathbb{Q}(\sqrt{-2})$.
- (5.9) This section gives a Liouville-type argument showing that our quadratic/Hermitian examples are badly approximable. In the imaginary quadratic case, we give estimates of the approximation constants for badly approximable Hermitian zeros.
- (5.10) Finally, we point out that among the Hermitian examples there are many *algebraic* vectors and give a characterization of these, with an emphasis on the imaginary quadratic case (if only for the pictures).

We end this introduction with a general discussion and some context. A number field F of degree r + 2s embeds naturally in the product of its Archimedean completions $F \otimes_{\mathbb{Q}} \mathbb{R} \cong \mathbb{R}^r \times \mathbb{C}^s$.

Given a vector $\mathbf{z} = (z_1, \ldots, z_{r+s}) \in \mathbb{R}^r \times \mathbb{C}^s$, one can ask how well \mathbf{z} can be approximated by elements of F. Following [EGL16] and [KL16], we will measure the quality of approximation by

$$\max_{i} \{ |q_i| \} \max_{i} \{ |q_i z_i - p_i| \}, \ p/q \in F, \ p, q \in \mathcal{O}_F,$$

where p_i and q_i are the images of p and q under r + s inequivalent embeddings $\sigma_i : F \to \mathbb{C}$ and $|\cdot|$ is the usual absolute value in \mathbb{R} or \mathbb{C} . The measure above is meaningful in the sense that all irrational vectors have infinitely many "good" approximations as demonstrated by the following Dirichlet-type theorem.

Theorem 5.1.1 (cf. [Quê91], Theorem 1). There is a constant C depending only on F such that for any $\mathbf{z} \notin F$

$$\max_{i} \{ |q_i| \} \max_{i} \{ |q_i z_i - p_i| \} \le C$$

has infinitely many solutions $p/q \in F$.

In what follows, we will give some explicit examples showing that the above theorem fails if the constant is decreased, i.e. there are *badly approximable* vectors, \mathbf{z} such that there exists C' > 0with

$$\max_{i} \{ |q_i| \} \max_{i} \{ |q_i z_i - p_i| \} \ge C'$$

for all $p/q \in F$. Our examples come from "obvious" compact totally geodesic subspaces of

$$SL_2(\mathcal{O}_F)\backslash (\mathcal{H}^2)^r \times (\mathcal{H}^3)^s,$$

where \mathcal{H}^n is hyperbolic *n*-space. Namely, these examples are associated to totally indefinite anisotropic *F*-rational binary quadratic forms over any number field (Proposition 5.6.1) and totally indefinite anisotropic *F*-rational binary Hermitian forms over CM fields (Proposition 5.6.3). Among the examples are algebraic vectors, i.e. vectors whose entries generate a non-trivial finite extension of *F*, including non-quadratic vectors in the CM case (Corollary 5.10.1). This is interesting in light of the following variation on Roth's theorem (which can be deduced from the Subspace Theorem for number fields). **Theorem 5.1.2** (cf. [Sch75b], Theorem 3). Suppose $\mathbf{z} \notin F$ has algebraic coordinates. Then for all $\epsilon > 0$, there exists a constant C' > 0 depending on \mathbf{z} and ϵ such that

$$\max_{i} \{ |q_i| \}^{1+\epsilon} \max_{i} \{ |q_i z_i - p_i| \} \ge C'$$

for all $p/q \in F$.

The "linear forms" notion of badly approximable defined above implies

$$\max_{i} \{ |z_i - p_i/q_i| \} \ge \frac{C'}{\max_i \{ |q_i|^2 \}} \text{ for all } p/q \in F,$$

which is perhaps the first notion of badly approximable that comes to mind. The two notions are equivalent when F has only one infinite place (i.e. $F = \mathbb{Q}$ or $\mathbb{Q}(\sqrt{-d})$) since the absolute value is multiplicative. (The notions are also equivalent for real quadratic and complex quartic F, [KL16] Proposition A.2.) However, in larger number fields it seems that some choice must be made and the linear choice appears naturally in the proof of Theorem 5.5.1.

Simultaneous approximation in this sense seems natural and has been explored by various authors, e.g. [EGL16], [Hat07], [KL16], [Quê91], [Sch75b], [Bur92]. Among known facts, we note that the set of badly approximable vectors has Lebesgue measure zero, full Hausdorff dimension, and is even "winning" when restricted to curves and various fractals in $\mathbb{R}^r \times \mathbb{C}^s$ ([EGL16], [KL16], [ESK10]).

5.2 Simple continued fractions, the modular surface, and quadratic irrationalities

Below are two characterizations of badly approximable real numbers. One uses continued fractions and the other bounded geodesic trajectories on the modular surface.

Theorem 5.2.1. An irrational real number $\xi = [a_0, a_1, a_2, ...]$ is badly approximable if and only if its partial quotients a_n are bounded.

Proof. If ξ is badly approximable, $|\xi - p/q| \ge C'/q^2$, then in particular

$$\frac{C'}{q_n^2} \le |\xi - p_n/q_n| = \frac{1}{q_n^2([a_{n+1}; a_{n+2}, \ldots] + [0; a_n, \ldots, a_1])} \le \frac{1}{q_n^2 a_{n+1}}$$
$$a_{n+1} \le 1/C'.$$

Conversely, if the partial quotients are bounded, $\sup_n \{a_n\} \leq M$, then for any p/q with $0 < q \leq q_n$

$$\begin{aligned} |\xi - p/q| &\geq |\xi - p_n/q_n| = \frac{1}{q_n^2(q_{n+1}/q_n + \xi_{n+1})} \\ &= \frac{1}{q_n^2([0; a_{n+2,\dots}] + [a_{n+1}; a_n, \dots, a_1])} \geq \frac{1}{q_n^2(a_{n+1} + 2)} \geq \frac{1}{q_n^2(M + 2)}, \end{aligned}$$

using the fact that the convergents p_n/q_n are the best approximations for the first inequality. \Box

Theorem 5.2.2. The number ξ is badly approximable if and only if the trajectory

$$\Omega_{\xi} = \left\{ SL_2(\mathbb{Z}) \begin{pmatrix} 1 & \xi \\ 0 & 1 \end{pmatrix} \begin{pmatrix} e^{-t} & 0 \\ 0 & e^t \end{pmatrix} : t \ge 0 \right\} \subseteq SL_2(\mathbb{Z}) \backslash SL_2(\mathbb{R})$$

is bounded (precompact).

Proof. Suppose ξ is badly approximable with $|q(q\xi + p)| \ge C'$ for all p, q. If

$$\left\| (q \ p) \begin{pmatrix} 1 \ \xi \\ 0 \ 1 \end{pmatrix} \begin{pmatrix} e^{-t} & 0 \\ 0 & e^t \end{pmatrix} \right\|_{\infty} = \| (e^{-t}q, e^t(q\xi + p)) \|_{\infty} < \sqrt{C'},$$

then taking the product of the coordinates gives a contradiction. Hence Ω_{ξ} is precompact by Mahler's criterion.

Conversely, if ξ is not badly approximable, there exist sequences a_n, b_n such that $|b_n(b_n\xi + a_n)| \leq 1/n^2$. If t_n is such that $e^{-t_n}|b_n| \leq 1/n$, then

$$||(e^{-t_n}b_n, e^{t_n}(b_n\xi + a_n))||_{\infty} \le 1/n,$$

and the trajectory Ω_{ξ} is not bounded (Mahler's criterion again).

The obvious (and conjecturally only) badly approximable real algebraic numbers are quadratic irrationals. Below we present three proofs of this fact, all of which we will revisit in the sequel.

Lemma 5.2.3 (Liouville, 1844). If ξ is a real algebraic number of degree n > 1, then there is a constant A > 0 (depending on ξ) such that

$$\left|\xi - \frac{h}{k}\right| > \frac{A}{k^n}$$

(In particular, real quadratics are badly approximable.)

Proof. Suppose $p(x) \in \mathbb{Z}[x]$ is irreducible of degree n with $p(\xi) = 0$. Then

$$p(\xi) - p(h/k) = (\xi - h/k)p'(\alpha)$$

for some α between ξ and h/k by the mean value theorem. The left hand side is a non-zero rational number $(p(\xi) = 0 \text{ and } p \text{ is irreducible so } h/k \text{ is not a root})$ with denominator less than k^n so that we get

$$\frac{1}{k^n} \le \left| \xi - \frac{h}{k} \right| \sup\{ p'(x) : x \in (\xi - 1, \xi + 1) \}.$$

Theorem 5.2.4 (Lagrange, 1770). The partial quotients of a quadratic irrational are eventually periodic (hence bounded).

Proof. (Cf. [Khi97] Theorem 28.) Let

$$\begin{aligned} Q(x,y) &= Ax^2 + Bxy + Cy^2 = (x \ y) \begin{pmatrix} A & B/2 \\ B/2 & C \end{pmatrix} \begin{pmatrix} x \\ y \end{pmatrix}, \\ A, B, C \in \mathbb{Z}, \ \Delta(Q) &= AC - B^2/4 < 0, \end{aligned}$$

be an indefinite integral binary quadratic form, and Z(Q) its zero set (i.e. two points in $P^1(\mathbb{R})$). $GL_2(\mathbb{Z})$ acts by change of variable on Q, by Möbius transformations on Z(Q), and the map $Q \mapsto Z(Q)$ is $GL_2(\mathbb{Z})$ -equivariant:

$$Z(Q^g) = Z(g^t Qg) = g^{-1} \cdot Z(Q).$$

Note that the action on forms preserves the determinant Δ . For $\xi \in \mathbb{R}$, $\xi = [a_0; a_1, \ldots]$, let

$$g_n = \begin{pmatrix} p_n & p_{n-1} \\ q_n & q_{n-1} \end{pmatrix} = \begin{pmatrix} a_0 & 1 \\ 1 & 0 \end{pmatrix} \cdots \begin{pmatrix} a_n & 1 \\ 1 & 0 \end{pmatrix},$$

so that $T^n(\xi_0) = 1/g_n^{-1}\xi$. If $Q(\xi, 1) = 0$ for some indefinite integral binary quadratic form, let $Q_n = Q^{g_n} = A_n x^2 + B_n xy + C_n y^2$, and note that $Q_n(1,\xi_n) = 0$. We claim that the collection $\{Q_n : n \ge 0\}$ is finite. The form Q_n is given by

$$\begin{pmatrix} p_n & q_n \\ p_{n-1} & q_{n-1} \end{pmatrix} \begin{pmatrix} A & B/2 \\ B/2 & C \end{pmatrix} \begin{pmatrix} p_n & p_{n-1} \\ q_n & q_{n-1} \end{pmatrix},$$

so that its coefficients satisfy

$$A_{n} = Ap_{n}^{2} + Bp_{n}q_{n} + C_{n}q_{n}^{2},$$

$$B_{n} = 2Ap_{n}q_{n} + B(p_{n}q_{n-1} + p_{n-1}q_{n}) + Cq_{n}q_{n-1},$$

$$C_{n} = Ap_{n-1}^{2} + Bp_{n-1}q_{n-1} + Cq_{n-1}^{2} = A_{n_{1}}.$$

Because $|\xi - p_n/q_n| \le 1/q_n^2$, we have $q_n\xi - p_n = \delta_n/q_n$ for some δ_n with $|\delta_n| \le 1$. Substituting this into the above gives

$$A_{n} = A(q_{n}\xi - \delta_{n}/q_{n})^{2} + B(q_{n}\xi - \delta_{n}/q_{n})q_{n} + Cq_{n}^{2}$$

= $(A\xi^{2} + B\xi + C)q_{n} + A(\delta_{n}^{2}/q_{n}^{2} - 2\xi\delta_{n}) - B\delta_{n}$
= $A(\delta_{n}^{2}/q_{n}^{2} - 2\xi\delta_{n}) - B\delta_{n},$
 $|A_{n}| \leq |2A\xi| + |A| + |B|.$

Hence A_n , $C_n = A_{n-1}$, and $B_n = \pm \sqrt{A_n C_n - 4\Delta}$ are all bounded in terms of A, B, C, and ξ . This proves the claim and shows that the sequence ξ_n and a_n are eventually periodic.

Theorem 5.2.5. Real quadratic irrationals are badly approximable (via the Dani correspondence).

Proof. The closed (compact/periodic) geodesics on the modular surface are the projections of the geodesics in \mathcal{H}^2 joining conjugate real quadratic irrationals (upper half-plane model). In one direction, the fixed points $z = \frac{az+b}{cz+d}$ of a hyperbolic element of $SL_2(\mathbb{Z})$ are real quadratic irrationals. In the other, the stabilizer of the quadratic form associated to the minimal polynomial of a quadratic irrational is generated by a hyperbolic element (which can be written explicitly in terms of a fundamental solution to Pell's equation, cf. [Lan58] Theorem 202).

If $Q(\xi, 1) = 0$ for an integral form, then the geodesic trajectory Ω_{ξ} is asymptotic to the closed geodesic joining ξ , $\bar{\xi}$. Therefore Ω_{ξ} is bounded and ξ is badly approximable.

5.3 Bounded partial quotients (nearest integer)

Badly approximable complex numbers (over the Euclidean imaginary quadratic fields) can also be characterized by boundedness of their partial quotients.

Theorem 5.3.1. A number $z \in \mathbb{C} \setminus K$ is badly approximable if and only if its partial quotients a_n are bounded (if and only if the remainders z_n are bounded away from zero). In particular, if $|a_n| \leq \beta$ for all n and $p/q \in K$, then $|z - p/q| \geq C'/|q|^2$ where

$$C' = \frac{1}{\alpha(\beta+1)(\beta+\rho+1)}$$

and $\rho = \rho_d$ is as in 3.3.

Proof. If z is badly approximable, then there is a C' > 0 such that for each convergent p_n/q_n to z, the disk $|w - p_n/q_n| \le C'/|q_n|^2$ does not contain z. Mapping p_n/q_n to ∞ via g_n^{-1} , where

$$g_n = \begin{pmatrix} a_0 & 1 \\ 1 & 0 \end{pmatrix} \dots \begin{pmatrix} a_n & 1 \\ 1 & 0 \end{pmatrix} = \begin{pmatrix} p_n & p_{n-1} \\ q_n & q_{n-1} \end{pmatrix}$$

,

maps the disk $|w - p_n/q_n| \le C'/|q_n|^2$ to the region $|w + q_{n-1}/q_n| \ge 1/C'$, centered at $g_n^{-1}(\infty) = -q_{n-1}/q_n$. Because $g_n^{-1}(z)$ is inside the disk of radius 1/C' centered at $-q_{n-1}/q_n$ and $|-q_{n-1}/q_n| < 1$, we have

$$a_{n+1} + z_{n+1} = 1/z_n = g_n^{-1}(z),$$

 $|a_{n+1}| \le |z_{n+1}| + |g_n^{-1}(z)| \le \rho + 1 + 1/C'$

Hence a_{n+1} is bounded. See Figure 5.1 below for an illustration.

By Lemma 3.3.3, for z and p/q with $|q_{n-1}| < |q| \le |q_n|$ we have

$$\begin{split} \left| z - \frac{p_n}{q_n} \right| &\leq \alpha \left| z - \frac{p}{q} \right| \left| \frac{q}{q_n} \right| \leq \alpha \left| z - \frac{p}{q} \right| \frac{|q|^2}{|q_n|^2} \left| \frac{q_n}{q_{n-1}} \right| \\ &= \alpha \left| z - \frac{p}{q} \right| \frac{|q|^2}{|q_n|^2} \left| a_n + \frac{q_{n-2}}{q_{n-1}} \right| \\ &\leq \alpha \left| z - \frac{p}{q} \right| \frac{|q|^2}{|q_n|^2} (|a_n| + 1), \\ |q_n|^2 \left| z - \frac{p_n}{q_n} \right| &\leq \alpha (|a_n| + 1) |q|^2 \left| z - \frac{p}{q} \right|. \end{split}$$

This shows that if z has bounded partial quotients, then z is badly approximable if and only if it is badly approximable *by its convergents*. For approximation by convergents, we have

$$\begin{split} \left| z - \frac{p_n}{q_n} \right| &= \frac{1}{|q_n|^2 |z_n^{-1} + q_{n-1}/q_n|} \\ &= \frac{1}{|q_n|^2 |a_{n+1} + z_{n+1} + q_{n-1}/q_n|} \geq \frac{1}{|q_n|^2 (|a_{n+1}| + \rho + 1)}, \end{split}$$

showing that if the partial quotients of z are bounded, then z is badly approximable by convergents and therefore badly approximable. For an approximation constant, the above discussion gives $|z - p/q| \ge C'/|q|^2$ for any $p/q \in K$ where

$$C' = \frac{1}{\alpha(\beta+1)(\beta+\rho+1)}$$

and β is an upper bound for the $|a_n|$.

5.4 Orbits bounded away from fixed points (right-angled)

Below is a characterization of badly approximable numbers in terms of its orbit under the octahedral and cubeoctahedral continued fraction maps of chapter 4.

Theorem 5.4.1. A number $z_0 \in \mathbb{C} \setminus \mathbb{Q}(i)$ is badly approximable if and only if its *T*-orbit is bounded away from the fixed points (six vertices of the octahedron).

Proof. If z_0 is badly approximable, let p/q be a convergent to z_0 , without loss of generality in the orbit of ∞ . If $g \in \Gamma$ is the element corresponding to repeated application of T, $g(p/q) = \infty$, then



Figure 5.1: Over $\mathbb{Q}(\sqrt{-1})$, we have the points $p_n/q_n = g_n(\infty)$ and $-q_{n-1}/q_n = g_n^{-1}(\infty)$, along with the unit circle and its image under g_n (black), circles of radius 1/C' and $C'/|q_n|^2$ (red), and the lines defining V and their images under g_n (blue).

the disk $|z - p/q| < C/|q|^2$ (not containing z_0) gets mapped to the exterior of the disk centered at $g(\infty)$ (which is in one of the regions adjacent to $\frac{1}{1-i}$) of radius 1/C. Hence $g(z_0)$ is bounded. Similarly (or applying an automorphism) for the other fixed points. If z_0 is not badly approximable (say by convergents in the orbit of ∞), then the above argument shows that $g(z_0)$ is contained in the exterior of arbitrarily large circles centered near $\frac{1}{1-i}$. Again, ∞ is not special in this argument, just convenient. We can limit ourselves to approximation by convergents since they exhaust all rationals with $|z - p/q| < \frac{1}{(1+q/\sqrt{2})|q|^2}$.

An argument in the same vein as the previous proof furnishes a similar statement for the cubeoctahedral continued fractions.

Theorem 5.4.2. A number $z_0 \in \mathbb{C} \setminus \mathbb{Q}(\sqrt{-2})$ is badly approximable if and only if its *T*-orbit is bounded away from the fixed points (12 vertices of the cubeoctahedron).

5.5 Bounded geodesic trajectories (Dani correspondence)

The previous two sections described characterizations of badly approximable numbers over in terms of continued fractions, but we can still give a characterization of badly approximable vectors via the Dani correspondence even when continued fractions aren't available.

Theorem 5.5.1 ([EGL16], Proposition 3.1). The vector $\mathbf{z} = (z_1, \ldots, z_n) \in \mathbb{R}^r \times \mathbb{C}^s$ is badly approximable over F if and only if the geodesic trajectory

$$\Gamma \cdot \Omega_{\mathbf{z}} \cdot K \subseteq \Gamma \setminus (\mathcal{H}^2)^r \times (\mathcal{H}^3)^s$$

is bounded, where

$$\Omega_{\mathbf{z}} = \left\{ \left(\left(\begin{array}{cc} 1 & z_1 \\ 0 & 1 \end{array} \right) \left(\begin{array}{cc} e^{-t} & 0 \\ 0 & e^t \end{array} \right), \dots, \left(\begin{array}{cc} 1 & z_n \\ 0 & 1 \end{array} \right) \left(\begin{array}{cc} e^{-t} & 0 \\ 0 & e^t \end{array} \right) \right) : 0 \le t \in \mathbb{R} \right\}.$$

This follows in a straight-forward fashion from the following version of Mahler's compactness criterion, stated here for SL_2 .
Theorem 5.5.2 ([EGL16], Theorem 2.2). A subset $\Gamma \cdot \Omega \subseteq \Gamma \backslash SL_2(F \otimes \mathbb{R})$ is precompact if and only if there exists $\epsilon > 0$ such that

$$\max\{\max_{i}\{|z_i|\}, \max_{i}\{|w_i|\}\} \ge \epsilon, \ (\mathbf{z}, \mathbf{w}) = (q, p)\omega,$$

for all $(0,0) \neq (q,p) \in \mathcal{O}_F^2$ and $\omega \in \Omega$. In other words, the two-dimensional \mathcal{O}_F -modules in $(F \otimes \mathbb{R})^2$ spanned by the rows of $\omega \in \Omega$ do not contain arbitrarily short vectors.

5.6 Examples (zeros of totally indefinite anisotropic rational binary quadratic and Hermitian forms)

In this section we give explicit examples of badly approximable vectors, namely zeros of totally indefinite F-rational binary quadratic and Hermitian forms.

5.6.1 Totally indefinite binary quadratic forms

As above, let F be a number field, $F \otimes \mathbb{R} \cong \mathbb{R}^r \times \mathbb{C}^s$, $\Gamma = SL_2(\mathcal{O}_F)$, and let

$$Q(x,y) = (x \ y) \begin{pmatrix} A & B/2 \\ B/2 & C \end{pmatrix} \begin{pmatrix} x \\ y \end{pmatrix} = Ax^2 + Bxy + Cy^2, \ A, B, C \in F$$

be an *F*-rational binary quadratic form with determinant $\Delta(Q) = AC - B^2/4$. We say *Q* is totally indefinite if $\sigma(\Delta) < 0$ for all real embeddings $\sigma : F \to \mathbb{R}$ and that *Q* is anisotropic if it has no non-trivial zeros in F^2 . Note that *Q* is anisotropic if and only if $-\Delta(Q)$ is not a square in *F*. Let Q_i be the form obtained by applying σ_i to the coefficients of *Q*, denote by $Z_i(Q)$ the zero set of Q_i

$$Z_i(Q) = \{ [z:w] \in P^1(\mathbb{C}) \text{ or } P^1(\mathbb{R}) : Q_i(z,w) = 0 \},\$$

and let $Z(Q) = \prod_i Z_i(Q)$ (a finite set of cardinality 2^n for totally indefinite Q).

The group Γ acts on binary quadratic forms by change of variable

$$Q^g(x,y) = (g^t Qg)(x,y) = Q(ax+by,cx+dy)$$

and also on $P^1(\mathbb{R})^r \times P^1(\mathbb{C})^s$ diagonally by linear fractional transformations

$$g \cdot ([z_1:w_1], \dots, [z_n:w_n]) = ([a_1z_1 + b_1w_1:c_1z_1 + d_1w_1], \dots, [a_nz_n + b_nw_n:c_nz_n + d_nw_n]),$$

where $a_i = \sigma_i(a)$ and similarly for b_i , c_i , and d_i . These actions are compatible in the sense that $g^{-1} \cdot Z(Q) = Z(Q^g)$. Without further remark, we identify $\mathbb{R}^r \times \mathbb{C}^s$ with a subset of $P^1(\mathbb{R})^r \times P^1(\mathbb{C})^s$ via $(z_i)_i \mapsto ([z_i : 1])_i$.

5.6.2 Totally indefinite binary Hermitian forms over CM fields

Let F be a CM field (an imaginary quadratic extension of a totally real field E) of degree 2nwith ring of integers \mathcal{O}_F and let H be an F-rational binary Hermitian form

$$H(z,w) = (\bar{z}\ \bar{w}) \begin{pmatrix} A & B \\ \overline{B} & C \end{pmatrix} \begin{pmatrix} z \\ w \end{pmatrix} = Az\bar{z} + \overline{B}z\bar{w} + B\bar{z}w + Cw\bar{w}, \ A, C \in E, \ B \in F,$$

where the overline is "complex conjugation" (the non-trivial automorphism of F/E). Let H_i be the form obtained by applying σ_i to the coefficients of H (noting that σ_i commutes with complex conjugation). We say H is totally indefinite if $\sigma_i(\Delta) < 0$ for all i, where $\Delta = \det(H) = AC - B\overline{B}$. We say H is anisotropic if $H(p,q) \neq 0$ for $(p,q) \in F^2 \setminus \{(0,0)\}$. Note that H is anisotropic if and only if $-\Delta$ is not a relative norm, $-\Delta \notin N_E^F(F)$. Denote by $Z_i(H)$ the zero set of H_i ,

$$Z_i(H) = \{ ([z:w] \in P^1(\mathbb{C}) : H_i(z,w) = 0 \},\$$

a circle in $P^1(\mathbb{C})$, and let $Z(H) = \prod_i Z_i(H)$. When H is totally indefinite, $Z(H) \cong (S^1)^s$ is an *s*-dimensional torus.

The group $\Gamma = SL_2(\mathcal{O}_F)$ acts on H by change of variable

$$H^{g}(z,w) = (\bar{g}^{t}Hg)(z,w) = H(az+bw,cz+dw), \ g = \begin{pmatrix} a & b \\ & \\ c & d \end{pmatrix} \in SL_{2}(\mathcal{O}_{F}),$$

and also on $P^1(\mathbb{C})^n$ diagonally by linear fractional transformations

$$g \cdot ([z_1:w_1], \dots, [z_n:w_n]) = ([a_1z_1 + b_1w_1:c_1z_1 + d_1w_1], \dots, [a_nz_n + b_nw_n:c_nz_n + d_nw_n])$$

where $a_i = \sigma_i(a)$ and similarly for b_i , c_i , and d_i . These actions are compatible in the sense that $g^{-1} \cdot Z(H) = Z(H^g)$. As before, we include $\mathbb{C}^s \hookrightarrow P^1(\mathbb{C})^s$ via $(z_i)_i \mapsto ([z_i : 1])_i$.

5.6.3 Examples from quadatic forms

The following is a generalization of the fact that quadratic irrationals are badly approximable over \mathbb{Q} (r = 1, s = 0), as discussed in 5.2 above. We should note that these examples can also be deduced from Theorem 6.4 of [Bur92] (with S the set of infinite places and N = 1).

Proposition 5.6.1. Let Q be a totally indefinite anisotropic F-rational binary quadratic form over a number field F. Then any vector $\mathbf{z} \in Z(Q)$ is badly approximable over F.

First we establish compactness of a subspace associated to Q. There are many references with discussions of compactness for anisotropic arithmetic quotients, e.g. [PR94], [Rag72], and [Mor15].

Lemma 5.6.2. Let Q be a totally indefinite F-rational anisotropic binary quadratic form, and let L_i be the line in \mathcal{H}^2 or \mathcal{H}^3 with endpoints $Z(Q_i)$. Then $\pi(\prod_i L_i)$ is compact in $\Gamma \setminus (\mathcal{H}^2)^r \times (\mathcal{H}^3)^s$, where $\pi : (\mathcal{H}^2)^r \times (\mathcal{H}^3)^s \to \Gamma \setminus (\mathcal{H}^2)^r \times (\mathcal{H}^3)^s$ is the quotient map.

Proof of Lemma 5.6.2. Without loss of generality, suppose Q has integral coefficients, $A, B, C \in \mathcal{O}_F$. Compactness of

$$SO(Q, \mathcal{O}_F) \setminus SO(Q, F \otimes \mathbb{R}) \subseteq \Gamma \setminus SL_2(F \otimes \mathbb{R}),$$

is a consequence of Mahler's compactness criterion as follows. For $g \in SO(Q, F \otimes \mathbb{R})$ and any $(0,0) \neq (\alpha,\beta) \in \mathcal{O}_F^2$, the quantity $\max_i \{ |\sigma_i(Q^g(\alpha,\beta))| \}$ is bounded away from zero because

$$0 \neq Q^g(\alpha, \beta) = Q(\alpha, \beta) \in \mathcal{O}_F,$$

and \mathcal{O}_F is discrete in $F \otimes \mathbb{R}$. By Mahler's criterion and continuity of Q viewed as a function $(\mathbb{R}^r \times \mathbb{C}^s)^2 \to \mathbb{R}^r \times \mathbb{C}^s, SO(Q, \mathcal{O}_F) \setminus SO(Q, F \otimes \mathbb{R})$ is precompact.¹ The inclusion above is a closed embedding, hence its image is compact.

To get the result in the locally symmetric space, note that

$$\pi(\prod_i L_i) = \Gamma \cdot SO(Q, F \otimes \mathbb{R})g \cdot K \subseteq \Gamma \backslash G/K,$$

¹ Due to choices of left/right actions, this technically shows that $SO(F \otimes \mathbb{R})/SO(\mathcal{O}_F)$ has compact closure in $SL_2(F \otimes \mathbb{R})/\Gamma$. However the left and right coset spaces are homeomorphic.

where $g \in SL_2(F \otimes \mathbb{R})$ is any element such that $gK \in \prod_i L_i$, and that $\Gamma \setminus G \to \Gamma \setminus G/K$ is proper. \Box

For a concrete example, the four vectors

$$\left(\pm\sqrt{2-\sqrt{2}},\pm\sqrt{2+\sqrt{2}}\right)\in\mathbb{R}^2$$

are badly approximable over $\mathbb{Q}(\sqrt{2})$ as they are roots of the totally indefinite anisotropic binary quadratic form $Q(x, y) = x^2 - (2 - \sqrt{2})y^2$ (anisotropic since $2 - \sqrt{2}$ is not a square in $\mathbb{Q}(\sqrt{2})$).

5.6.4 Examples from Hermitian forms

The following is a generalization of the fact that zeros of anisotropic binary Hermitian forms are badly approximable over imaginary quadratic fields (r = 0, s = 1). As in the case of quadratic irrationals over \mathbb{Q} , this can be demonstrated with continued fractions when the imaginary quadratic field is Euclidean, $F = \mathbb{Q}(\sqrt{-d}), d = 1, 2, 3, 7, 11$. Details for the imaginary quadratic case can be found in [Hin18a] or further on in this chapter.

Proposition 5.6.3. If $\mathbf{z} = (z_1, \ldots, z_n) \in \mathbb{C}^n$ is a zero of the totally indefinite anisotropic *F*-rational binary Hermitian form *H*, *i.e.* $\mathbf{z} \in Z(H)$, then \mathbf{z} is badly approximable.

As before, we first establish compactness of a subspace associated to H (once again, cf. [?], [Rag72], or [Mor15]).

Lemma 5.6.4. If H is a totally indefinite anisotropic F-rational binary Hermitian form, then $\pi(\prod_i P_i)$ is compact in $\Gamma \setminus (\mathcal{H}^3)^n$ where P_i is the geodesic plane in the *i*th copy of \mathcal{H}^3 whose boundary at infinity is the zero set $Z_i(H)$ and $\pi: (\mathcal{H}^3)^n \to \Gamma \setminus (\mathcal{H}^3)^n$ is the quotient map.

Proof of Lemma 5.6.4. Without loss of generality, suppose H has integral coefficients, $A, C \in \mathcal{O}_F$, $B \in \mathcal{O}_E$. Compactness of

$$SU(H, \mathcal{O}_F) \setminus SU(H, F \otimes \mathbb{R}) \subseteq \Gamma \setminus SL_2(F \otimes \mathbb{R}),$$

follows from Mahler's compactness criterion as follows. For $g \in SU(H, F \otimes \mathbb{R})$ and any $(0, 0) \neq (\alpha, \beta) \in \mathcal{O}_F^2$, the quantity $\max_i \{ |\sigma_i(H^g(\alpha, \beta))| \}$ is bounded away from zero because

$$0 \neq H^g(\alpha, \beta) = H(\alpha, \beta) \in \mathcal{O}_E$$

and \mathcal{O}_E is discrete in $F \otimes \mathbb{R}$. By Mahler's criterion and continuity of H viewed as a function $(\mathbb{C}^s)^2 \to \mathbb{C}^s, SU(H, \mathcal{O}_F) \setminus SU(H, F \otimes \mathbb{R})$ is precompact.² The inclusion above is a closed embedding, hence its image is compact.

To get the result in the locally symmetric space, note that $\pi(\prod_i P_i) = \Gamma \cdot SU(H, F \otimes \mathbb{R})g \cdot K \subseteq \Gamma \setminus G/K$ where $g \in SL_2(F \otimes \mathbb{R})$ is any element such that $gK \in \prod_i P_i$, and that $\Gamma \setminus G \to \Gamma \setminus G/K$ is proper. \Box

For a concrete example, the entire torus

$$\{(\sqrt{3}\cos s + i\sqrt{3}\sin s, \sqrt{3}\cos t + i\sqrt{3}\sin t) : s, t \in [0, 2\pi)\} \subseteq \mathbb{C}^2$$

is badly approximable over $\mathbb{Q}(\sqrt{5}, \sqrt{-1})$ as it is the zero set of the totally indefinite anisotropic $H(z, w) = |z|^2 - 3|w|^2$ (anisotropic since 3 is not a relative norm).

5.7 Examples via nearest integer continued fractions

In this section we focus on producing z with bounded partial quotients, extending the results of [BG12] to all of the Euclidean imaginary quadratic fields $K_d = \mathbb{Q}(\sqrt{-d}), d = 1, 2, 3, 7, 11$. We will show that there are countably many circles in the complex plane all of whose points have bounded partial quotients in their nearest integer continued fractions.

We will consider equivalence classes of indefinite integral binary Hermitian forms. A binary Hermitian form H(z, w) is a function of the form

$$H(z,w) = (\overline{z},\overline{w}) \begin{pmatrix} A & -B \\ -\overline{B} & C \end{pmatrix} \begin{pmatrix} z \\ w \end{pmatrix}$$
$$= Az\overline{z} - B\overline{z}w - \overline{B}z\overline{w} + Cw\overline{w}, \ A, C \in \mathbb{R}, \ B \in \mathbb{C}$$

We denote by $\Delta(H)$ the determinant $\det(H) = AC - |B|^2$ of the Hermitian matrix defining H. The binary Hermitian form H is *integral* over K if the matrix entries of H are integers, i.e. $A, C \in \mathbb{Z}$ and $B \in \mathcal{O}$. The form is *indefinite* (takes on both positive and negative values) if and only if $\Delta(H) < 0$.

² Once again, due to choices of left/right actions, this technically shows that $SU(H, F \otimes \mathbb{R})/SU(H, \mathcal{O}_F)$ has compact closure in $SL_2(F \otimes \mathbb{R})/\Gamma$. However the left and right coset spaces are homeomorphic.

$$Z(H) := \{ [z:w] \in P^1(\mathbb{C}) : H(z,w) = 0 \}$$

which is either a circle or a line in the chart $\mathbb{C}_z = \{[z:1] \in P^1(\mathbb{C})\}$

$$Z(H) \cap \mathbb{C}_z = \begin{cases} \{z : |z - B/A|^2 = -\Delta/A^2\} & \text{if } A \neq 0, \\ \{z : \operatorname{Re}(\overline{B}z) = C\} & \text{if } A = 0. \end{cases}$$

We will be interested in equivalence of forms over $GL_2(\mathcal{O})$, where $g \in GL_2(\mathbb{C})$ acts as a normalized linear change of variable on the left, ${}^{g}H = |\det(g)|(g^{-1})^*Hg^{-1}$ (here * denotes the conjugate transpose), and also with the Möbius action of $GL_2(\mathbb{C})$ on $P^1(\mathbb{C})$,

$$g \cdot [z:w] = [az + bw:cz + dw], \ g = \left(egin{array}{c} a & b \\ c & d \end{array}
ight)$$

We collect some easily verified facts in the following lemma.

Lemma 5.7.1. The following hold for the action ${}^{g}H = |\det(g)|(g^{-1})^{*}Hg^{-1}$, $g \in GL_{2}(\mathbb{C})$, on indefinite binary Hermitian forms.

- The action of $GL_2(\mathbb{C})$ is determinant preserving, i.e. $\Delta({}^{g}H) = \Delta(H)$.
- The map $H \mapsto Z(H)$ is $GL_2(\mathbb{C})$ -equivariant (i.e. $g \cdot Z(H) = Z({}^gH)$).

Furthermore, an integral form H is isotropic (i.e. H(z, w) = 0 for some $[z : w] \in P^1(K)$) if and only if $-\Delta(H)$ is in the image of the norm map $N_{\mathbb{Q}}^K : K \to \mathbb{Q}$.

Proof. For $g \in GL_2(\mathbb{C})$ we have

$$\det({}^{g}H) = |\det(g)|^{2} \det((g^{-1})^{*}Hg^{-1}) = \frac{|\det(g)|^{2} \det(H)}{\det(g)\overline{\det(g)}} = \det(H).$$

The second bullet follows from

$$((\bar{z}\ \bar{w})g^*)({}^gH)(g(z\ w)^t) = |\det(g)|((\bar{z}\ \bar{w})g^*)((g^{-1})^*Hg^{-1})g(z,w)^t$$
$$= |\det(g)|H(z,w).$$

Finally, the factorization (assuming $A \neq 0$ else $-\Delta = |B|^2$ is a norm and H(1,0) = 0)

$$AH(z,w) = |Az - Bw|^2 + \Delta |w|^2$$

shows that $-\Delta$ is a norm if and only if there are $z, w \in K$ not both zero with H(z, w) = 0. \Box

Suppose H is an indefinite integral binary Hermitian form of determinant Δ and $z = [a_0; a_1, \ldots]$ satisfies H(z, 1) = 0. Define $H_n = g_n^{-1}H$, where

$$g_n^{-1} = \begin{pmatrix} 0 & 1 \\ 1 & -a_n \end{pmatrix} \cdots \begin{pmatrix} 0 & 1 \\ 1 & -a_0 \end{pmatrix}, \ g_n = \begin{pmatrix} p_n & p_{n-1} \\ q_n & q_{n-1} \end{pmatrix},$$

with notation

$$H_n = \left(\begin{array}{cc} A_n & -B_n \\ \\ -\overline{B_n} & C_n \end{array}\right)$$

In particular, we have

$$A_n = H(p_n, q_n),$$

 $C_n = H(p_{n-1}, q_{n-1}) = A_{n-1}$

Note that $H_n(1, z_n) = 0$ for all $n \ge 1$ because $g_n(1/z_n) = z$.

The main observation for us is the following theorem, which is essentially Theorem 4.1 of [BG12] generalized to the other Euclidean K and arbitrary integral binary Hermitian forms. One could follow the inductive geometric proof of [BG12], but we give an algebraic proof analogous to one demonstrating that real quadratic irrationals have eventually periodic simple continued fraction expansions, e.g. [Khi97], Theorem 28. In fact, we may as well note that the proof of Theorem 5.7.2 below applies *mutatis mutandis* to integral binary quadratic forms over K, showing that the continued fraction expansions of quadratic irrationals over K are eventually periodic as expected.

Theorem 5.7.2. If [z : 1] is a zero of an indefinite integral binary Hermitian form H, then the collection $\{H_n : n \ge 0\}$ is finite.

Proof. In what follows, $\Delta = \Delta(H) = \Delta(H_n)$. The inequality

$$|z - p_n/q_n| \le \frac{\kappa}{|q_n|^2}$$

allows us to write

$$p_n = q_n z + \frac{\gamma_n}{q_n}, \ |\gamma_n| \le \kappa$$

where $\kappa = \sup_{z,n} \{ |q_n| |q_n z - p_n| \} < 2$ is the best constant from [Lak73] used in the proof of Lemma 3.3.3.

Substituting this into the formula for A_n above gives

$$A_{n} = H(q_{n}z + \gamma_{n}/q_{n}, q_{n})$$

$$= |q_{n}|^{2}H(z, 1) + A\left(\overline{q_{n}z}\frac{\gamma_{n}}{q_{n}} + q_{n}z\frac{\overline{\gamma_{n}}}{\overline{q_{n}}} + \left|\frac{\gamma_{n}}{q_{n}}\right|^{2}\right) - \overline{B}\frac{\overline{q_{n}}}{q_{n}}\gamma_{n} - B\frac{q_{n}}{\overline{q_{n}}}\overline{\gamma_{n}}$$

$$= A\left(\overline{q_{n}z}\frac{\gamma_{n}}{q_{n}} + q_{n}z\frac{\overline{\gamma_{n}}}{\overline{q_{n}}} + \left|\frac{\gamma_{n}}{q_{n}}\right|^{2}\right) - \overline{B}\frac{\overline{q_{n}}}{q_{n}}\gamma_{n} - B\frac{q_{n}}{\overline{q_{n}}}\overline{\gamma_{n}},$$

and

$$|A_n| \le |A|\kappa^2 + 2|B|\kappa + 2|A||z|\kappa \le |A|\kappa^2 + 4|B|\kappa + 2\kappa\sqrt{-\Delta}$$
$$\le \max\{|A|, |A|\kappa^2 + 4|B|\kappa + 2\kappa\sqrt{-\Delta}\} =: \eta.$$

(We take the max above so that the parameter η is useful for bounds on z_n , a_n when n = 0, cf. Corollary 5.7.4 below.) From this it follows that

$$\begin{aligned} |C_n| &= |A_{n-1}| \le \eta, \\ |B_n| &= \sqrt{A_n C_n - \Delta} \le \sqrt{\eta^2 - \Delta}, \end{aligned}$$

so that there are only finitely many possibilities for H_n .

By requiring H to be anisotropic, we bound the finitely many circles $Z(H_n)$ away from zero and infinity, obtaining bounded partial quotients.

Corollary 5.7.3. If [z : 1] is a zero of an anisotropic indefinite integral binary Hermitian form H, then z has bounded partial quotients in its nearest integer continued fraction expansion (and is therefore badly approximable over K).

104

A quantitative measure of the "hole" around zero (see Figures 5.5, 5.6) can be given in terms of the determinant $\Delta = \det(H)$ of the form, which in turn bounds the partial quotients and controls the approximation constant $|z - p/q| \ge C'/|q|^2$.

Corollary 5.7.4. If $z \in Z(H)$ is a zero of the anisotropic integral indefinite binary Hermitian form H of determinant Δ , then the remainders z_n , $n \ge 0$, are bounded below by

$$|z_n| \ge \frac{1}{\sqrt{-\Delta} + \sqrt{\eta^2 - \Delta}},$$

with partial quotients bounded above by

$$|a_n| \le \rho + \sqrt{-\Delta} + \sqrt{\eta^2 - \Delta},$$

where η is as in the proof of Theorem 5.7.2. From these bounds, one can produce a C'(H) > 0 such that $|z - p/q| \ge C'/|q|^2$ for all $p/q \in K$ and $[z:1] \in Z(H)$.

Proof. We use the notation of Theorem 5.7.2. Since $1/z_n$ lies on $Z(H_n) \cap \mathbb{C}_z$, which has radius $\sqrt{-\Delta}/|A_n|$ and center B_n/A_n , we have

$$\frac{1}{|z_n|} \le \left|\frac{B_n}{A_n}\right| + \frac{\sqrt{-\Delta}}{|A_n|} \le \sqrt{\eta^2 - \Delta} + \sqrt{-\Delta},$$
$$|z_n| \ge \frac{1}{\sqrt{-\Delta} + \sqrt{\eta^2 - \Delta}}.$$

We also have

$$a_{n+1} + z_{n+1} = \frac{1}{z_n}$$

so that

$$|a_{n+1}| \le |z_{n+1}| + \frac{1}{|z_n|} \le \rho + \sqrt{-\Delta} + \sqrt{\eta^2 - \Delta}.$$

From Theorem 5.3.1, it follows that $|z - p/q| \ge C'/|q|^2$ for all $p/q \in K$, where

$$C' = \frac{1}{\alpha(\beta+1)(\beta+\rho+1)}, \ \beta = \rho + \sqrt{-\Delta} + \sqrt{\eta^2 - \Delta}.$$

We conclude this section with an example of the above over $\mathbb{Q}(i)$. The number

$$\sqrt{3}e^{2\pi i/5} = [1+2i; -1+i, -3, 2+2i, -1+3i, -2, -2i, 2+2i, 3-i, -2+2i, \ldots]$$

is a zero of the anisotropic indefinite integral binary Hermitian form $H(z, w) = |z|^2 - 3|w|^2$. Data from 10,000 iterations of the continued fraction algorithm and bounds from Corollary 5.7.4 are:

$$\max_{\substack{0 \le n < 10000}} \{|a_n|\} = 4.47213 \dots \le 7.22749 \dots,$$
$$\min_{\substack{0 \le n < 10000}} \{|z_n|\} = 0.25201 \dots \ge 0.15336 \dots,$$
$$\min_{\substack{0 \le n < 10000}} \{|q_n(q_n z - p_n)|\} = 0.28867 \dots \ge 0.00563 \dots.$$

There are 64 distinct partial quotients a_n and 56 distinct forms H_n . The remainders z_n are shown on the left in Figure 5.2 below. The remainders appear to have different densities along the arcs $Z(H_n) \cap V$, spending more time on the circles of radius $\sqrt{3}$ (show in black in Figure 5.2) than on the circles of radius $\sqrt{3}/2$ (shown in red in Figure 5.2). The bound on the normalized error seems to be rather poor, but it is interesting to note that the minimum 0.28867... above agrees with $\frac{1}{2\sqrt{3}}$ to high precision. We give some explanation for this last point in 5.9, where better bounds lower bound for $\liminf_{|q|\to\infty} \{|q(qz-p)|\}$ are given for zeros of Hermitian forms.



Figure 5.2: The first 10,000 remainders z_n of $z = \sqrt{3}e^{2\pi i/5}$ over $\mathbb{Q}(i)$ (left) and the arcs $Z(H_n) \cap V$ (right).

5.8 Examples via right-angled continued fractions

We describe a "reduction theory" of sorts for binary Hermitian forms over $\mathbb{Q}(i)$. Call Hreduced if $S(H) \cap \mathcal{F} \neq 0$, where S(H) is the geodesic hemisphere with ideal boundary Z(H) and \mathcal{F} is the fundamental octahedron.

Proposition 5.8.1. There are only finitely many reduced indefinite integral binary Hermitian forms H of a given determinant $D = \det(H)$.

Proof. If a = 0 and S(H) is a vertical plane, then $D = -|b|^2$ and there are only finitely many choices for b. Also, the line $Z(H) = \operatorname{Re}(z\overline{b}) = c$ must intersect the unit square, so there are only finitely many choices for c. For $a \neq 0$, we will show that there are only finitely many reduced forms intersecting the part of \mathcal{F} near infinity (say \mathcal{F}_{∞} , the sixth of \mathcal{F} above the unit hemispheres centered at 0, 1, i, 1 + i). By symmetry (the existence of isometries permuting the six cuspidal pieces), this will suffice. The radius of Z(H), $-D/a^2$, is bounded above by -D and bounded below by $1/\sqrt{2}$ since it intersects \mathcal{F}_{∞} . Hence |a| is bounded. The center of Z(H), b/a is also bounded (else S(H)will not intersect \mathcal{F}_{∞}). Hence |b| is bounded. Therefore, there are only finitely many reduced forms.

Now, if z_0 is a zero of a reduced indefinite integral binary Hermitian form H, then applying the transformations associated to $T^n(z_0)$ gives a sequence of Hermitian forms H_n which are also reduced. Hence the orbit $\{H_n\}_{n\geq 0}$ is finite. If H is anisotropic, then the finitely many boundary circles of the finitely many forms stay away from the fixed points, and z_0 will be badly approximable. See Figure 5.3 for an example of one of these special orbits.

A similar argument can be given for the cubeoctahedral case. See Figure 5.4 for an example orbit (under T_B). [Note that 7 is not a norm from $\mathbb{Q}(\sqrt{-1})$ nor $\mathbb{Q}(\sqrt{-2})$ so that the orbits are bounded away from the fixed points in Figures 5.3 and 5.4.]



Figure 5.3: 10000 iterations of $T = T_B$ over $\mathbb{Q}(\sqrt{-1})$ on $z = \sqrt{7}(\cos(2\pi/5) + i\sin(2\pi/5))$.



Figure 5.4: 10000 iterations of T_B over $\mathbb{Q}(\sqrt{-2})$ on $z = \sqrt{7}(\cos(2\pi/5) + i\sin(2\pi/5))$.

5.9 Examples à la Liouville

Finally, we note that Proposition 5.6.1 and 5.6.3 have elementary proofs along the lines of Liouville's theorem. Let J be a totally indefinite, anisotropic, integral binary quadratic or Hermitian form. For $\mathbf{z} \in Z(J)$ and $p/q \in F$ with $\max_i \{|z_i - p_i/q_i|\} \le 1$, we have

$$|J_i(p_i/q_i, 1)| = |J(z_i, 1) - J(p_i/q_i, 1)| \le \kappa_i |z_i - p_i/q_i|$$

for some constant $\kappa_i > 0$ depending on z_i and J_i by the mean value theorem. Multiplying by $|q_i|^2$ we have

$$|J_i(p_i, q_i)| \le \kappa_i |q_i(q_i z_i - p_i)|.$$

Because J is anisotropic and integral, for any $0 < \lambda \leq \min\{\max_i\{|a_i|\} : 0 \neq a \in \mathcal{O}_F\}$ we have

$$\max_{i}\{|J_i(p_i, q_i)|\} \ge \lambda$$

Hence for some i_0 we have

$$\lambda \le \kappa_{i_0} |q_{i_0}(q_{i_0} z_{i_0} - p_{i_0})| \le \kappa_{i_0} \max_i \{ |q_i| \} \max_i \{ |q_i z_i - p_i| \},$$

and \mathbf{z} is badly approximable

$$\max_{i}\{|q_i|\} \max_{i}\{|q_i z_i - p_i|\} \ge \lambda \kappa_{i_0}^{-1}.$$

We refine the above argument for Hermitian forms in the imaginary quadratic case. For the related topic of approximation properties of quadratic irrationals in Z(H), see [Vul10], Theorem 1.1.

Theorem 5.9.1. Let $H(z, w) = Az\bar{z} - B\bar{z}w - \overline{B}z\bar{w} + Cw\bar{w}$ be a binary Hermitian form that is indefinite, anisotropic, and integral. Let $\Delta(H) = AC - |B|^2 < 0$ be the determinant of H and

$$\mu(H) = \min\{|H(p,q)| : (0,0) \neq (p,q) \in \mathcal{O}_K^2\} > 0$$

its absolute minimum. If H(z, 1) = 0, then z is badly approximable with

$$|q(qz-p)| \ge \frac{\mu(H)}{2\sqrt{-\Delta} + 2|A|\epsilon}$$

for any $p/q \in K$ with $|z - p/q| \leq \epsilon$. Hence

$$\liminf_{|q|\to\infty} \{|q(qz-p)|: p, q \in \mathcal{O}_K, q \neq 0\} \ge \frac{\mu(H)}{2\sqrt{-\Delta(H)}}.$$

Proof. We will apply the mean value theorem to an $f : \mathbb{R}^2 \to \mathbb{R}$ in the form

$$f(b_1, b_2) - f(a_1, a_2) = \left(\frac{\partial f}{\partial x}(c_1, c_2), \frac{\partial f}{\partial y}(c_1, c_2)\right) \cdot (b_1 - a_1, b_2 - a_2)$$

for some (c_1, c_2) on the line segment joining (a_1, a_2) and (b_1, b_2) . By the mean value theorem, the Cauchy-Schwarz inequality, and because H(z, 1) = 0, we have

$$\begin{aligned} |H(p/q,1)| &= |H(p/q,1) - H(z,1)| \le |2A(c_1 + c_2i) - 2B||z - p/q| \\ &= 2|A||(c_1 + c_2i) - B/A||z - p/q| \le (\sqrt{-\Delta} + |A|\epsilon)|z - p/q| \\ &= C_{\epsilon}|z - p/q|, \ C_{\epsilon} = 2\sqrt{-\Delta(H)} + 2|A|\epsilon, \end{aligned}$$

for any $p/q \in K$ with $|z - p/q| \leq \epsilon$. (Above we are using the fact that the point $c_1 + c_2 i$ is at most ϵ from the disk of radius $\sqrt{-\Delta}/|A|$ centered at B/A.) Multiplying by $|q|^2$ gives

$$0 < \mu(H) \le |A|p|^2 - B\bar{p}q - \overline{B}p\bar{q} + C|q|^2| \le C_{\epsilon}|q|^2|z - p/q|,$$

since H is anisotropic and integral. Therefore z is badly approximable with

$$|q(qz-p)| \ge \mu(H)/C_{\epsilon} > 0$$

Letting ϵ tend to zero gives

$$\liminf_{|q|\to\infty} \{|q(qz-p)|: p,q \in \mathcal{O}_K, q\neq 0\} \ge \frac{\mu(H)}{2\sqrt{-\Delta(H)}}.$$

5.10 Non-quadratic algebraic examples over CM fields

It should be emphasized that the badly approximable product of circles $Z(H) \subseteq \mathbb{C}^n$ contains non-quadratic algebraic vectors, parameterized as follows (cf. [Hin18b]). **Corollary 5.10.1.** Choose real algebraic numbers $\alpha_i \in [-2, 2], 1 \leq i \leq n, f \in F$, and a totally positive $e \in E \setminus N_E^F(F)$. Then the vectors

$$\mathbf{z} = (z_1, \dots, z_n), \ z_i = f_i + \sqrt{e_i} \cdot \frac{\alpha_i \pm \sqrt{\alpha_i^2 - 4}}{2}$$

are algebraic and badly approximable, where $f_i = \sigma_i(f)$, $e_i = \sigma_i(e)$.

We discuss the above over imaginary quadratic fields K; the general case follows the same argument. For z such that $|z|^2 = s/t \in \mathbb{Q}$ is not a norm from K, the anisotropic integral form $\begin{pmatrix} t & 0 \\ 0 & -s \end{pmatrix}$ shows that z is badly approximable. This essentially exhausts all of the examples we've given as we can translate an integral form by a rational to center it at zero, then clear denominators to obtain an integral form as described, i.e.

$$H = \begin{pmatrix} A & -B \\ -\overline{B} & C \end{pmatrix}, \ g = \begin{pmatrix} 1 & -B/A \\ 0 & 1 \end{pmatrix}, \ A \cdot {}^{g}H = \begin{pmatrix} A^{2} & 0 \\ 0 & AC - B\overline{B} \end{pmatrix}.$$

For some specific algebraic examples, consider quadratically scaled roots of unity $z = \sqrt{n\zeta}$ for $|z|^2 = n \in \mathbb{Q}$ not a norm, or the generalizations of examples from [BG12], $z = \sqrt[m]{a} + \sqrt{\sqrt[m]{a^2} - n}$ for $a \in \mathbb{Q}$, $\sqrt[m]{a^2} < n$, and $n = |z|^2$ not a norm. See Figures 5.5, 5.6 for visualizations of the orbits of algebraic numbers satisfying $|z|^2 = n$ for various n and d = 1, 3.



Figure 5.5: 20,000 iterates of T for $z = \sqrt[3]{2} + \sqrt{\sqrt[3]{4} - n}$ over $\mathbb{Q}(\sqrt{-1})$ with n = 4, 5, 6, 7.

We would like to characterize the badly approximable algebraic numbers captured above. One such characterization comes from a parameterization of the algebraic numbers on the unit circle (taken from the mathoverflow post [Con10]).



Figure 5.6: 10,000 iterates of T for $z = \sqrt[3]{2} + \sqrt{\sqrt[3]{4} - n}$ over $\mathbb{Q}(\sqrt{-3})$ with n = 2, 3, 4, 5.

Lemma 5.10.2 ([Con10]). The algebraic numbers w on the unit circle, $w\overline{w} = 1$, are those numbers of the form

$$w = \frac{u \pm \sqrt{u^2 - 4}}{2}$$

where u is a real algebraic number in the interval [-2,2]. If $u \neq \pm 2$, the minimal polynomial f of w is

$$f(t) = t^m g(t+1/t)$$

where g(t) is the minimal polynomial of u, $\deg(g) = m$. In particular, the degree of f is even.

Proof. We know that $w, \overline{w} = 1/w$ have the same minimal polynomial $f(x) \in \mathbb{Q}[x]$, say of degree d. One can deduce that $x^d f(1/x) = f(x)$ so that f(x) is palindromic (if $f(x) = \sum_k f_k x^k$ then $f_{d-k} = f_k$) and since the roots come in reciprocal pairs, d is even. The even degree palindromic polynomials are of the form $f(x) = x^{d/2}g(x+1/x)$ for some polynomial g

$$f(x) = \sum_{k=0}^{d} f_k x^k = x^{d/2} \sum_{k=0}^{d/2} f_k (x^k + 1/x^k)$$
$$= x^{d/2} \left(r(x+1/x) + \sum_{k=0}^{d/2} f_k (x+1/x)^k \right)$$
$$= x^{d/2} g(x+1/x), \ g(x) = r(x) + \sum_{k=0}^{d/2} f_k x^k,$$

noting that the difference $(x + 1/x)^k - (x^k + 1/x^k)$ is a polynomial in x + 1/x by symmetry of the binomial coefficients. The roots of the even degree palindromic f on the circle double cover the roots of g in the interval (-2, 2) (via $w = e^{i\theta} \mapsto u = 2\cos\theta$). Conversely, taking an irreducible

polynomial $g(x) \in \mathbb{Q}[x]$ of degree d/2 with a root in the interval (-2, 2) gives a degree d irreducible polynomial $f(x) = x^{d/2}g(x+1/x)$ with roots on the unit circle.

Hence we have the following corollary describing the badly approximable algebraic numbers coming from (indefinite anisotropic K-rational binary) Hermitian forms over imaginary quadratic fields.

Corollary 5.10.3. For any imaginary quadratic field K, there are algebraic numbers of each even degree over \mathbb{Q} that are badly approximable over K. Specifically, for any real algebraic number $u \in [-2, 2]$, any positive $n \in \mathbb{Q} \setminus N_{\mathbb{Q}}^{K}(K)$, and any $t \in K$, the number

$$z = t + \sqrt{n} \cdot \frac{u \pm \sqrt{u^2 - 4}}{2}$$

is badly approximable.

For instance, the examples in Figures 5.5 and 5.6 have t = 0 and $u = 2^{4/3}/\sqrt{n}$.

Chapter 6

Continued fractions for weakly Euclidean orders in definite Clifford algebras

6.1 Introduction

The "accidental" isomorphisms

$$\operatorname{Isom}^+(\mathcal{H}^2) = O_{\mathbb{R}}(1,2)^{\circ} \cong PSL_2(\mathbb{R}), \ \operatorname{Isom}^+(\mathcal{H}^3) = O_{\mathbb{R}}(1,3)^{\circ} \cong PSL_2(\mathbb{C})$$

relate two- and three-dimensional hyperbolic geometry and simultaneous approximation over a number field F via the arithmetic lattice $SL_2(\mathcal{O}_F)$.

Can one use hyperbolic geometry in higher dimensions to approximate $x \in \mathbb{R}^n \subseteq \partial \mathcal{H}^{n+1}$ in a similar fashion? One can describe \mathcal{H}^{n+1} as a subset of a Clifford algebra and Isom⁺(\mathcal{H}^{n+1}) as a group of two-by-two matrices (with entries in the Clifford algebra) acting as fractional linear transformations on this subset and as projective transformations on S^n as a sort of projective line over this Clifford algebra. This generalizes the two cases above, where the Clifford algebras are \mathbb{C} and \mathbb{H} in dimensions 2 and 3 respectively.

Given an order \mathcal{O} in this Clifford algebra, we can approximate $x \in \mathbb{R}^n$ as a quotient of two "integers" $p, q \in \mathcal{O}$ such that $pq^{-1} \in \mathbb{R}^n$. This too generalizes the above, although it seems less natural, using integers in a 2^n -dimensional algebra to approximate vectors in \mathbb{R}^n .

In 6.6, we look in particular at the order $\mathbb{Z}[i, j, ij] \subseteq \mathbb{H}$ (i.e. approximation in \mathbb{R}^3), giving a convergent nearest integer continued fraction algorithm. We prove (Theorem 6.6.1) that the partial quotients of zeros of (anisotropic indefinite rational binary) Hermitian forms are bounded under the assumption that the convergent denominators are increasing for the algorithm (in the same

manner as Theorem 5.7.2). We also give a Liouville type argument that such zeros of Hermitian forms are badly approximable in Theorem 6.6.2 (in the same manner as Theorem 5.9.1).

References for the " SL_2 " model of hyperbolic isometries include [Wat93], [Ahl84], [Ahl85], [Ahl86] [Por95], [McI16], [Lou01]. Some work on continued fractions and approximation in this setting have been considered as well: [Bea03], [LV18], [Sch69], [Sch74], [Vul93], [Vul95a], [Vul99]. More discussion and application of arithmetic subgroups in this " SL_2 " isometry group can be found in: [EGM87], [EGM88], [EGM90], [Maa49], [MWW89].

6.2 Clifford algebras

Let V be a finite dimensional vector space over a field K (of characteristic not equal to 2) and $Q : V \to K$ a quadratic form $(Q(\alpha v) = \alpha^2 Q(v))$, with associated symmetric blinear form 2B(v,w) = Q(v+w) - Q(v) - Q(w). We will furthermore assume throughout that B is non-degenerate $(B(w,V) = 0 \Rightarrow w = 0)$. The Clifford algebra Cliff(V,Q) associated to Q is $T(V)/(v^2 = Q(v))$, the tensor algebra over V modulo the ideal generated by forcing squaring $v \in V$ to be evaluation of Q at v. Let e_1, \ldots, e_n be an orthogonal basis for the quadratic space (V,Q). Then Cliff(V,Q) has dimension 2^n with basis $\{e_{i_1} \cdots e_{i_k} : 1 \leq i_1 < \ldots < i_k \leq n\}$ and relations generated by $e_i^2 = Q(e_i), e_i e_j = -e_j e_i$.

We have three involutions x', x^* , and $\bar{x} = (x')^* = (x^*)'$ whose action on the basis elements are

$$(e_{i_1} \cdots e_{i_k})' = (-e_{i_1}) \cdots (-e_{i_k}) = (-1)^k e_{i_1} \cdots e_{i_k},$$
$$(e_{i_1} \cdots e_{i_k})^* = e_{i_k} \cdots e_{i_1} = (-1)^{\frac{k(k-1)}{2}} e_{i_1} \cdots e_{i_k},$$
$$\overline{e_{i_1} \cdots e_{i_k}} = (-1)^{\frac{k(k+1)}{2}} e_{i_1} \cdots e_{i_k},$$

with $\ensuremath{^*}$ and $\ensuremath{^-}$ reversing the order of multiplication.

Note that the anisotropic vectors $Q(v) \neq 0$ are invertible, $v^2 = Q(v) \neq 0$ implies $v^{-1} = v/Q(v)$, and that $v\bar{v} = vv' = -Q(v)$. This extends to products of vectors, $x\bar{x} = (-1)^k Q(v_1) \cdots Q(v_k)$ if $x = v_1 \cdots v_k$. Hence the collection of products of anisotropic vectors forms a group, G(V,Q). This is a (rather small) subgroup of $\operatorname{Cliff}(V, Q)^{\times}$, but it is directly related to the orthogonal group O(V, Q) as follows.

Proposition 6.2.1. For $a \in \text{Cliff}(V,Q)^{\times}$ and $v \in V$, let $\rho_a(v) = a'va^{-1}$. Then $\rho_a \in GL(V)$ iff $a \in G(V,Q)$. Moreover, for $w \in V$, $\rho_w(v) = v - 2\frac{B(v,w)}{Q(w)}w$ is the reflection with respect to v. Hence we have an isomorphism

$$\rho: G(V,Q)/K^{\times} \to O(V,Q), \ aK^{\times} \mapsto \rho_a$$

6.3 A special case

We will be working with $K = \mathbb{R}$, $V = \mathbb{R}^{n-1}$, $Q(x) = -\sum_{k=1}^{n-1} x_i^2$, with $\operatorname{Cliff}(V,Q) =: \mathcal{C}_n$. We will identify \mathbb{R}^n with $\langle 1, e_1, \ldots, e_{n-1} \rangle_{\mathbb{R}} \subseteq \mathcal{C}_n$. In this case, we have $O(V,Q) \cong G_n^1 / \pm 1$, where $G_n^1 = \{x \in G_n : x\bar{x} = 1\}$. We also note that $\mathcal{C}_n \subseteq \mathcal{C}_{n+1}$ in an obvious way. We extend Q to \mathbb{R}^n by $\tilde{Q} = x_0^2 - Q(x_1, \ldots, x_{n-1}) = \sum_{i=0}^{n-1} x_i^2 = x\bar{x} = |x|^2$. Let \tilde{G}_n be the group generated by \mathbb{R}^n . We then have the following.

Proposition 6.3.1. There is an isomomorphism

$$\widetilde{\rho}: \widetilde{G}_n/\mathbb{R}^{\times} \to SO(\mathbb{R}^n, \widetilde{Q}), \ a\mathbb{R}^{\times} \mapsto axa'^{-1}.$$

A reflection σ_y perpendicular to y is given by $x \mapsto -y\bar{x}\bar{y}^{-1} = -yx'y'^{-1}$, and the composition of two reflections $\sigma_{y_2} \circ \sigma_{y_1}$ perpendicular to $y_1, y_2 \in \mathbb{R}^n$ is given by ρ_a where $a = y_2y_1^{-1}$ or $y_2y'_1$.

Proof. We have

$$\sigma_y x = x - 2\widetilde{B}(x, y) \frac{y}{\widetilde{Q}(y)}$$
$$= x - (x\overline{y} + y\overline{x})\overline{y}^{-1}$$
$$= -y\overline{x}\overline{y}^{-1}.$$

Composing two of these gives ρ_a as stated, noting that \bar{a} and ' are equal on \mathbb{R}^n , and $SO(\mathbb{R}^n, \tilde{Q})$ is the collection of all even products of reflections. Those $a \in \tilde{G}_n$ such that $axa'^{-1} = x$ for all $x \in \mathbb{R}^n$ are the non-zero reals; we have a = a' so a is even, and ax = xa with $x = e_i$ shows that a cannot contain monomials with e_i .

6.4 S^n as a projective line and the Vahlen group

Consider the set of pairs $(x, y) \in \widetilde{G}_n$ such that y = 0 or $xy^{-1} \in \mathbb{R}^n$, and call two such pairs equivalent if there exists $z \in \widetilde{G}_n$ such that (x, y) = (xz, yz) (scalars on the right). We can identify these equivalence classes with $\mathbb{R}^n \cup \{\infty\}$ via $[x : y] \sim [xy^{-1} : 1]$ or $[x : 0] \sim [1 : 0] =: \infty$, or with S^n , the unit *n*-sphere. We want to identify a subgroup of $GL_2(\mathcal{C}_n)$ that acts as fractional linear transformations on $\mathbb{R}^n \cup \{\infty\}$ or projective automorphisms (matrix multiplication on the left)

$$[x:y] \mapsto [ax + by : cx + dy], \ xy^{-1} \mapsto (axy^{-1} + b)(cxy^{-1} + d)^{-1}.$$

If the matrix g induces the identity function, then it is diagonal with entries in $Z(\mathcal{C}_n) \setminus \{0\}$, where the center $Z(\mathcal{C}_n)$ is \mathbb{R} if n is odd and $\mathbb{R} + \mathbb{R}e_1 \cdots e_{n-1}$ if n is even. One can check that composition of functions and matrix multiplication agree. If $y = (ax + b)(cx + d)^{-1}$ is bijective with $x, y \in \mathbb{R}^n$, then $x = (-d^*y + b^*)(c^*y - a^*)^{-1}$ and composing in either order gives equality of functions

$$1 = \begin{pmatrix} -ad^* + bc^* & ab^* - ba^* \\ -cd^* + dc^* & cb^* - da^* \end{pmatrix} = \begin{pmatrix} -d^*a + b^*c & -d^*b + b^*d \\ c^*a - a^*c & c^*b - a^*d \end{pmatrix}.$$

Therefore we have

$$ab^* = ba^*, \ cd^* = dc^*, \ a^*c = c^*a, \ b^*d = d^*b$$

 $\Delta = ad^* - bc^* = da^* - cb^* \in Z(\mathcal{C}_n), \ \Delta' = d^*a - b^*c = a^*d - c^*b \in Z(\mathcal{C}_n)$

If g is bijective, then the conditions above show that Δ , Δ' are multiplicative. Also $g(0) = bd^{-1}$, $g(\infty) = ac^{-1}, g^{-1}(0) = -b^*(a^*)^{-1}, g^{-1}(\infty) = -d^*(c^*)^{-1}$ are all in \mathbb{R}^n .

Now if g_1 , g_2 have entries in $\widetilde{G}_n \cup \{0\}$, then the product also has entries in $\widetilde{G}_n \cup \{0\}$, e.g. $a_1a_2 + b_1c_2 = b_1(b_1^{-1}a_1 + c_2a_2^{-1})a_2$. One can verify that if $a, c \in \widetilde{G}_n \cup \{0\}$, then $ac^{-1} \in \mathbb{R}^n$ iff $ac^* \in \mathbb{R}^n$.

Some convenient generators for the (orientation preserving) Möbius transformations are special orthogonal transformations, translations, similarities, and -1/x. In matrices these are

$$\begin{pmatrix} a & 0 \\ 0 & a' \end{pmatrix}, a \in \widetilde{G}_n, \begin{pmatrix} 1 & x \\ 0 & 1 \end{pmatrix}, x \in \mathbb{R}^n, \begin{pmatrix} t & 0 \\ 0 & 1/t \end{pmatrix}, t \in \mathbb{R}, \begin{pmatrix} 0 & -1 \\ 1 & 0 \end{pmatrix}.$$

Theorem 6.4.1. The collection $SV_n(\mathbb{R})$ of matrices¹ $g = \begin{pmatrix} a & b \\ & \\ c & d \end{pmatrix}$ such that

$$a, b, c, d \in \widetilde{G}_n \cup \{0\}, \ \Delta(g) = ad^* - bc^* = 1, \ ac^*, bd^* \in \mathbb{R}^n \ (ac^{-1}, bd^{-1} \in S^n)$$

form a group, acting bijectively on $\mathbb{R}^n \cup \{\infty\}$ via $(ax + b)(cx + d)^{-1}$. The fractional linear action extends to $\mathcal{H}^{n+1} = \{x_0 + x_1 e_1 + \ldots + x_n e_n : x_i \in \mathbb{R}, x_n > 0\} \subseteq \mathcal{C}_{n+1}$ and $PSV_n(\mathbb{R}) = SV_n(\mathbb{R})/\{\pm 1\}$ is isomorphic to $\mathrm{Isom}^+(\mathcal{H}^{n+1})$.

Proof. If $c \neq 0$ we have

$$\begin{pmatrix} a & b \\ c & d \end{pmatrix} = \begin{pmatrix} 1 & ac^{-1} \\ 0 & 1 \end{pmatrix} \begin{pmatrix} (c^*)^{-1} & 0 \\ 0 & c \end{pmatrix} \begin{pmatrix} 0 & -1 \\ 1 & 0 \end{pmatrix} \begin{pmatrix} 1 & c^{-1}d \\ 0 & 1 \end{pmatrix}.$$

For more details see [Wat93].

6.5 Continued fractions for weakly Euclidean orders

Let $\Lambda \subseteq \mathbb{R}^n$ be a lattice and suppose there is a fundamental domain $\mathcal{F} \subset \mathbb{R}^n$ for the action of $(\Lambda, +)$ on \mathbb{R}^n such that $\mathcal{F} \subseteq \{x \in \mathbb{R}^n : |x| < 1\}$. Then we have a "Euclidean algorithm" for pairs $a, b \in \Lambda, ab^{-1} \in \mathbb{R}^n$

$$\begin{pmatrix} a \\ b \end{pmatrix} = \begin{pmatrix} a_0 & 1 \\ 1 & 0 \end{pmatrix} \cdots \begin{pmatrix} a_n & 1 \\ 1 & 0 \end{pmatrix} \begin{pmatrix} r \\ 0 \end{pmatrix} = \begin{pmatrix} p_n & p_{n-1} \\ q_n & q_{n-1} \end{pmatrix} \begin{pmatrix} r \\ 0 \end{pmatrix},$$

where $a_0 \in \Lambda$ is the lattice point such that $x = ab^{-1} = a_0 + x_0$, $x_0 \in \mathcal{F}$, a_1 is the lattice point such that $(x - a_0)^{-1} = a_1 + x_1$, $x_1 \in \mathcal{F}$, etc. The element r is a "right GCD" for a, b in that

$$a = p_n r, \ b = q_n r, \ r = q_{n-1}^* a - p_{n-1}^* b.$$

Define $T: \mathcal{F} \to \mathcal{F}$ by Tx = y where 1/x = a + y, $a \in \Lambda$, $y \in \mathcal{F}$. For $x \in \mathbb{R}^n$ let $x = a_0 + x_0$, $x_n = T^n x_0$, $a_n + x_n = 1/x_{n-1}$. Then

$$xq_n - p_n = (-1)^n x_0 x_1 \cdots x_n.$$

118

¹ The Vahlen group, after K. Th. Vahlen [Vah02]. He was an unrepentant Nazi, perhaps the reason why his name is sometimes avoided in the literature.

We will furthermore assume that $\Lambda = \mathcal{O} \cap \mathbb{R}^n$ where \mathcal{O} is a discrete subring of \mathcal{C}_n closed under *, so that the matrices coming from iteration of T and their inverses have entries in \mathcal{O} . If the order \mathcal{O} is Euclidean, then the conditions above are statisfied, but it isn't necessary. We will say that such orders satisfy the *weak Euclidean property*.

If \mathcal{O} satisfies the weak Euclidean property, then we have the following.

Proposition 6.5.1. For $x \in \mathbb{R}^n \setminus \{pq^{-1} : p, q \in \mathcal{O}\}$, $p_n q_n^{-1}$ converges to x (exponentially in n). If $x = ab^{-1} \in \mathbb{R}^n \cap \mathcal{OO}^{-1}$, then for some n, $x_n = 0$ and $p_n/q_n = x$. If the denominators q_n are strictly increasing in norm, then there exists C > 0 such that $|x - p_n q_n^{-1}| < C/|q_n|^2$ for all x and n.

Proof. Since $q_n \in \mathcal{O}$ is discrete, q_n is uniformly bounded below unless $q_n = 0$. If n is least such that $q_n = 0$, then $x_n = 0$ and $x \in \mathcal{OO}^{-1}$. For $x \in \mathbb{R}^n \setminus \mathcal{OO}^{-1}$, the equality $xq_n - p_n = (-1)^n x_0 x_1 \cdots x_n$ then shows that $|x - p_n q_n^{-1}| \leq m M^n$ where $M = \sup_{v \in \mathcal{F}} \{|v|\} < 1$ and $m = \min_{a \in \mathcal{O}} \{|a|\}$.

For $x = ab^{-1} \in \mathbb{R}^n \cap \mathcal{OO}^{-1}$, the weak Euclidean property gives a "Euclidean algorithm"

$$\begin{split} a &= q_1 b + r_1, \ q_1, r_1 \in \mathcal{O}, \ |r_1| < |b| \\ b &= q_2 r_1 + r_2, \ q_2, r_2 \in \mathcal{O}, \ |r_2| < |r_1| \\ \text{etc.}, \end{split}$$

terminating in finite time since \mathcal{O} is discrete. This shows, taking the quotient $q = a_n$ and remainder $r_n = x_n$ at each step to be as given by the continued fraction algorithm, that the continued fraction algorithm terminates for $x = ab^{-1} \in \mathcal{OO}^{-1}$.

Finally, taking norms of the equality

$$x - p_n q_n^{-1} = (-1)^n x_0 x_1 \cdots x_n = (-1)^n (q_n^*)^{-1} (x_n^{-1} + q_n^{-1} q_{n-1})^{-1} q_n^{-1}$$

and bounding $|x_n^{-1}| > M > 1$, $|q_n^{-1}q_{n-1}| < m < 1$ gives

$$|x - p_n q_n^{-1}| \ge \frac{1}{(M - m)|q_n|^2}$$

6.6 Approximation in \mathbb{R}^3 via quaternions

Let $\mathcal{O} = \mathbb{Z}[i, j, ij]$ be the Lipschitz quaternions. Then $\Lambda = \mathcal{O} \cap \mathbb{R}^3 = \mathbb{Z}^3$ has the weak Euclidean property (the unit cube centered at zero is contained in the unit sphere). For $x \in \mathbb{R}^3 \setminus \{ab^{-1} : a, b \in \mathcal{O}\}$, we obtain a convergent infinite continued fraction and rational approximations $p_n q_n^{-1} \in \mathcal{OO}^{-1} \cap \mathbb{R}^3 \subseteq \mathbb{Q}^3$

$$x = [a_0; a_1, a_2, \ldots], \ p_n q_n^{-1} = [a_0; a_1, \ldots, a_n], \ \begin{pmatrix} p_n & p_{n-1} \\ q_n & q_{n-1} \end{pmatrix} = \begin{pmatrix} a_0 & 1 \\ 1 & 0 \end{pmatrix} \cdots \begin{pmatrix} a_n & 1 \\ 1 & 0 \end{pmatrix}$$

satisfying

$$x = (p_n + p_{n-1}x_n)(q_n + q_{n-1}x_n)^{-1}, \ xq_n - p_n = (-1)^n x_0 x_1 \cdots x_n = (-1)^n (q_n^*)^{-1} (x_n^{-1} + q_n^{-1}q_{n-1})^{-1}.$$

Assuming the denominators are increasing in norm, $|q_n| > |q_{n-1}|$, all of the convergents satisfy $|x - p_n q_n^{-1}| < C/|q_n|^2$ for some constant C > 0. This seems to be true experimentally (cf. Figure 6.1) and a proof analogous to those in 3.3.1 could probably be given with some effort. We also have $SV(\mathcal{O}) \cdot \infty = \mathbb{Q}^3$ in this case (i.e. every rational triple can be expressed as ab^{-1} for some $a, b \in \mathcal{O}$).

6.6.1 Badly approximable triples from Hermitian forms

Let $A, C \in \mathbb{R}$ and $B \in \mathbb{R}^n \subseteq \mathcal{C}_n$ with $\Delta = AC - |B|^2 < 0$. For $[x : y] \in S^n$ we consider the zero sets Z(H) of indefinite Hermitian forms

$$H(x,y) = (\bar{x},\bar{y}) \begin{pmatrix} A & B \\ \bar{B} & C \end{pmatrix} \begin{pmatrix} x \\ y \end{pmatrix},$$

which are spheres $S^{n-1} \subseteq S^n$. If $A \neq 0$, then the zero set is given by

$$\{x: H(x,1) = 0\} = \{x \in \mathbb{R}^n : |x + B/A|^2 = -\Delta/A\}.$$

The group SV_n has a right action on such H, $H^g = H \circ g = g^{\dagger}Hg$, where g^{\dagger} is the conjugate transpose, ($^-$ each entry and transpose the matrix). We have equality of zero sets $Z(H^g) =$



Figure 6.1: The ratios $q_{n-1}q_n^{-1}$, $1 \le n \le 10$, for 5000 randomly chosen points of $[-1/2, 1/2)^3$. All of the points lie in the unit sphere, giving experimental evidence that the convergent denominators are increasing.

 $g^{-1}Z(H)$. The quadratic form $\Delta(H) = AC - |B|^2$ is signature (1, n) and is SV_n -invariant. For reference, we have

$$H^{g} = \begin{pmatrix} A|\alpha|^{2} + \bar{\gamma}\bar{b}\alpha + \bar{\alpha}B\gamma + \bar{\gamma}C\gamma & A\bar{\alpha}\beta + \bar{\gamma}\overline{B}\beta + \bar{\alpha}B\delta + C\bar{\gamma}\delta \\ A\bar{\beta}\alpha + \bar{\delta}\overline{B}\alpha + \bar{\beta}B\gamma + C\bar{\delta}\gamma & A|\beta|^{2} + \bar{\delta}\overline{B}\beta + \bar{\beta}B\delta + C|\delta|^{2} \end{pmatrix}, \ g = \begin{pmatrix} \alpha & \beta \\ \gamma & \delta \end{pmatrix}.$$

We restrict our attention and consider three-dimensional continued fractions over the Lipschitz quaternions, i.e. $\mathbb{R}^3 \cong \langle 1, i, j \rangle_{\mathbb{R}}$. If $x \in \mathbb{R}^3$ lies on a rational sphere (i.e. [x : 1] is a root of a rational indefinite binary Hermitian form H) and the denominators are increasing for the algorithm, then the orbit $T^n(x)$ is constrained to lie on finitely many spheres (i.e. the zero sets of the orbit of H). If H is anisotropic, then $T^n(x)$ is bounded away from zero. We summarize this as follows.

Theorem 6.6.1. Assume the convergent denominators are increasing in norm. If x is a zero of an indefinite integral binary Hermitian form H and $H_n = H^{g_n}$ is the form obtained by the change of variable

$$g_n = \begin{pmatrix} p_n & p_{n-1} \\ q_n & q_{n-1} \end{pmatrix} = \begin{pmatrix} a_0 & 1 \\ 1 & 0 \end{pmatrix} \begin{pmatrix} a_1 & 1 \\ 1 & 0 \end{pmatrix} \cdot \dots \cdot \begin{pmatrix} a_n & 1 \\ 1 & 0 \end{pmatrix}$$

then the orbit $\{H_n\}_{n\geq 0}$ is finite. In particular, if $x \in \mathbb{R}^3$ is a zero of an indefinite anisotropic integral binary Hermitian form,

$$H(x,1) = 0, \ H(x,y) = (\bar{x},\bar{y}) \begin{pmatrix} A & B \\ \bar{B} & C \end{pmatrix} \begin{pmatrix} x \\ y \end{pmatrix},$$

 $A,C\in\mathbb{Z},\ B\in\langle 1,i,j\rangle_{\mathbb{Z}},\ \Delta:=AC-|B|^2<0,-\Delta\ not\ a\ sum\ of\ three\ squares,$

then the partial quotients of x are bounded. [Recall that n is a sum of three squares unless $n = 4^{a}(8b+7)$ for some a, b.]

Proof. Let

$$H_n = H^{g_n} = \begin{pmatrix} A_n & B_n \\ \overline{B}_n & C_n \end{pmatrix}$$

so that

$$A_{n} = A|p_{n}|^{2} + \bar{q}_{n}\overline{B}p_{n} + \bar{p}_{n}Bq_{n} + C|q_{n}|^{2},$$

$$B_{n} = A\bar{p}_{n}p_{n-1} + \bar{q}_{n}\overline{B}p_{n-1} + \bar{p}_{n}Bq_{n-1} + c\bar{q}_{n}q_{n-1},$$

$$C_{n} = A|p_{n-1}|^{2} + \bar{q}_{n-1}\overline{B}p_{n-1} + \bar{p}_{n-1}Bq_{n-1} + C|q_{n-1}|^{2}$$

$$= A_{n-1}.$$

As noted earlier, we have

$$xq_n - p_n = (-1)^n (q_n^*)^{-1} (x_n^{-1} + q_n^{-1}q_{n-1})^{-1}.$$

Assuming the convergent denominators increase in norm, this gives

$$|xq_n - p_n| \le \frac{K}{|q_n|}$$

for some K > 0. Hence $p_n = xq_n + \kappa q_n^{-1}$ for some $\kappa \in \mathbb{H}$ with $|\kappa| \leq K$. Putting this into the formulas for the coefficients of H_n above, we get

$$\begin{aligned} A_n &= H_n(p_n, q_n) = H_n((xq_n + \kappa q_n^{-1}, q_n) \\ &= A|xq_n + \kappa q_n^{-1}|^2 + \bar{q}_n \overline{B}(xq_n + \kappa q_n^{-1}) + (\bar{q}_n \bar{x} + \bar{q}_n^{-1} \bar{\kappa}) Bq_n + C|q_n|^2 \\ &= \bar{q}_n H(x, 1)q_n + A(xq_n \bar{q}_n^{-1} \bar{\kappa} + \kappa q_n^{-1} \bar{q}_n \bar{x} + |\kappa|^2 / |q_n|^2) + \bar{q}_n \overline{B} \kappa q_n^{-1} + \bar{q}_n^{-1} \bar{\kappa} Bq_n \\ &= 0 + A(xq_n \bar{q}_n^{-1} \bar{\kappa} + \kappa q_n^{-1} \bar{q}_n \bar{x} + |\kappa|^2 / |q_n|^2) + \bar{q}_n \overline{B} \kappa q_n^{-1} + \bar{q}_n^{-1} \bar{\kappa} Bq_n, \\ |A_n| \leq 2|A||x||\kappa| + |A||\kappa|^2 / |q_n|^2 + 2|B||\kappa| \leq |A|K^2 + 2|A|K + 2|B|K. \end{aligned}$$

Therefore the coefficients A_n are bounded. Since $C_n = A_{n-1}$, the coefficients C_n are also bounded. Finally, the action of SV_3 preserves the determinant Δ of the form H, so that $|B_n| = \sqrt{A_n C_n - \Delta}$ is bounded as well.

We can unconditionally show that zeros of anisotropic indefinite integral binary Hermitian forms are *badly approximable* in the sense that there exists $\lambda > 0$ such that

$$|\zeta - pq^{-1}| \ge \frac{\lambda}{|q|^2}, \ p, q \in \mathbb{Z}[i, j, k], \ pq^{-1} \in \mathbb{R}^3,$$

by the arguments of 5.9.

Theorem 6.6.2. Suppose $\zeta \in \mathbb{R}^3$ is a zero of the anisotropic indefinite integral binary Hermitian form

form

$$H = \begin{pmatrix} A & B \\ \overline{B} & C \end{pmatrix}, \ A, C \in \mathbb{Z}, \ B \in \mathbb{Z}[i, j], \ AC - |B|^2 < 0,$$
$$H(\zeta, 1) = A|\zeta|^2 + \overline{B}\zeta + \overline{\zeta}B + C = 0.$$

Then ζ is badly approximable as defined above. Specifically, if $|\zeta - pq^{-1}| < \epsilon$ then

$$|\zeta - pq^{-1}| \ge \frac{\mu(H)}{2|q|^2(|A|\epsilon + \sqrt{-\Delta})}$$

and

$$\liminf\{|q||\zeta q - p| : (p,q) \neq (0,0), p,q \in \mathbb{Z}[i,j,k], \ pq^{-1} \in \mathbb{R}^3\} \ge \frac{\mu(H)}{2\sqrt{-\Delta}}.$$

Proof. A version of the mean value theorem for $f: \mathbb{R}^3 \to \mathbb{R}$ is

$$f(b_1, b_2, b_3) - f(a_1, a_2, a_3) = \left(\frac{\partial f}{\partial x}(c_1, c_2, c_3), \frac{\partial f}{\partial y}(c_1, c_2, c_3), \frac{\partial f}{\partial z}(c_1, c_2, c_3)\right) \cdot (b_1 - a_1, b_2 - a_2, b_3 - a_3)$$

for some (c_1, c_2, c_3) on the line segment between (a_1, a_2, a_3) and (b_1, b_2, b_3) . Applying this to the function $H(\cdot, 1)$ and using the fact that $H(\zeta, 1) = 0$ along with the Cauchy-Schwarz inequality gives

$$|H(pq^{-1},1)| = |H(\zeta,1) - H(pq^{-1},1)| \le |2Ax + 2B||\zeta - pq^{-1}|$$

for some x between ζ and pq^{-1} . Supposing $|\zeta - pq^{-1}| < \epsilon$ gives

$$|H(pq^{-1},1)| \le 2|A||x + B/A||\zeta - pq^{-1}| \le 2|A|(\sqrt{-\Delta}/A + \epsilon)|\zeta - pq^{-1}|$$

= $2(\sqrt{-\Delta} + |A|\epsilon)|\zeta - pq^{-1}|.$

Multiplication by $|q|^2$ gives

$$\mu(H) \le |A|p|^2 + \bar{q}\overline{B}p + \bar{p}Bq + C|q|^2| \le 2(|A|\epsilon + \sqrt{-\Delta})|q|^2, \ q \ne 0,$$

where $\mu(H) = \min\{|H(p,q)| : (p,q) \neq (0,0), \ p,q \in \mathbb{Z}[i,j,k]\}$. Hence

$$|\zeta - pq^{-1}| \ge \frac{\mu(H)}{2|q|^2(|A|\epsilon + \sqrt{-\Delta})}.$$

Letting ϵ tend to zero gives

$$\liminf\{|q||\zeta q - p| : (p,q) \neq (0,0), p,q \in \mathbb{Z}[i,j,k], \ pq^{-1} \in \mathbb{R}^3\} \ge \frac{\mu(H)}{2\sqrt{-\Delta}}.$$

6.6.2 Experimentation and visualization for $\mathbb{Z}^3 \subseteq \mathbb{R}^3$

Here we include some experimentation and visualization.

• The number $\xi = (\sqrt{2} - 1)(1 + i + j)$ has periodic expansion

$$[0; \overline{1 - i - j, -2 - 2i - 2j, 1 - i - j, 2 + 2i + 2j}].$$

The number $\xi + 1 + i + j$ is a zero of the Hermitian form $|u|^2 - 2|v|^2$, and the symmetry of the coefficients keeps the remainders on the union of the lines x = y = z and -x = y = z.

• Shown in Figure 6.2 are the first 20000 remainders for $\xi = \sqrt{e/10}i + \sqrt{\pi/10}j + \sqrt{1 - (e + \pi)/10}$, which is a zero of the (isotropic, indefinite, integral) Hermitian form $|x|^2 - |y|^2$. By Theorem 6.6.1, conditional on the increasing of the denominators, the orbit should lie on finitely many spheres, which it seems to do.



Figure 6.2: 20000 remainders for $\sqrt{e/10}i + \sqrt{\pi/10}j + \sqrt{1 - (e + \pi)/10}$ lying on a finite collection of spheres.

• Here we have remainders for $\xi = \sqrt{2}i + \sqrt{3}j$, all of which lie on the plane x = 0. The

number ξ is a zero of $|x|^2 - 5|y|^2$, so the orbit should lie on finitely many spheres (which give circles when intersected with the plane).



Figure 6.3: 10000 remainders for $\sqrt{2}i + \sqrt{3}j$ all lying in the plane x = 0.

Figure 6.4, left, shows the orbit of ξ = √2 + √3i + √2j which, due to the symmetry of the coefficients, must lie in the planes x = ±z. The number ξ is also a zero of the (anisotropic) Hermitian form |x|² - 7|y|² so the remainders should like on finitely many spheres (circles in the planes x = ±z here) and stay bounded away from zero. On the right in Figure 6.4 is part of the orbit of π + ei + πj, which also lies on the planes x = ±z, but doesn't satisfy any integral Hermitian form.

6.7 Some examples of orders

Let $a_i, 1 \leq i \leq n-1$ be positive integers, $q(v) = -\sum_i a_i v_i^2$ negative definite, and $\text{Cliff}(\mathbb{R}^{n-1}, q) \subseteq \mathcal{C}_n$ the associated Clifford algebra. We have orders \mathcal{O}_q coming from the integer points,

$$\mathcal{O}_q = \mathbb{Z} + \mathbb{Z}\sqrt{a_1}e_1 + \ldots + \mathbb{Z}\sqrt{a_{n-1}}e_{n-1} + \ldots + \mathbb{Z}\sqrt{a_1\cdots a_{n-1}}e_1\cdots e_{n-1}$$



Figure 6.4: Left are 10000 remainders for $\sqrt{2} + \sqrt{3}i + \sqrt{2}j$ all lying in the planes $x = \pm z$. Compare with 10000 remainders for $\pi + ei + \pi j$ on the right.

For instance, when n = 2, we have orders in imaginary quadratic fields and there are five orders with the weak Euclidean property (maximal orders in $\mathbb{Q}(\sqrt{-d})$, d = 1, 2, 3, 7, 11). For some examples of weakly Euclidean orders when n = 3 and applications to sphere packings, see [She18].

In general, finding and classifying such orders seems like a difficult task (I couldn't identify the maximal order corresponding to $a_i = 1$ for all i) and one might conjecture that there are only finitely many weakly Euclidean orders in negative definite quaternion algebras over \mathbb{Q} (over all dimensions).

Chapter 7

Nearest integer continued fractions over norm-Euclidean number rings and simultaneous approximation

7.1 Introduction

This chapter is an attempt to develop continued fractions in the setting of approximation in $K \otimes \mathbb{R}$, where K is a norm-Euclidean number field, i.e. a number field for which the absolute value of the field norm is a Euclidean function on \mathcal{O}_K .

The link between Euclidean (not necessarily for the norm!) and class number one (PID) is quite strong. Conditional on GRH, any number ring with infinitely many units is a PID if and only if it is Euclidean [Wei73]. The unconditional results are good as well, for instance: if K/\mathbb{Q} is Galois and $rk(\mathcal{O}_K^{\times}) > 3$, then \mathcal{O}_K Euclidean if and only if \mathcal{O}_K is a PID [HM04]. There are number rings that are Euclidean but not norm-Euclidean, e.g. $\mathbb{Z}[\sqrt{14}]$ [Har04], but every number ring with unit rank ≥ 1 is k-stage Euclidean if and only if it has class number one (cf. [Coo76a], [Coo76b]). We also note that the S-integers of a number ring are Euclidean for large enough S [Mar75]. For a survey of the Euclidean algorithm in algebraic number fields, see [Lem95].

The absolute value of the field norm is convenient because it is multiplicative, but difficult to work with because it isn't convex, doesn't induce a metric, doesn't separate points, etc., unless $K = \mathbb{Q}$ or K is imaginary quadratic. Because of these difficulties, I could not prove much of anything about the proposed algorithm. At the end of the chapter, we discuss an accessible example, $\mathbb{Q}(\sqrt{2})$.

7.2 Simultaneous approximation over norm-Euclidean number fields via continued fractions

Let K/\mathbb{Q} be an extension of degree r + 2s, where r is the number of real embeddings σ_i : $K \hookrightarrow \mathbb{R}$ and s the number of pairs of complex conjugate embeddings $\tau_i : K \hookrightarrow \mathbb{C}$. The induced map

$$\Phi: K \to \mathbb{R}^r \times \mathbb{C}^s, \ \Phi(a) = (\sigma_1(a), \dots, \sigma_r(a), \tau_1(a), \dots, \tau_s(a)),$$

embeds the ring of integers \mathcal{O}_K as a discrete subring of $\mathbb{R}^r \times \mathbb{C}^s$ with covolume $2^{-s}\sqrt{|D_K|}$, where D_K is the field discriminant. One can ask about the approximation properties of $\Phi(K)$ inside $\mathbb{R}^r \times \mathbb{C}^s$, i.e. how well and in what manner can one approximate $X \in \mathbb{R}^r \times \mathbb{C}^s$ by elements $\Phi(a)$, $a \in K$. When $K = \mathbb{Q}$ or $K = \mathbb{Q}(\sqrt{-d})$, d = 1, 2, 3, 7, 11 (the norm-Euclidean imaginary quadratic fields), one approach is via continued fractions, which we'd like to generalize.

We view $\mathbb{R}^r \times \mathbb{C}^s$ in the boundary of $(\mathcal{H}^2)^r \times (\mathcal{H}^3)^s$, the product of r copies of hyperbolic twospace and s copies of hyperbolic three-space (upper half-space model). The group $\Gamma := PGL_2(\mathcal{O}_K)$ is a finite covolume discrete subgroup of $\operatorname{Isom}((\mathcal{H}^2)^r \times (\mathcal{H}^3)^s)$, where $g \in \Gamma$ acts via the Poincaré extension of the fractional linear action on the various ideal boundaries,

$$g = \begin{pmatrix} a & b \\ c & d \end{pmatrix} \in \Gamma, \ X = (x_1, \dots, x_r, z_1, \dots, z_s) \in \mathbb{R}^r \times \mathbb{C}^s$$
$$g \cdot X = \left(\frac{\sigma_1(a)x_1 + \sigma_1(b)}{\sigma_1(c)x_1 + \sigma_1(d)}, \dots, \frac{\tau_s(a)z_s + \tau_s(b)}{\tau_s(c)z_s + \tau_s(d)}\right).$$

If K is norm-Euclidean, i.e. there is a fundamental domain \mathcal{F} for the action of $(\mathcal{O}_K, +)$ on $\mathbb{R}^r \times \mathbb{C}^s$ contained in the open set

$$V_1 := \{ X \in \mathbb{R}^r \times \mathbb{C}^s : |x_1 \cdots x_r| |z_1 \cdots z_s|^2 < 1 \},\$$

then we can uniquely write $X = \lfloor X \rfloor + \{X\}$ with $\lfloor X \rfloor \in \Phi(\mathcal{O}_K)$ and $\{X\} \in \mathcal{F}$ and try to write X as a continued fraction

$$X = a_0 + \frac{1}{a_1 + \frac{1}{a_2 + \dots}} = [a_0; a_1, a_2, \dots]$$

131

where $a_0 = \{X\}, a_1 = \lfloor 1/X \rfloor$, etc., the norm-Euclidean condition ensuring that $\mathcal{F}^{-1} = \{1/X : X \in \mathcal{F}\}$ is contained in the complement of V_1 . If $\mathcal{F} \subseteq V_{\epsilon} = \{X \in \mathbb{R}^r \times \mathbb{C}^s : |x_1 \cdots x_r| |z_1 \cdots z_s|^2 < \epsilon\}, 0 < \epsilon < 1$, then \mathcal{F}^{-1} is in the complement of $V_{1/\epsilon}$ with $1/\epsilon > 1$. We use the notation

$$N(X) = |x_1| \cdots |z_s|^2$$

for the absolute value of the extension of the field norm below. The algorithm and notation are as follows:

$$X_{0} = X - a_{0}, \ a_{0} = \lfloor X \rfloor, \ \frac{1}{X_{n}} = a_{n+1} + X_{n+1}, \ X_{n+1} = \left\{ \frac{1}{X_{n}} \right\}, \ a_{n} = \left\lfloor \frac{1}{X_{n}} \right\rfloor,$$
$$X_{n} = T^{n}(X), \ T : \mathcal{F} \to \mathcal{F}, \ T(X) = \left\{ \frac{1}{X} \right\}.$$

The convergents $p_n/q_n \in K$ to $X, p_n, q_n \in \mathcal{O}_K$, are defined recursively by

$$p_{-1} = 1, q_{-1} = 0, p_n = a_n p_{n-1} + p_{n-2}, q_n = a_n q_{n-1} + q_{n-2}$$

i.e.

$$\begin{pmatrix} p_n & p_{n-1} \\ q_n & q_{n-1} \end{pmatrix} = \begin{pmatrix} a_0 & 1 \\ 1 & 0 \end{pmatrix} \cdots \begin{pmatrix} a_n & 1 \\ 1 & 0 \end{pmatrix}.$$

Here and elsewhere we abuse notation, ignoring the distinction between $a \in K$ and $a \in \Phi(K)$ whenever convenient. We introduce the notation

$$X = [a_0; a_1, a_2, \ldots] = a_0 + \frac{1}{a_1 + \frac{1}{a_2 + \cdots}}$$

although we are currently unable to show that the right hand side exists analytically.

The following lemma lists some purely algebraic properties of the continued fraction algorithm.

Lemma 7.2.1. For $X \in \mathbb{R}^r \times \mathbb{C}^s \setminus K^{r+s}$ we have

$$q_n \neq 0, \ (X_n)_i \neq 0 \text{ for } 1 \leq i \leq r+s, \ a_n \neq 0 \text{ for } n \geq 1,$$

$$X = \frac{p_n + X_n p_{n-1}}{q_n + X_n q_{n-1}}, \ q_n X_n - p_n = (-1)^n X_0 \cdots X_n, \ X - \frac{p_n}{q_n} = \frac{(-1)^n}{q_n^2 (1/X_n + q_{n-1}/q_n)},$$
$$\frac{q_n}{q_{n-1}} = a_n + \frac{q_{n-1}}{q_{n-2}} = a_n + \frac{1}{a_{n-1} + \frac{1}{\dots + \frac{1}{a_1}}} \stackrel{alg.}{=} [a_n; a_{n-1}, \dots, a_1],$$

where by the overset "alg." we emphasize that the algorithm, when applied to the rational q_n/q_{n-1} , does not necessarily produce the continued fraction on the right.

The following lemma gives convergence with respect to the extension of the field norm.

Lemma 7.2.2. For $X \in \mathbb{R}^r \times \mathbb{C}^s \setminus K^{r+s}$ we have

$$\lim_{n \to \infty} N(X - p_n/q_n) = 0$$

Proof. We have

$$N(X - p_n/q_n) = \frac{N(X_0 \cdots X_n)}{N(q_n)} \le N(X_0 \cdots X_n) \le \epsilon^{n+1} \to 0.$$

-	-	-	٦
			1
	_	_	_

7.3 Experimentation in $\mathbb{Q}(\sqrt{2})$

The simplest example that comes to mind is shown in Figure 7.1 for the norm-Euclidean field $\mathbb{Q}(\sqrt{2})$. The unit hyperbola $N(x_1, x_2) = 1$ is shown in red, a fundamental domain for the additive group of the integers on the plane is shown as a blue rectangle inside the unit hyperbola, the integers $\mathbb{Z}[\sqrt{2}]$ are in green, and the inversion of the fundamental domain is also shown in blue outside the unit hyperbola. The restriction of the algorithm to the diagonal (x, x) gives the usual nearest integer continued fraction algorithm (considered by Hurwitz [Hur89] and generalized by Nakada to α -continued fractions [Nak81]). It seems that the norms of the denominators are not necessarily increasing for this algorithm, but they "mostly" do so as shown in Figure 7.2 (i.e. the dots mostly seem to stay within the unit hyperbola). The table below shows relevant information for a somewhat random vector ("somewhat" because it was chosen to have non-increasing denominators early on)

$$(x_1, x_2) = (-0.164685..., 0.347722...)$$

"= " [0; -2 - 3\sqrt{2}, 4 + 2\sqrt{2}, 1 - \sqrt{2}, 12 - 10\sqrt{2}, -1 + \sqrt{2}, ...]



Figure 7.1: Setup for nearest integer continued fractions over $\mathbb{Q}(\sqrt{2})$: \mathcal{F} , \mathcal{F}^{-1} blue, $\mathbb{Z}[\sqrt{2}]$ green dots, N(x) = 1 red.

n	1	2	3	4	5	6
a_n	0	$-2 - 3\sqrt{2}$	$4+2\sqrt{2}$	$1-\sqrt{2}$	$12 - 10\sqrt{2}$	$-1 + 2\sqrt{2}$
$rac{p_n}{q_n}$	$\frac{0}{1}$	$\frac{1}{-2-3\sqrt{2}}$	$\frac{4+2\sqrt{2}}{-19-16\sqrt{2}}$	$\frac{1-2\sqrt{2}}{11}$	$\frac{56 - 32\sqrt{2}}{113 - 126\sqrt{2}}$	$\frac{-183+142\sqrt{2}}{-606+352\sqrt{2}}$
$rac{p_n}{q_n}pprox$	0	(-0.160, 0.445)	(-0.164, 0.322)	(-0.166, 0.348)	(-0.164, 0.347)	(-0.164, 0.347)
$ N(q_n) $	1	12	151	121	11711	119428

For all computations I've done, the algorithm seems to converge quickly (as it does for the example in the table).

Finally, we discuss periodicity for quadratic irrationals. For example, the vector

$$\left(\sqrt{3+\sqrt{2}},\sqrt{3-\sqrt{2}}\right) = (2.1010\dots, 1.2592\dots)$$

is a root of the totally indefinite anisotropic rational quadratic form $Q(x,y) = x^2 - (3 + \sqrt{2})y^2$ and is therefore badly approximable as shown in 5.6.3 and 5.9. As expected, the continued fraction expansion of the above vector seems to be eventually periodic

$$\left(\sqrt{3+\sqrt{2}},\sqrt{3-\sqrt{2}}\right)$$
 "=" $[2;\overline{4+4\sqrt{2},4}].$

and convergent, approximations to the first few convergents being

(2,2), (2.103, 1.396), (2.100, 1.289), (2.101, 1.266), (2.101, 1.260), (2.101, 1.259).



Figure 7.2: Ratio of the denominators q_n/q_{n+1} for 5000 random numbers and $1 \le n \le 10$.

Chapter 8

Miscellaneous results on the discrete Markoff spectrum

8.1 Introduction

In this chapter we give a quick overview of the discrete Markoff spectrum and give some identities for sums over Markoff numbers. Unfortunately for me, these sums are special cases of Mcshane's identity for the following sum over the lengths of simple closed geodesics on any oncepunctured torus (with complete hyperbolic metric):

$$\sum_{\gamma} \frac{1}{1 + e^{l(\gamma)}} = \frac{1}{2}.$$

Moreover a proof of Mcshane's identity along the lines of what is presented below was already given in [Bow96].

We also prove (although this is basically an observation I hadn't seen elsewhere) that the limits of the roots of Markoff forms down non-trivial paths in the associated tree of forms are transcendental. This follows from the fact that the continued fraction expansion of such numbers begin with arbitrarily large repeated words (so-called "stammering" continued fractions) and a result of [AB05] (an application of the Subspace Theorem of W. M. Schmidt). These numbers are badly approximable (with only ones and twos in their continued fraction expansion) and well-approximated by quadratic irrationals (the roots of the Markoff forms). Geometrically, they correspond to some of the simple non-closed geodesics maximally distant from the cusp of the once-punctured torus associated to the commutator subgroup of $SL_2(\mathbb{Z})$. These transcendental numbers are also given as infinite sums of rational numbers that can be written explicitly in terms of the path taken to construct them (in the tree of Markoff numbers). As an example, the number

$$\sum_{i=0}^{\infty} (-1)^{i+1} \left(3 - \frac{m_{i+2}}{m_i m_{i+1}}\right), \ m_{i+3} = 3m_{i+2}m_{i+1} - m_i, \ (m_1, m_2, m_3) = (5, 13, 194).$$

is transcendental, corresponding to a periodic path towards $\frac{3-\sqrt{5}}{2}$.

References for this chapter include: [CF89], [Aig13], [Cas57], [Mar79], [Mar80], [PM93], [Fro13], [Coh55], [Sch76].

8.2 Markoff numbers and forms, Frobenius coordinates, and Cohn matrices

Let

$$A = M_{\frac{0}{1}} = \begin{pmatrix} 1 & 2 \\ 2 & 5 \end{pmatrix}, \ B = M_{\frac{1}{1}} = \begin{pmatrix} 0 & -1 \\ 1 & 3 \end{pmatrix},$$

and recursively define $M_{\frac{p+p'}{q+q'}} = M_{\frac{p}{q}}M_{\frac{p'}{q'}}$ for p/q < p'/q' and pq' - p'q = -1, associating an element of $SL_2(\mathbb{Z})$ to each number $\mathbb{Q} \cap [0,1]$. Define $m_{\frac{p}{q}} := \frac{1}{3} \operatorname{tr} M_{\frac{p}{q}}$ and let $\mathcal{M} = \{m_{\frac{p}{q}} : p/q \in \mathbb{Q} \cap [0,1]\}$ be the *multiset* of these values. For each $M_{\frac{p}{q}}$ define the indefinite integral binary quadratic form

$$f_{\frac{p}{q}}(x,y) = cx^2 + (d-a)xy - by^2, \ M_{\frac{p}{q}} = \begin{pmatrix} a & b \\ c & d \end{pmatrix}.$$

The $f_{\frac{p}{q}}$ are fixed-point forms of the $M_{\frac{p}{q}}$. For three adjacent rationals in [0, 1],

$$\frac{p}{q} < \frac{p''}{q''} = \frac{p+p'}{q+q'} < \frac{p'}{q'}, \ pq'-p'q = -1,$$

we have

$$m_{\frac{p}{q}}^2 + m_{\frac{p'}{q'}}^2 + m_{\frac{p''}{q''}}^2 = 3m_{\frac{p}{q}}m_{\frac{p'}{q'}}m_{\frac{p''}{q''}},$$

and every unordered positive triple of integer solutions to the Markoff equation $x^2 + y^2 + z^2 = 3xyz$ except (1, 1, 1) and (1, 1, 2) occurs in this fashion.

For any solution $m \ge n_1, n_2$ of the Markoff equation, let u be the smallest non-negative solution to

$$u \equiv \pm n_1/n_2 \mod m$$
,

and let $v = \frac{u^2+1}{m}$. Then the matrices $M_{\frac{p}{q}}$ and the fixed-point forms $f_{\frac{p}{q}}$ satisfy (with $m = m_{\frac{p}{q}}$)

$$M_{\frac{p}{q}} = \begin{pmatrix} u & 3u - v \\ m & 3m - u \end{pmatrix}, \ f_{\frac{p}{q}}(x, y) = mx^2 + (3m - 2u)xy + (v - 3u)y^2.$$

The Markoff forms and Markoff equation were introduced by Markoff to characterize the indefinite binary quadratic forms $f(x, y) = ax^2 + bxy + cy^2$ (with real coefficients) satisfying

$$\frac{\mu_f}{\sqrt{\Delta_f}} > \frac{1}{3}, \ \mu_f = \inf\{|f(x,y)| : (x,y) \in \mathbb{Z}^2 \setminus \{(0,0)\}\}, \ \Delta_f = b^2 - 4ac.$$

A version of the coordinates p/q for Markoff numbers was introduced by Frobenius [Fro13]. The use of the matrices above is due to Cohn [Coh55]. A natural choice of generators (motivated by Markoff's characterization in terms of continued fractions) is

$$\widetilde{A} = 2 + \frac{1}{2 + \frac{1}{x}} = \begin{pmatrix} 5 & 2\\ 2 & 1 \end{pmatrix}, \quad \widetilde{B} = 1 + \frac{1}{1 + \frac{1}{x}} = \begin{pmatrix} 2 & 1\\ 1 & 1 \end{pmatrix}.$$

The matrices A, B, are conjugate to \widetilde{A} , \widetilde{B} , shifting the roots of the fixed point forms by -2 (so that the "first root" lies in the interval [0, 1]). In any case, any relevant pair generates the commutator subgroup Γ' of $\Gamma = SL_2(\mathbb{Z})$, with quotient $\Gamma' \setminus \mathcal{H}^2$ the maximally symmetric once-punctured torus.

Markoff [Mar79], [Mar80] showed the following.

Theorem 8.2.1. The set of values

$$(1/3,\infty) \cap \left\{ \frac{\mu_f}{\sqrt{\Delta_f}} : f = ax^2 + bxy + cy^2, \ a, b, c \in \mathbb{R}, \ \Delta_f > 0 \right\} = \left\{ \frac{1}{\sqrt{5}}, \ \frac{1}{\sqrt{2}}, \ \frac{5}{\sqrt{221}}, \dots \right\}$$

is discrete with 1/3 as its only limit point. Each form contributing is $SL_2(\mathbb{Z})$ -equivalent to exactly one of the Markoff forms $f_{\frac{p}{q}}$.

Geometrically, the Markoff forms correspond to the "minimal height" geodesics in the modular surface (the number $\frac{\sqrt{\Delta_f}}{2|a|}$ is the *y*-coordinate of the geodesic of the top of the semi-circle connecting the roots of the quadratic form f). Analytically, the Lagrange spectrum is the set of values

$$\liminf_{q \to \infty} \{ |q(q\alpha - p)| \}, \ \alpha \in \mathbb{R},$$

Lagrange spectrum (i.e. the discrete set of values less than 3).

For proofs (and much more), see [Aig13], [CF89], [Bom07], or [Cas57]. For geometric/topological discussions and results, see [Riv12], [CDG⁺98], [Ser85], [She85], [Spr17].

8.3 A variation of Mcshane's identity

Proposition 8.3.1. The function (see Figure 8.3)

$$\phi: \mathbb{Q} \cap [0,1] \to [0,1], \ p/q \mapsto u/m$$

is decreasing with

$$\lim_{r \to p/q^-} \phi(r) - \phi(p/q) = \phi(p/q) - \lim_{r \to p/q^+} \phi(r) = \frac{3}{2} - \frac{\sqrt{9m^2 - 4}}{2m}.$$

Consequently,

$$\sum_{m \in \mathcal{M}} 3 - \frac{\sqrt{9m^2 - 4}}{m} = \frac{7 - \sqrt{5} - \sqrt{8}}{2}$$

Proof. Suppose

$$\frac{p}{q} < \frac{p''}{q''} = \frac{p+p'}{q+q'} < \frac{p'}{q'}, \ pq'-p'q = -1,$$

with corresponding Markoff numbers m, m', and m''. Multiplying the Cohn matrices

$$\left(\begin{array}{cc} u & 3u-v \\ m & 3m-u \end{array}\right) \left(\begin{array}{cc} u' & 3u'-v' \\ m' & 3m'-u' \end{array}\right) = \left(\begin{array}{cc} u'' & 3u''-v'' \\ m'' & 3m''-u'' \end{array}\right)$$

gives

$$u'' = uu' + 3um' - \frac{u^2 + 1}{m}m', \ m'' = mu' - um' + 3mm'.$$

The formula for m'' above gives

$$\frac{u}{m} - \frac{u'}{m'} = \frac{3}{2} - \sqrt{\frac{9}{4} - \frac{1}{m^2} - \frac{1}{(m')^2}} = 3 - \frac{m''}{mm'},$$

and letting m or m' to infinity gives the limits in the Proposition.

139



Figure 8.1: A plot of (p/q, u/m) at depth ≤ 10 in the Farey tree (excluding (0, 1/2) and (1, 0)).

141

In order to relate this to Mcshane's identity, we note that the length of the geodesic associated to a primitive hyperbolic element g of $PSL_2(\mathbb{R})$ is

$$l(g) = 2 \operatorname{arccosh}\left(\frac{\operatorname{tr}(g)}{2}\right)$$

so that

$$\frac{1}{1+e^l} = 1 - \frac{\sqrt{t^2 - 1}}{t}, \ t = \operatorname{tr}(g), \ l = l(g)$$

For the Cohn matrices, we have tr(g) = 3m so that

$$\frac{1}{1+e^l} = \frac{1}{3} \left(3 - \frac{\sqrt{9m^2 - 4}}{m} \right).$$

The sum in the proposition above does not sum over all of the simple geodesics of the once-punctured torus associated with the commutator subgroup of $SL_2(\mathbb{Z})$. See [Bow96] for a proof of Mcshane's identity using the tree structure associated to the simple closed geodesics of a once punctured torus.

8.4 Transcendence of limits of Markoff forms

The normalized Markoff forms

$$x^{2} + \left(3 - 2\frac{u}{m}\right)xy + \left(\frac{u^{2}}{m^{2}} - 3\frac{u}{m} + \frac{1}{m^{2}}\right)y^{2}$$

tend, as $u/m \to \alpha$, to the form

$$x^{2} + (3 - 2\alpha)xy + \alpha(\alpha - 3)y^{2} = (x - \alpha y)(x - (\alpha - 3)y)$$

which has discriminant 9, absolute minimum 1, and roots α , $\alpha - 3$. Any such α is badly approximable, with partial quotients in $\{1, 2\}$. The "trivial" paths in the Farey tree (i.e. those representing rational numbers) give nothing new. The left and right limits give the forms with

$$\alpha = \frac{u}{m} \pm \left(\frac{3}{2} - \frac{\sqrt{9m^2 - 4}}{2m}\right)$$

whose roots are quadratic irrationals equivalent to the roots of the Markoff forms themselves. However, all of the other limiting forms give *transcendental* α . This is a consequence of [AB05], Theorem 1, a special case of which we give below, which in turn follows from the Subspace Theorem. **Theorem 8.4.1** (cf. [AB05], Theorem 1, with w = 2). Let $\alpha = [0; a_1, a_2, ...]$ and suppose there are arbitrarily large values of n such that

$$a_i = a_{n+i}, \ 1 \le i \le n,$$

i.e. the sequence of partial quotients has longer and longer prefixes that are "doubled." If α is not quadratic (the sequence of partial quotients is not eventually periodic), then α is transcendental.

Proposition 8.4.2. Let $\theta \in [0,1]$ be irrational and let $\alpha = \lim_{r \to \theta} \phi(r)$ be a limiting value of u/m down a non-trivial path in the Farey tree. Then α is transcendental.

Proof. The conditions of the previous theorem are satisfied along any non-trivial path via the "concatenation" rules for the continued fraction expansion of the roots of the Markoff forms. Every left going down the Farey tree doubles a prefix in the continued fraction expansion; see Figure 8.4. These prefixes will grow as long as there are infinitely many right and left turns (i.e. we follow a non-trivial path). \Box

For instance, following the path to $\theta = \frac{3-\sqrt{5}}{2}$ gives $\alpha = \frac{2}{5} + \sum_{i=0}^{\infty} (-1)^{i+1} \left(3 - \frac{m_{i+2}}{m_i m_{i+1}} \right) = 0.38658908134980549908 \dots,$ $2 + \alpha = [2_2, 1_2, 2_2, 1_4, 2_2, 1_2, 2_2, 1_4, 2_2, 1_2, 2_2, 1_4, 2_2, 1_2, 2_2, 1_4, 2_2, 1_4, \dots],$

where the subscript indicates repeated digits and m_i are the Markoff numbers in the recurrence

$$m_{i+3} = 3m_{i+2}m_{i+1} - m_i, \ (m_1, m_2, m_3) = (5, 13, 194).$$

More generally, the transcendental α above can be written as

$$\frac{2}{5} + \sum_{i=0}^{\infty} \epsilon_i \left(3 - \frac{m_{i+2}}{m_i m_{i+1}}\right)$$

with $\epsilon_i = \pm 1$ according one turns right (-) or left (+) down the Farey tree (starting at p/q = 1/2) and (m_i, m_{i+1}, m_{i+2}) is the corresponding Markoff triple.

These "irrational slopes" were studied in [Coh74], and a description for all forms with $\mu/\sqrt{\Delta} = 1/3$ was determined in [Yas98]. The Markoff forms correspond to simple closed geodesics





Figure 8.2: Top: Part of the "half-topograph" (or tree of triples) of words in A and B giving Markoff forms (the fixed point forms) and Markoff numbers (one-third of the trace). Bottom: The continued fraction expansion of the roots of the fixed point forms from the \tilde{A} , \tilde{B} tree.

Bibliography

[AB05]	Boris Adamczewski and Yann Bugeaud. On the complexity of algebraic numbers. II. Continued fractions. <u>Acta Math.</u> , 195:1–20, 2005.
[AB07a]	Boris Adamczewski and Yann Bugeaud. Palindromic continued fractions. <u>Ann. Inst.</u> <u>Fourier (Grenoble)</u> , 57(5):1557–1574, 2007.
[AB07b]	Boris Adamczewski and Yann Bugeaud. A short proof of the transcendence of Thue- Morse continued fractions. <u>Amer. Math. Monthly</u> , 114(6):536–540, 2007.
[Ahl84]	Lars V. Ahlfors. Old and new in Möbius groups. <u>Ann. Acad. Sci. Fenn. Ser. A I Math.</u> , 9:93–105, 1984.
[Ahl85]	Lars V. Ahlfors. Möbius transformations and Clifford numbers. In <u>Differential</u> geometry and complex analysis, pages 65–73. Springer, Berlin, 1985.
[Ahl86]	Lars V. Ahlfors. Clifford numbers and Möbius transformations in \mathbb{R}^n . In <u>Clifford</u> algebras and their applications in mathematical physics (Canterbury, 1985), volume 183 of <u>NATO Adv. Sci. Inst. Ser. C Math. Phys. Sci.</u> , pages 167–175. Reidel, Dordrecht, 1986.
[Aig13]	Martin Aigner. <u>Markov's theorem and 100 years of the uniqueness conjecture</u> . Springer, Cham, 2013. A mathematical journey from irrational numbers to perfect matchings.
[Art24]	Emil Artin. Ein mechanisches system mit quasiergodischen bahnen. <u>Abh. Math. Sem.</u> <u>Univ. Hamburg</u> , 3(1):170–175, 1924.
[Ati87]	Michael Atiyah. The logarithm of the Dedekind $\eta\text{-function}.$ <u>Math. Ann.</u> , 278(1-4):335–380, 1987.
[Bea95]	Alan F. Beardon. <u>The geometry of discrete groups</u> , volume 91 of <u>Graduate Texts in</u> <u>Mathematics</u> . Springer-Verlag, New York, 1995. Corrected reprint of the 1983 original.
[Bea03]	A. F. Beardon. Continued fractions, Möbius transformations and Clifford algebras. Bull. London Math. Soc., 35(3):302–308, 2003.
[BG12]	Wieb Bosma and David Gruenewald. Complex numbers with bounded partial quo- tients. J. Aust. Math. Soc., 93(1-2):9–20, 2012.
[Bia92]	Luigi Bianchi. Sui gruppi di sostituzioni lineari con coefficienti appartenenti a corpi quadratici immaginarî. Math. Ann., $40(3)$:332–412, 1892.

[Bil65]	Patrick Billingsley. <u>Ergodic theory and information</u> . John Wiley & Sons, Inc., New York-London-Sydney, 1965.
[Bom07]	Enrico Bombieri. Continued fractions and the Markoff tree. <u>Expo. Math.</u> , 25(3):187–213, 2007.
[Bor69]	Armand Borel. <u>Introduction aux groupes arithmétiques</u> . Publications de l'Institut de Mathématique de l'Université de Strasbourg, XV. Actualités Scientifiques et Industrielles, No. 1341. Hermann, Paris, 1969.
[Bow96]	B. H. Bowditch. A proof of McShane's identity via Markoff triples. <u>Bull. London</u> <u>Math. Soc.</u> , 28(1):73–78, 1996.
[Bre91]	Claude Brezinski. <u>History of continued fractions and Padé approximants</u> , volume 12 of <u>Springer Series in Computational Mathematics</u> . Springer-Verlag, Berlin, 1991.
[BS12]	Mladen Bestvina and Gordan Savin. Geometry of integral binary Hermitian forms. <u>J. Algebra</u> , 360:1–20, 2012.
[Bur92]	Edward B. Burger. Homogeneous Diophantine approximation in S-integers. Pacific J. Math., 152(2):211–253, 1992.
[Cas57]	J. W. S. Cassels. <u>An introduction to Diophantine approximation</u> . Cambridge Tracts in Mathematics and Mathematical Physics, No. 45. Cambridge University Press, New York, 1957.
[CDG+98]	David Crisp, Susan Dziadosz, Dennis J. Garity, Thomas Insel, Thomas A. Schmidt, and Peter Wiles. Closed curves and geodesics with two self-intersections on the punctured torus. <u>Monatsh. Math.</u> , 125(3):189–209, 1998.
[CF89]	Thomas W. Cusick and Mary E. Flahive. <u>The Markoff and Lagrange spectra</u> , volume 30 of <u>Mathematical Surveys and Monographs</u> . American Mathematical Society, Providence, RI, 1989.
[CFHS17]	Sneha Chaubey, Elena Fuchs, Robert Hines, and Katherine E. Stange. The dynamics of super-apollonian continued fractions. https://arxiv.org/abs/1703.08616, 2017.
[CM93]	David J. Crisp and William Moran. Single self-intersection geodesics and the Markoff spectrum. In <u>Number theory with an emphasis on the Markoff spectrum (Provo, UT, 1991)</u> , volume 147 of <u>Lecture Notes in Pure and Appl. Math.</u> , pages 83–93. Dekker, New York, 1993.
[Coh55]	Harvey Cohn. Approach to Markoff's minimal forms through modular functions. Ann. of Math. (2), 61:1–12, 1955.
[Coh74]	Harvey Cohn. Some direct limits of primitive homotopy words and of Markoff geodesics. pages 81–98. Ann. of Math. Studies, No. 79, 1974.
[Con10]	K. Conrad. http://mathoverflow.net/questions/38680, 2010.
[Coo76a]	George E. Cooke. A weakening of the Euclidean property for integral domains and applications to algebraic number theory. I. J. Reine Angew. Math., 282:133–156, 1976.

- [Coo76b] George E. Cooke. A weakening of the Euclidean property for integral domains and applications to algebraic number theory. II. J. Reine Angew. Math., 283/284:71–85, 1976.
- [Dan85] S. G. Dani. Divergent trajectories of flows on homogeneous spaces and Diophantine approximation. J. Reine Angew. Math., 359:55–89, 1985.
- [Dan15] S. G. Dani. Continued fraction expansions for complex numbers—a general approach. Acta Arith., 171(4):355–369, 2015.
- [Dan17] S. G. Dani. Convergents as approximants in continued fraction expansions of complex numbers with eisenstein integers. https://arxiv.org/abs/1703.07672, 2017.
- [Dim14] Vesselin Dimitrov. http://mathoverflow.net/questions/177481, 2014.
- [Ebe96] Patrick B. Eberlein. <u>Geometry of nonpositively curved manifolds</u>. Chicago Lectures in Mathematics. University of Chicago Press, Chicago, IL, 1996.
- [EGL16] Manfred Einsiedler, Anish Ghosh, and Beverly Lytle. Badly approximable vectors, C^1 curves and number fields. Ergodic Theory Dynam. Systems, 36(6):1851–1864, 2016.
- [EGM87] J. Elstrodt, F. Grunewald, and J. Mennicke. Vahlen's group of Clifford matrices and spin-groups. Math. Z., 196(3):369–390, 1987.
- [EGM88] J. Elstrodt, F. Grunewald, and J. Mennicke. Arithmetic applications of the hyperbolic lattice point theorem. Proc. London Math. Soc. (3), 57(2):239–283, 1988.
- [EGM90] J. Elstrodt, F. Grunewald, and J. Mennicke. Kloosterman sums for Clifford algebras and a lower bound for the positive eigenvalues of the Laplacian for congruence subgroups acting on hyperbolic spaces. Invent. Math., 101(3):641–685, 1990.
- [EGM98] J. Elstrodt, F. Grunewald, and J. Mennicke. <u>Groups acting on hyperbolic space</u>. Springer Monographs in Mathematics. Springer-Verlag, Berlin, 1998. Harmonic analysis and number theory.
- [EINN19] Hiromi Ei, Shunji Ito, Hitoshi Nakada, and Rie Natsui. On the construction of the natural extension of the Hurwitz complex continued fraction map. <u>Monatsh. Math.</u>, 188(1):37–86, 2019.
- [EL07] Nicholas Eriksson and Jeffrey C. Lagarias. Apollonian circle packings: number theory.
 II. Spherical and hyperbolic packings. Ramanujan J., 14(3):437–469, 2007.
- [ESK10] R. Esdahl-Schou and S. Kristensen. On badly approximable complex numbers. <u>Glasg.</u> Math. J., 52(2):349–355, 2010.
- [EW11] Manfred Einsiedler and Thomas Ward. Ergodic theory with a view towards number theory, volume 259 of Graduate Texts in Mathematics. Springer-Verlag London, Ltd., London, 2011.
- [FG07] Vladimir V. Fock and Alexander B. Goncharov. Dual Teichmüller and lamination spaces. In <u>Handbook of Teichmüller theory. Vol. I</u>, volume 11 of <u>IRMA Lect. Math.</u> Theor. Phys., pages 647–684. Eur. Math. Soc., Zürich, 2007.

- [Fro13] G. Frobenius. Über die *Markoffschen Zahlen*. Berl. Ber., 1913:458–487, 1913.
- [Ghy07] Étienne Ghys. Knots and dynamics. In <u>International Congress of Mathematicians</u>. Vol. I, pages 247–277. Eur. Math. Soc., Zürich, 2007.
- [GLM⁺03] Ronald L. Graham, Jeffrey C. Lagarias, Colin L. Mallows, Allan R. Wilks, and Catherine H. Yan. Apollonian circle packings: number theory. J. Number Theory, 100(1):1– 45, 2003.
- [GLM⁺05] Ronald L. Graham, Jeffrey C. Lagarias, Colin L. Mallows, Allan R. Wilks, and Catherine H. Yan. Apollonian circle packings: geometry and group theory. I. The Apollonian group. Discrete Comput. Geom., 34(4):547–585, 2005.
- [GLM⁺06a] Ronald L. Graham, Jeffrey C. Lagarias, Colin L. Mallows, Allan R. Wilks, and Catherine H. Yan. Apollonian circle packings: geometry and group theory. II. Super-Apollonian group and integral packings. Discrete Comput. Geom., 35(1):1–36, 2006.
- [GLM⁺06b] Ronald L. Graham, Jeffrey C. Lagarias, Colin L. Mallows, Allan R. Wilks, and Catherine H. Yan. Apollonian circle packings: geometry and group theory. III. Higher dimensions. <u>Discrete Comput. Geom.</u>, 35(1):37–72, 2006.
- [Har04] Malcolm Harper. $\mathbb{Z}[\sqrt{14}]$ is Euclidean. Canad. J. Math., 56(1):55–70, 2004.
- [Hat07] Toshiaki Hattori. Some Diophantine approximation inequalities and products of hyperbolic spaces. J. Math. Soc. Japan, 59(1):239–264, 2007.
- [Hen06] Doug Hensley. <u>Continued fractions</u>. World Scientific Publishing Co. Pte. Ltd., Hackensack, NJ, 2006.
- [HH93] R. R. Hall and M. N. Huxley. Dedekind sums and continued fractions. <u>Acta Arith.</u>, 63(1):79–90, 1993.
- [Hin18a] Robert Hines. Badly approximable numbers over imaginary quadratic fields. https: //arxiv.org/abs/1707.07231, 2018.
- [Hin18b] Robert Hines. Examples of badly approximable vectors over number fields. https://arxiv.org/abs/1809.07404, 2018.
- [HM04] Malcolm Harper and M. Ram Murty. Euclidean rings of algebraic integers. <u>Canad. J.</u> Math., 56(1):71–76, 2004.
- [Hop71] Eberhard Hopf. Ergodic theory and the geodesic flow on surfaces of constant negative curvature. Bull. Amer. Math. Soc., 77:863–877, 1971.
- [Hur87] A. Hurwitz. Über die Entwicklung complexer Grössen in Kettenbrüche. <u>Acta Math.</u>, 11(1-4):187–200, 1887.
- [Hur89] A. Hurwitz. Uber eine besondere Art der Kettenbruch-Entwicklung reeller Grössen. Acta Math., 12(1):367–405, 1889.

- [HZ74] F. Hirzebruch and D. Zagier. <u>The Atiyah-Singer theorem and elementary number</u> <u>theory</u>. Publish or Perish, Inc., Boston, Mass., 1974. Mathematics Lecture Series, No. <u>3</u>.
- [Khi97] A. Ya. Khinchin. <u>Continued fractions</u>. Dover Publications, Inc., Mineola, NY, russian edition, 1997. With a preface by B. V. Gnedenko, Reprint of the 1964 translation.
- [KL16] Dmitry Kleinbock and Tue Ly. Badly approximable *S*-numbers and absolute Schmidt games. J. Number Theory, 164:13–42, 2016.
- [KN17] Alex Kontorovich and Kei Nakamura. Geometry and arithmetic of crystallographic sphere packings. https://arxiv.org/abs/1712.00147, 2017.
- [Kuz32] R. Kuzmin. Sur un problème de Gauss. Atti Congresso Bologna 6, 83-89 (1932)., 1932.
- [Lak73] Richard B. Lakein. Approximation properties of some complex continued fractions. Monatsh. Math., 77:396–403, 1973.
- [Lak74] Richard B. Lakein. Continued fractions and equivalent complex numbers. <u>Proc. Amer.</u> Math. Soc., 42:641–642, 1974.
- [Lan58] Edmund Landau. <u>Elementary number theory</u>. Chelsea Publishing Co., New York, N.Y., 1958. Translated by J. E. Goodman.
- [Lee97] John M. Lee. <u>Riemannian manifolds</u>, volume 176 of <u>Graduate Texts in Mathematics</u>. Springer-Verlag, New York, 1997. An introduction to curvature.
- [Lem95] Franz Lemmermeyer. The Euclidean algorithm in algebraic number fields. Exposition. Math., 13(5):385–416, 1995.
- [Lou01] Pertti Lounesto. <u>Clifford algebras and spinors</u>, volume 286 of <u>London Mathematical</u> <u>Society Lecture Note Series</u>. Cambridge University Press, Cambridge, second edition, 2001.
- [Lun36] P. Lunz. Kettenbrüche, deren Teilnenner dem Ring der Zahlen 1 und $\sqrt{-2}$ angehören. PhD thesis, München, 1936.
- [LV18] Anton Lukyanenko and Joseph Vandehey. Ergodicity of iwasawa continued fractions via markable hyperbolic geodesics. https://arxiv.org/abs/1805.09312, 2018.
- [Maa49] Hans Maass. Automorphe Funktionen von meheren Veränderlichen und Dirichletsche Reihen. Abh. Math. Sem. Univ. Hamburg, 16(nos. 3-4):72–100, 1949.
- [Mah46] K. Mahler. On lattice points in *n*-dimensional star bodies. I. Existence theorems. <u>Proc.</u> Roy. Soc. London. Ser. A., 187:151–187, 1946.
- [Mar79] A. A. Markoff. Sur les formes quadratiques binaires indéfinies. <u>Math. Ann.</u>, 15:381–407, 1879.
- [Mar80] A. Markoff. Sur les formes quadratiques binaires indéfinies. <u>Math. Ann.</u>, 17(3):379–399, 1880.
- [Mar81] A. Markoff. Sur une question de Jean Bernoulli. Math. Ann., 19(1):27–36, 1881.

- [Mar75] Raj Markanda. Euclidean rings of algebraic numbers and functions. <u>J. Algebra</u>, 37(3):425–446, 1975.
- [McI16] Justin McInroy. Vahlen groups defined over commutative rings. <u>Math. Z.</u>, 284(3-4):901–917, 2016.
- [Mor15] Dave Witte Morris. Introduction to arithmetic groups. Deductive Press, [place of publication not identified], 2015.
- [MR03] Colin Maclachlan and Alan W. Reid. <u>The arithmetic of hyperbolic 3-manifolds</u>, volume 219 of Graduate Texts in Mathematics. Springer-Verlag, New York, 2003.
- [MWW89] C. Maclachlan, P. L. Waterman, and N. J. Wielenberg. Higher-dimensional analogues of the modular and Picard groups. Trans. Amer. Math. Soc., 312(2):739–753, 1989.
- [Nak81] Hitoshi Nakada. Metrical theory for a class of continued fraction transformations and their natural extensions. Tokyo J. Math., 4(2):399–426, 1981.
- [Nak88a] Hitoshi Nakada. On ergodic theory of A. Schmidt's complex continued fractions over Gaussian field. Monatsh. Math., 105(2):131–150, 1988.
- [Nak88b] Hitoshi Nakada. On metrical theory of Diophantine approximation over imaginary quadratic field. Acta Arith., 51(4):399–403, 1988.
- [Nak90] Hitoshi Nakada. The metrical theory of complex continued fractions. <u>Acta Arith.</u>, 56(4):279–289, 1990.
- [Per33] Oskar Perron. Diophantische Approximationen in imaginären quadratischen Zahlkörpern, insbesondere im Körper $\Re(i\sqrt{2})$. Math. Z., 37(1):749–767, 1933.
- [Per54] Oskar Perron. <u>Die Lehre von den Kettenbrüchen. Bd I. Elementare Kettenbrüche</u>. B.
 G. Teubner Verlagsgesellschaft, Stuttgart, 1954. 3te Aufl.
- [PM93] Andrew D. Pollington and William Moran, editors. <u>Number theory with an emphasis</u> on the Markoff spectrum, volume 147 of <u>Lecture Notes in Pure and Applied</u> Mathematics. Marcel Dekker, Inc., New York, 1993.
- [Por95] Ian R. Porteous. <u>Clifford algebras and the classical groups</u>, volume 50 of <u>Cambridge</u> Studies in Advanced Mathematics. Cambridge University Press, Cambridge, 1995.
- [PR94] Vladimir Platonov and Andrei Rapinchuk. <u>Algebraic groups and number theory</u>, volume 139 of <u>Pure and Applied Mathematics</u>. Academic Press, Inc., Boston, MA, 1994. Translated from the 1991 Russian original by Rachel Rowen.
- [Quê91] Roland Quême. On Diophantine approximation by algebraic numbers of a given number field: a new generalization of Dirichlet approximation theorem. <u>Astérisque</u>, (198-200):273–283 (1992), 1991. Journées Arithmétiques, 1989 (Luminy, 1989).
- [Rag72] M. S. Raghunathan. <u>Discrete subgroups of Lie groups</u>. Springer-Verlag, New York-Heidelberg, 1972. Ergebnisse der Mathematik und ihrer Grenzgebiete, Band 68.
- [Rat06] John G. Ratcliffe. <u>Foundations of hyperbolic manifolds</u>, volume 149 of <u>Graduate Texts</u> in Mathematics. Springer, New York, second edition, 2006.

- [RG72] Hans Rademacher and Emil Grosswald. <u>Dedekind sums</u>. The Mathematical Association of America, Washington, D.C., 1972. The Carus Mathematical Monographs, No. 16.
- [Riv12] Igor Rivin. Geodesics with one self-intersection, and other stories. <u>Adv. Math.</u>, 231(5):2391–2412, 2012.
- [Sch67a] Asmus L. Schmidt. Farey triangles and Farey quadrangles in the complex plane. <u>Math.</u> Scand., 21:241–295 (1969), 1967.
- [Sch67b] Wolfgang M. Schmidt. On simultaneous approximations of two algebraic numbers by rationals. Acta Math., 119:27–50, 1967.
- [Sch69] Asmus L. Schmidt. Farey simplices in the space of quaternions. <u>Math. Scand.</u>, 24:31–65, 1969.
- [Sch74] Asmus L. Schmidt. On the approximation of quaternions. <u>Math. Scand.</u>, 34:184–186, 1974.
- [Sch75a] Asmus L. Schmidt. Diophantine approximation of complex numbers. <u>Acta Math.</u>, 134:1–85, 1975.
- [Sch75b] Wolfgang M. Schmidt. Simultaneous approximation to algebraic numbers by elements of a number field. Monatsh. Math., 79:55–66, 1975.
- [Sch76] Asmus L. Schmidt. Minimum of quadratic forms with respect to Fuchsian groups. I.J. Reine Angew. Math., 286/287:341–368, 1976.
- [Sch77] Asmus L. Schmidt. Minimum of quadratic forms with respect to Fuchsian groups. II.J. Reine Angew. Math., 292:109–114, 1977.
- [Sch78] Asmus L. Schmidt. Diophantine approximation in the field $\mathbf{Q}(i(11^{1/2}))$. J. Number Theory, 10(2):151–176, 1978.
- [Sch80] Wolfgang M. Schmidt. <u>Diophantine approximation</u>, volume 785 of <u>Lecture Notes in</u> <u>Mathematics</u>. Springer, Berlin, 1980.
- [Sch82] Asmus L. Schmidt. Ergodic theory for complex continued fractions. <u>Monatsh. Math.</u>, 93(1):39–62, 1982.
- [Sch11] Asmus L. Schmidt. Diophantine approximation in the field $\mathbb{Q}(i\sqrt{2})$. J. Number Theory, 131(10):1983–2012, 2011.
- [Ser85] Caroline Series. The geometry of Markoff numbers. <u>Math. Intelligencer</u>, 7(3):20–29, 1985.
- [She85] Mark Sheingorn. Characterization of simple closed geodesics on Fricke surfaces. <u>Duke</u> Math. J., 52(2):535–545, 1985.
- [She18] Arseniy Sheydvasser. Quaternion orders and sphere packings. https://arxiv.org/ abs/1708.03405, 2018.

[Sie 89]Carl Ludwig Siegel. Lectures on the geometry of numbers. Springer-Verlag, Berlin, 1989. Notes by B. Friedman, Rewritten by Komaravolu Chandrasekharan with the assistance of Rudolf Suter, With a preface by Chandrasekharan. [Spr17] Boris Springborn. The hyperbolic geometry of Markov's theorem on Diophantine approximation and quadratic forms. Enseign. Math., 63(3-4):333-373, 2017. [Sta16] Katherine E. Stange. The sensual Apollonian circle packing. Expo. Math., 34(4):364– 395, 2016. [Sta18] Katherine E. Stange. The Apollonian structure of Bianchi groups. Trans. Amer. Math. Soc., 370(9):6169–6219, 2018. [SV17] K. Spalding and A. P. Veselov. Lyapunov spectrum of Markov and Euclid trees. Nonlinearity, 30(12):4428–4453, 2017. [Vah02] K. Th. Vahlen. Uber Bewegungen und komplexe Zahlen. Math. Ann., 55:585–593, 1902. [Vul93] L. Ya. Vulakh. Higher-dimensional analogues of Fuchsian subgroups of $PSL(2, \mathfrak{o})$. Trans. Amer. Math. Soc., 337(2):947–963, 1993. [Vul95a] Diophantine approximation in \mathbf{R}^n . Trans. Amer. Math. Soc., L. Ya. Vulakh. 347(2):573-585, 1995.[Vul95b] L. Ya. Vulakh. Diophantine approximation on Bianchi groups. J. Number Theory, 54(1):73-80, 1995.[Vul99] L. Ya. Vulakh. Farey polytopes and continued fractions associated with discrete hyperbolic groups. Trans. Amer. Math. Soc., 351(6):2295–2323, 1999. [Vul10] L. Ya. Vulakh. Hermitian points in Markov spectra. Int. J. Number Theory, 6(4):713– 730, 2010. [Wal82] Peter Walters. An introduction to ergodic theory, volume 79 of Graduate Texts in Mathematics. Springer-Verlag, New York-Berlin, 1982. P. L. Waterman. [Wat93] Möbius transformations in several dimensions. Adv. Math., 101(1):87-113, 1993.[Wei73] Peter J. Weinberger. On Euclidean rings of algebraic integers. pages 321–332, 1973. [Yas98] Shin-Ichi Yasutomi. The continued fraction expansion of α with $\mu(\alpha) = 3$. Acta Arith., 84(4):337-374, 1998.[Yas10] Dan Yasaki. Hyperbolic tessellations associated to Bianchi groups. In Algorithmic number theory, volume 6197 of Lecture Notes in Comput. Sci., pages 385–396. Springer, Berlin, 2010. [Zag82] Don Zagier. On the number of Markoff numbers below a given bound. Math. Comp., 39(160):709-723, 1982.[Zim84] Robert J. Zimmer. Ergodic theory and semisimple groups, volume 81 of Monographs

in Mathematics. Birkhäuser Verlag, Basel, 1984.